EXPLORATION OF SMART INFRASTRUCTURE FOR DRIVERS OF AUTONOMOUS VEHICLES

DESIGN SPACE, OUT-OF-VIEW VISUALIZATION AND EXPLANATIONS

DISSERTATION

an der Fakultät für Mathematik, Informatik und Statistik der Ludwig-Maximilians-Universität München

vorgelegt von

GESA WIEGAND

M.Sc. Elektro- und Informationstechnik

München, den 02.06.2020

Erstgutachter: Prof. Dr. Heinrich Hußmann Zweitgutachter: Prof. Dr. Susanne Boll

Tag der mündlichen Prüfung: 22.07.2020

ABSTRACT

The connection between vehicles and infrastructure is an integral part of providing autonomous vehicles information about the environment. Autonomous vehicles need to be safe and users need to trust their driving decision. When smart infrastructure information is integrated into the vehicle, the driver needs to be informed in an understandable manner what the smart infrastructure detected. Nevertheless, interactions that benefit from smart infrastructure have not been the focus of research, leading to knowledge gaps in the integration of smart infrastructure information in the vehicle. For example, it is unclear, how the information from two complex systems can be presented, and if decisions are made, how these can be explained. Enriching the data of vehicles with information from the infrastructure opens unexplored opportunities. Smart infrastructure provides vehicles with information to predict traffic flow and traffic events. Additionally, it has information about traffic events in several kilometers distance and thus enables a look ahead on a traffic situation, which is not in the immediate view of drivers. We argue that this smart infrastructure information can be used to enhance the driving experience. To achieve this, we explore designing novel interactions, providing warnings and visualizations about information that is out of the view of the driver, and offering explanations for the cause of changed driving behavior of the vehicle.

This thesis focuses on exploring the possibilities of smart infrastructure information with a focus on the highway. The first part establishes a design space for 3D in-car augmented reality applications that profit from smart infrastructure information. Through the input of two focus groups and a literature review, use cases are investigated that can be introduced in the vehicle's interaction interface which, among others, rely on environment information. From those, a design space that can be used to design novel in-car applications is derived.

The second part explores out-of-view visualizations before and during take over requests to increase situation awareness. With three studies, different visualizations for out-of-view information are implemented in 2D, stereoscopic 3D, and augmented reality. Our results show that visualizations improve the situation awareness about critical events in larger distances during take over request situations.

In the third part, explanations are designed for situations in which the vehicle drives unexpectedly due to unknown reasons. Since smart infrastructure could provide connected vehicles with out-ofview or cloud information, the driving maneuver of the vehicle might remain unclear to the driver. Therefore, we explore the needs of drivers in those situations and derive design recommendations for an interface which displays the cause for the unexpected driving behavior.

This thesis answers questions about the integration of environment information in vehicles'. Three important aspects are explored, which are essential to consider when implementing use cases with smart infrastructure in mind. It enables to design novel interactions, provides insights on how out-of-view visualizations can improve the drivers' situation awareness and explores unexpected driving situations and the design of explanations for them.

Overall, we have shown how infrastructure and connected vehicle information can be introduced in vehicles' user interface and how new technology such as augmented reality glasses can be used to improve the driver's perception of the environment.

ZUSAMMENFASSUNG

Autonome Fahrzeuge werden immer mehr in den alltäglichen Verkehr integriert. Die Verbindung von Fahrzeugen mit der Infrastruktur ist ein wesentlicher Bestandteil der Bereitstellung von Umgebungsinformationen in autonome Fahrzeugen. Die Erweiterung der Fahrzeugdaten mit Informationen der Infrastruktur eröffnet ungeahnte Möglichkeiten. Intelligente Infrastruktur übermittelt verbundenen Fahrzeugen Informationen über den prädizierten Verkehrsfluss und Verkehrsereignisse. Zusätzlich können Verkehrsgeschehen in mehreren Kilometern Entfernung übermittelt werden, wodurch ein Vorausblick auf einen Bereich ermöglicht wird, der für den Fahrer nicht unmittelbar sichtbar ist. Mit dieser Dissertation wird gezeigt, dass Informationen der intelligenten Infrastruktur benutzt werden können, um das Fahrerlebnis zu verbessern. Dies kann erreicht werden, indem innovative Interaktionen gestaltet werden, Warnungen und Visualisierungen über Geschehnisse außerhalb des Sichtfelds des Fahrers vermittelt werden und indem Erklärungen über den Grund eines veränderten Fahrzeugverhaltens untersucht werden. Interaktionen, welche von intelligenter Infrastruktur profitieren, waren jedoch bisher nicht im Fokus der Forschung. Dies führt zu Wissenslücken bezüglich der Integration von intelligenter Infrastruktur, mit einem Fokus auf die Autobahn.

Der erste Teil erstellt einen Design Space für Anwendungen von augmentierter Realität (AR) in 3D innerhalb des Autos, die unter anderem von Informationen intelligenter Infrastruktur profitieren. Durch das Ergebnis mehrerer Studien werden Anwendungsfälle in einem Katalog gesammelt, welche in die Interaktionsschnittstelle des Autos einfließen können. Diese Anwendungsfälle bauen unter anderem auf Umgebungsinformationen. Aufgrund dieser Anwendungen wird der Design Space entwickelt, mit Hilfe dessen neuartige Anwendungen für den Fahrzeuginnenraum entwickelt werden können.

Der zweite Teil exploriert Visualisierungen für Verkehrssituationen, die außerhalb des Sichtfelds des Fahrers sind. Es wird untersucht, ob durch diese Visualisierungen der Fahrer besser auf ein potentielles Übernahmeszenario vorbereitet wird. Durch mehrere Studien wurden verschiedene Visualisierungen in 2D, stereoskopisches 3D und augmentierter Realität implementiert, die Szenen außerhalb des Sichtfelds des Fahrers darstellen. Diese Visualisierungen verbessern das Situationsbewusstsein über kritische Szenarien in einiger Entfernung während eines Übernahmeszenarios.

Im dritten Teil werden Erklärungen für Situationen gestaltet, in welchen das Fahrzeug ein unerwartetes Fahrmanöver ausführt. Der Grund des Fahrmanövers ist dem Fahrer dabei unbekannt. Mit intelligenter Infrastruktur verbundene Fahrzeuge erhalten Informationen, die außerhalb des Sichtfelds des Fahrers liegen oder von der Cloud bereit gestellt werden. Dadurch könnte der Grund für das unerwartete Fahrverhalten unklar für den Fahrer sein. Daher werden die Bedürfnisse des Fahrers in diesen Situationen erforscht und Empfehlungen für die Gestaltung einer Schnittstelle, die Erklärungen für das unerwartete Fahrverhalten zur Verfügung stellt, abgeleitet.

Zusammenfassend wird gezeigt wie Daten der Infrastruktur und Informationen von verbundenen Fahrzeugen in die Nutzerschnittstelle des Fahrzeugs implementiert werden können. Zudem wird aufgezeigt, wie innovative Technologien wie AR Brillen, die Wahrnehmung der Umgebung des Fahrers verbessern können. Durch diese Dissertation werden Fragen über Anwendungsfälle für die Integration von Umgebungsinformationen in Fahrzeugen beantwortet. Drei wichtige Themengebiete wurden untersucht, welche bei der Betrachtung von Anwendungsfällen der intelligenten Infrastruktur essentiell sind. Durch diese Arbeit wird die Gestaltung innovativer Interaktionen ermöglicht, Einblicke in Visualisierungen von Informationen außerhalb des Sichtfelds des Fahrers gegeben und es wird untersucht, wie Erklärungen für unerwartete Fahrsituationen gestaltet werden können.

COLLABORATION STATEMENT

This thesis presents the results of the research I carried out between August 2017 to June 2020. While Chapter 2 and Chapter 7 present original content which was exclusively written for this thesis, parts of Chapter 1, Chapter 3, Chapter 4, Chapter 5 and Chapter 6 are based on collaborations with coauthors, students and supervisors. In the course of this thesis, colleagues, students and supervisors contributed to this work. Throughout the thesis I use the scientific "We" to acknowledge this great support. I worked in close collaboration with my supervisor Heinrich Hußmann, who provided feedback on study designs, research ideas and manuscripts. The students I supervised contributed with their Bachelor, Master Thesis and semester projects to this thesis. In my role as main supervisor, I determined the research topics, ideas and goals. Regarding the steps of the student work (e.g prototypes, concepts, study designs, evaluations, implementations or conducting the study) I made the final decision. Table 1 presents the respective contributions of others, to paper and projects.

My contribution	Collaboration		
Chapter 1: Introduction			
Section 1.1: Leading author of paper [297]	Feedback for the paper [297] (Yuanting Liu)		
Chapter 3: Stakeholder			
Conducting and evaluating interviews for Section 3.2 and 3.1.1			
Conducting and evaluating focus group Section 3.1.2	Conducting focus group Section 3.3 (Markus Bonk)		
Conducting and evaluating focus group Section 3.1.2	Conducting focus group Section 5.5 (Markus Bonk)		
Evaluating focus group Section 3.3			
Chapter 4: Design Space			
Recearch idea	Research idea (Christian Mai)		
L iterature review	Literature review (Christian Mai, Kai Holländer),		
Conducting and evaluating both focus groups	Evaluating focus group results (Christian Mai),		
Loading outher of regulting poper [202]	Writing the paper (Christian Mai, Kai Holländer) [298],		
Leading aution of resulting paper [298]	Feedback for the paper (Heinrich Hußmann) [298]		
Chapter 5: Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality			
	Implementation of the study,		
Section 5.1: Original research idea,	conducting the study,		
study design, statistical analysis	statistical analysis		
	(Antoine Poppe, Dominik Hiemer, Ahmed Zitoun)		
Section 5.2: Original research idea,	Implementation and conducting the study (Hua Wentao),		
Supervision of the study design and implementation,	Writing the paper (Christian Mai)		
Leading author of the resulting paper [299]	Feedback for the paper [299] (Yuanting Liu, Heinrich Hußmann)		
Section 5.3: Original research idea,	In all months in and an all statices the state (Destation Com)		
Supervision of the study design and implementation,	Implementation and conducting the study (Beatrice Sax),		
Statistical analysis	Feedback for the research (Heinrich Hußmann)		
Chapter 6: Identifying the Needs of Drivers in Unexpected Driving Situations			
Centing (1, Original according to the	Study design,		
Section 6.1: Original research idea,	Implementation and conducting the user study,		
Study design and reedback for implementation, Conducting the study,	statistical analysis,		
Statistical analysis,	writing the paper [300] (Matthias Schmidmaier, Thomas Weber)		
Leading author of the resulting paper [300]	Feedback for the paper [300] (Yuanting Liu, Heinrich Hußmann)		
Section 6.2: Original research idea,			
Supervision of study design,	Implementation and conducting the study (Maximilian Haubelt),		
Implementation and conducting the study,	Coding of qualitative data (Maximilian Haubelt),		
Web search,	Writing the paper [296] (Malin Eiband)		
Data Analysis,	Feedback for the paper [296] (Heinrich Hußmann)		
Leading author of the resulting paper [296]			

Table 1: Collaboration in this thesis.

ACKNOWLEDGMENTS

During my PhD thesis I received great help and encouragement from different sides. First and foremost I want to thank Professor **Heinrich Hußmann** for supervising me, his great support and the numerous discussions that helped me find and focus my topic. Next, I would like to thank Professor **Susanne Boll-Westermann** for being on my committee and, more importantly, the warm welcome in the AutoUI community. I also thank **Professor Schubert** to be a member of my committee.

I want to thank fortiss in general, for giving me the opportunity to pursue my degree and especially **Mr. Vallon** for supporting me. During the project Providentia, I worked with different colleagues on developing use cases and putting up highway infrastructure. Therefore, I would like to thank **Markus Bonk** for always keeping the spirit and cheering us on, **Juri Kuhn** for introducing me to the Bavarian culture, **Annkathrin Krämer** for being the most knowledgeable data fusion expert, **Christoph Schöller** for all the insights in his research, **Gereon Hinz** for his visions, **Philipp Quentin** and **Maximilian Schnettler** for knowing each and every screw of the infrastructure, **Martin Büchel** for the sport talks and **Venkat Lakshmi** for being a person one can always turn to with problems. Thank you for managing all the up and downs together during the project. Furthermore, I would like to thank **Zhiwei Han** for having a great working spirit, **Andreas Bayha** and **Johannes Eder** for the great talks on the corridor, **Frederik Diehl** for bringing in a calm and relaxed spirit to the office and **Thomas Brunner** for sharing the office twice with me. Additionally, I would like to thank **Matthias Schmidmaier**, for introducing me to slow coffee brewing and **Thomas Weber** for my great PhD hat.

At the LMU I would like to thank everyone from the media informatics institute for welcoming and integrating me in great group. I thank Professor Andreas Butz, Florian Alt and Albrecht Schmidt for creating such a great atmosphere at the media informatics institute and for enabling all the trips to connect with other researchers. Thank you Franziska Schwamb, Christa Feulner and Rainer Fink for keeping everything running and the very nice conversations. Thank you Steeven Villa and **Florian Bemman** for sharing your office with me and the great discussions about climbing and haptic feedback. Thank you Sarah Aragon Bartsch for the great room escape, Daniel Buschek for being the paper writing expert, Heiko Drewes for sharing all the eye tracking knowledge, Malin Eiband for giving all the great feedback, Amy Yanhong Li for sharing all those drawings, Sylvia Rothe for sharing the nuts from your garden and being a great person to talk about film, Christina Schneegaß for being such a committed organizer, Nada Terzimehic for being the expert of good taste, Sarah Völkel for your organisation talent, Dennis Dietz for being always in a great mood, Changkun Ou for all your knowledge about segmentation in lunch breaks, Beat Rossmy for your inspiring prototype ideas, Maria Fysaraki for all the chats when sharing the office with me, Michael Braun for the fun times at conferences, Kai Holländer for your unusual views and your enthusiasm, Jingyi Li for the nice coffee breaks, Christian Mai for your crazy energy, Bastian Pfleging for organizing the lecture with me, Thomas Kosch for your straightforward, warm-heartedly attitude to solve the chaos, Pascal Knierim for being the fastest snowboarder and the rest of the crew: Fiona Draxler, Martin Eberl, Ceenu George, Michael Chromik, David Englmeier, Linda Hirsch, Yong Ma, Carl Oechsner, Axel Hösl, Sarah Prange, Lukas Mecke, Hanna Schneider, Mariam Hassib, Mohamed Khamis, Renate Häuslschmid, Francesco Chiossi, Luke Haliburton, Jakob Karolus, Tonja Machulla, Sebastian Feger, Lewis Chuang, Jesse Grootjen, Matthias Hoppe, Florian Lang and Tobias Seitz.

Thank you to all the student's that worked with me and trusted me with their bachelor and master thesis: **Katharina**, **Maximilian**, **Wentao**, **Linda**, **Bea**, **Laura**, **Philipp** and **Nadine**.

A special thanks goes to the ones closest to me, that shape and support me for years and every day. Thank you **Vanessa** and **Lea**, for the great time I always have with you and for pushing me, as the strong persons you are. Thank you **Guntraud** and **Tine** for our inspiring conversations and for being such a great role model. My sisters and brothers are the toughest and kindest persons to talk to. Thank you **Anne, Elin, Elmar** and **Lennart** for being the strongest opinionated persons, for being the best travel companions, for always welcoming me at your homes and for always having my back! I know I can always rely on your support and this is invaluable for me! **Katharina** and **Christoph**, thank you for being now part of my family!

Vor allem möchte ich meiner **Mutter** und meinem **Vater** danken. Danke, dass ihr alle meine verrückten Ideen mittragt, indem ihr mich unterstützt und dass ich mich immer auf euch verlassen kann. Eure Hingabe und Unterstützung sind unglaublich wertvoll! Last but definitely not least: **Niels**, thank you for your support, strength, love and laughter during my thesis!

TABLE OF CONTENTS

1	Int	roduction	1
	1.1	Problem Statement and Research Questions	2
	1.2	Research Approach	3
	1.3	Main Contributions	4
	1.4	Thesis Setting	5
		1.4.1 Technical Description	6
		1.4.2 Use Cases and Scenarios	6
	1.5	Thesis Structure	7
2	Rel ture	ated Work: Designing for Autonomous Vehicles in Smart Infrastruc- es	9
	2.1	Autonomous Driving	9
		2.1.1 Situation Awareness of Drivers	11
		2.1.2 Take Over Request (TOR)	12
		2.1.3 Trust in Automation	14
	2.2	Smart Infrastructure and Connected Vehicles	16
	2.3	Out-of-View Visualization	18
	2.4	Use of Augmented Reality in Vehicles	20
	2.5	Design Spaces for Automotive Interfaces	22
	2.6	Explainable Smart Systems	25
		2.6.1 AI Design and Mental Models	26
		2.6.2 Perception of Explanations of Reality	27
	0.7	2.0.3 Explanations and Technology in Autonomous venicles	28
	2.7	Lessons Learned	29
3	Sta	keholder and Use Case Evaluation of Smart Infrastructure	33
	3.1	Drivers of Connected Vehicles	35
		3.1.1 Interview: OEM of Autonomous Vehicles	36
		3.1.2 Focus Group: Smart Infrastructure Experts	38
		3.1.3 Focus Group: Trame Scenarios	39
	3.2	Emergency Services	40
	3.3	Highway Operator	41
	3.4	Lessons Learned	43
4	Des	sign Space for In-Car 3D Augmented Reality Applications	47
	4.1	Methodology	49
	4.2	Creation of the Design Space	49
		4.2.1 Literature Review	49

		4.2.2 Focus Groups 51 4.2.3 Building the Design Space 51
	4.3	Design Space
		4.3.1 User
		4.3.2 Context
		4.3.3 Visualization
		4.3.4 Interaction
		4.3.5 Technology
		4.3.6 Differentiation From Other Design Spaces
	4.4	Usage of the Design Space
		4.4.1 Personas and Use Case
		4.4.2 Example Scenario 60
	4.5	Experience with the Design Process
	4.6	Discussion
	4.7	Lessons Learned
5	Ou Rea	t-of-View Visualization of Highly Automated Vehicles in Augmented lity 67
	5.1	Sketching Context to Improve the Trust in the Driving Decision
		5.1.1 AR Warnings in a Take Over Request
	5.2	Preparing Take Over Requests in a Stereoscopic 3D Representation
		5.2.1 Method
		5.2.2 Results
		5.2.3 Discussion
	5.3	Visualizations that Look Beyond
		5.3.1 Identification of Visualizations
		5.3.2 Method
		5.3.3 Results
		5.3.4 Discussion
	5.4	Lessons Learned
6	Ide	entifying the Needs of Drivers in Unexpected Driving Situations 89
	6.1	Influence of Unexpected Driving Behavior on the Driver of an Autonomous Vehicle . 90
		6.1.1 Extracting Scenarios
		6.1.2 Method
		6.1.3 Results
		6.1.4 Discussion
		6.1.5 Design Implications
	6.2	Developing Adequate Explanations in Unexpected Driving Situations 105
		6.2.1 Method
		6.2.2 Study
		6.2.3 Results

		6.2.4 Discussion	111 113
	6.3	Lessons Learned	114
7	Co	nclusion	117
	7.1	Summary of the Contributions	120
	7.2	Directions of Future Work	121
List of Figures 12			123
L	ist of	f Tables	125
R	efere	ences	127

1 Introduction

Smart infrastructure can be the enabler for limitless autonomous driving. Road technology is getting smarter every day, by integrating infrastructure that can transmit information to vehicles. These infrastructure-to-vehicle (I2V) connections provide the connected parties with data and information that adds to the vehicle's own data and information. Autonomous vehicles can use this information to predict their driving paths, extend their data range or use the infrastructure data as redundant sensor set. In the context of cities, it is called smart cities that set up a digitally connected sensor network. Smart infrastructure is called for, to improve traffic safety, time spend in traffic and to reduce the environmental effects of traffic. In 2019, 3059 road users were killed in Germany¹ and 67.967 road users were gravely injured in accidents in Germany². Traffic accidents result from several causes, such as drunk driving (2,4% of accidents with injured persons in 2018), turning off mistakes, mistake in starting off or entering the road from premises et cetera (10,6% of accidents with injured persons in 2018) ³. If the traffic participants had an awareness of the environment, for example with smart infrastructure, accidents could have been prevented.

Smart infrastructure offers the potential to improve the information of drivers, improve safety, increase the comfort of drivers, efficient and anticipatory driving and reduced pollution⁴ [120, 121, 152, 216]. McKinsey published a report on smart cities [306], investigating the effect of smart cities on not just mobility advantages, but on every aspect of living, for example, cost of living, safety of jobs. Mobility benefits, among others, from digital infrastructure by predictive maintenance of transportation infrastructure, real-time road navigation or congestion pricing. Smart city applications can improve the so-called quality-of-life indicators, such as health, environmental quality, safety or time and convenience, by 10 to 30%. Considering safety it would mean a decrease of 8 to 10% fatalities and of 20 to 35% shorter emergency response time. Greenhouse gas emission would be lower and air quality higher. Due to these advantages there are numerous smart city and smart infrastructure projects that focus on the adaptation of this new technology [54, 216].

This thesis investigates how smart infrastructure can be used to make the driver aware of the vehicle's environment and intention. This awareness enables user interfaces for intervention that effectively prevents user damage. Concretely, with additional data, drivers can be provided with information preparing them for take over requests (TORs) or warn them about traffic situations several kilometers away. If the vehicle reacts on information from smart infrastructure, drivers also need to be informed in an efficient and understandable manner.

¹ https://www.destatis.de/EN/Press/2020/02/PE20_061_46241.html, last accessed May 2020

² https://www.destatis.de/DE/Themen/Gesellschaft-Umwelt/Verkehrsunfaelle/Publikationen/ Downloads-Verkehrsunfaelle/verkehrsunfaelle-jahr-2080700187004.pdf?__blob=publicationFile, last accessed May 2020

³ https://www.destatis.de/EN/Themes/Society-Environment/Traffic-Accidents/Tables/drivermistakes.html, last accessed May 2020

⁴ https://www.kassel.de/buerger/verkehr_und_mobilitaet/verkehrsprojekte/veronika.php, last access April 2020

1.1 Problem Statement and Research Questions

With the potential of smart infrastructure many novel research problems are created that need to be addressed for a successful integration of smart infrastructure in current traffic systems. While some use cases and user groups are mentioned in smart infrastructure projects, a structured approach of identifying them is scarcely researched. Those need to be identified to derive solutions to integrate smart infrastructure information in existing interfaces, resulting in a true benefit for the user groups.

While some design spaces for the automotive domain exist [108, 144] in autonomous vehicles, new design requirements emerge through a changed use of the vehicle. Instead of driving manually and being turned towards the steering wheel, it is likely that drivers are immersed in secondary tasks and interact with the vehicle in new ways. Nevertheless, there is no structured way of designing new applications for future connected vehicles.

Considering the driver interface, there are knowledge gaps of how drivers should be informed about smart infrastructure information. While visualization concepts exist that provide drivers with information out of the view of the driver [101], the concrete problem of how to convey information that is farther away and comes from other sources than the own vehicle is not explored in depth. To improve the safety of drivers, it is crucial to gain in-depth information how smart infrastructure information can be transmitted to the driver to improve situation awareness and driving performance.

Autonomous vehicles connected to smart infrastructure, combine two highly complex algorithms, with each one providing possibly conflicting information to the driver of the vehicle. They derive their driving decision or traffic information from algorithms that remain non-transparent to drivers, if no information is provided to them. This applies all the more to smart infrastructure information which is part of the vehicle's driving decision.

With this thesis in-depth knowledge about the solution to this problem space is provided by answering following research questions (see Table 1.1):

Research Question	No.	Chapter	
I Identifying Use Cases and Stakeholders			
What are potential user groups and use cases of smart highway infrastructure?	(RQ1)	Chapter 3	
II Design Guidelines for Novel Interfaces			
What are design dimensions for 3D in-car AR applications when 5G and smart infrastructure enable communication, entertainment or more information about the environment?	(RQ2)	Chapter 4	
III Visualization of Information Outside the Driver's View			
How can we display information about events outside driver's field of view?	(RQ3)	Chapter 5	

IV Explaining Unexpected Vehicle Behavior			
What do drivers of autonomous vehicles in unexpected driving situations need and how do they want to interact with the vehicle in that moment?	(RQ4)	Chapter 6	
How can vehicle behavior be explained if an autonomous vehicle reacts unexpectedly on smart infrastructure information?	(RQ5)	Chapter 6	

Table 1.1: Summary of the research questions addressed in this thesis.

1.2 Research Approach

We approach the topic of designing for highly automated vehicles in smart infrastructure by starting with the definition of a problem space. In this thesis, each research question assumes that the vehicle uses vehicle-to-vehicle, infrastructure-to-vehicle or cloud information. Therefore, highly automated vehicles are considered that receive information from the outside which may be out of the driver's view or lead to an earlier information or warning of critical traffic situations. We investigated drivers of highly automated vehicles (SAE Level 3-5 [130]), even though manual driven vehicles could also benefit from smart infrastructure. With those level, drivers are either required to supervise the driving task and take over control of the vehicle in case of the vehicle reaching its system limitations or, in the other case, drivers drive fully autonomously and can immerse themselves in a secondary task. In initial brain storming sessions and focus groups (see Chapter 3), we identify use cases and users of smart infrastructure (RQ1). Next we focus on passengers of highly automated vehicles that are connected with smart infrastructure. From a use case set of applications of in-car 3D AR, we derive a design space for in-car 3D AR applications (RQ2). Afterward, we focus on drivers that still need to interact with the vehicle, but get infrastructure information to be prepared for an intervention in the driving task. We conducted four driving simulator studies to evaluate how smart infrastructure can increase situation awareness and driving performance (RQ3). Next we put a focus on the human needs in unexpected driving situations. If the vehicle acts on information from the infrastructure, do drivers accept it? Do they need an explanation? What are other needs that might arise from this? We did three studies to identify the drivers' needs in unexpected driving situations and to identify the explanation that one should aim for when giving context information and explanations in those situations. In this thesis, we used different research methods [163] to explore the problem space:

Informal Case Study An informal case study can be used when there is a lack of understanding of the application domain or if initial responses are aimed for. With observational techniques and interviews, we conducted an informal case study in Chapter 3 to first collect initial statements of potential stakeholder of smart infrastructure. In Chapter 5 we collected insights in visualization of critical traffic scenarios.

Ethnography Lazar et al. [163] describe ethnography in HCI as a combination of observation, interviews and participation to explore understanding for work cultures, in which the researcher does not have any expertise. In the work with firefighters and highway operators, we used this technique to get an understanding of their work. In Chapter 3, we visited a firefighter station and a highway operation center to observe the workplace and understand how these user groups conduct their everyday work. While we did not participate, we observed the workplace.

Interview In Chapter 3 we explored the problem space and potentials of smart infrastructure through interviews. From this we derived deep insights of experts in their field. Those are limited to single opinions or views, but we gained insights to an unknown field.

Focus Groups Focus groups can provide information from several persons at once and through brainstorming and discussion, aspects that were unknown could be discovered. Therefore, throughout the thesis (Chapters 3, 4, 5, 6), we did several focus groups.

Quantitative Study Within this thesis we conducted several quantitative studies (Chapters 5 and 6) to compare different condition.

Qualitative Study Qualitative Studies are good to gain in depth knowledge about the thoughts of the participants. We collected several statements during studies. In Section 6.1, we did a qualitative study through think-aloud statements of the participants. We develop a well-grounded theory from the qualitative data [163].

1.3 Main Contributions

We contribute with this thesis valuable results for the use of smart infrastructure for drivers of autonomous and manually driven vehicles. First, in an informal case study we identify potential use cases and user groups. Next, we develop a design space for in-car 3D augmented reality applications. With this design space researchers and practitioners can design visualizations that increase situation awareness in take over situations. Finally, we identify categories for a codebook that are important for drivers in unexpected driving situations. The research contributions can be classified by Wobbrock et al. [305], who introduced seven types of contributions, among others artifact contributions, methodological contributions or theoretical contributions. We follow with their description of the corresponding contribution.

Theoretical Contribution: Design Space for In-Car 3D Augmented Reality Applications The potential of connecting a vehicle to infrastructure and driving autonomously calls for novel vehicle interfaces. To facilitate the process of designing future applications, we derive a design space for in-car 3D augmented reality applications. In a structured approach, we set up a set of use cases of novel applications in 3D AR. From this we conclude a design space for which we present an example design process of an in-car 3D AR application.

Empirical and Artifact Contribution: Out-of-View Visualizations Since smart infrastructure will provide information that is not necessarily in sight of the driver, we explore, how those visualizations can be presented in the vehicle cockpit. We analyze how out-of-view visualizations can be used to convey context and hints on other vehicles and critical situations to prepare the driver for a take over request. For this purpose we implement different, some novel, visualizations, ranging from high detail level with a whole overview of the traffic situation to low detail level, with hints on other

traffic participants. With those implemented visualizations, we improve the situations awareness of an upcoming take over request and the driving performance of drivers.

Methodological and Empirical Contribution: Explanation Design Guidelines for Drivers in Unexpected Driving Situations Smart infrastructure information causes the problem of providing information without the user exactly knowing the source and resulting outcomes of the system. In a qualitative study, we derive explanation design guidelines for drivers in unexpected driving situations. We approach this by first deriving unexpected driving situations that are based on naturalistic situations to get a high ecological validity [8]. While this approach is novel, it results in many scenarios that can be used as a basis to collect statements of drivers in these situations. From these statements, the needs of drivers in unexpected driving situations were analyzed in a codebook. This codebook then is used to identify design guidelines. In a next step an explanation is provided to the driver. From this, we derive an adaptable explanation that increases situations awareness of the driving situation and a target mental model which should be considered when designing explanations for drivers in unexpected driving situations.

1.4 Thesis Setting

This section is partly based on the following publication:

• Gesa Wiegand and Yuanting Liu. 2018. Highway Sensor System as Enabler for Autonomous Driving. In:vol. CHI 2018 Workshop - Interacting with Autonomous Vehicles: Learning from other Domains (2018).

Please refer to the beginning of this thesis for a detailed statement of collaboration.



Figure 1.1: Sensor setup on the highway.

This thesis is written as part of the smart infrastructure project Proactive Video-Based Use of Telecommunication Technologies in Innovative Highway Scenarios (Providentia) [120]. The project duration was three years, lasted from December 2016 until December 2019 and it was funded by the Federal Ministry of Transportation and Digital Infrastructure (BMVI, Budget: 7.67 Million Euro). In the course of the project, a sensor system was built on poles and sign gantries along the A9 highway in Munich, Germany. With this sensor system, traffic on a stretch of the highway is tracked. The main vision of this project, was to provide a far-reaching view of the highway to the users, track and predict traffic flow and connect vehicles with the infrastructure.

1.4.1 Technical Description

Providentia's sensor system consists of several measurement points. One measurement point on a pole consists of four radars and four cameras to cover far and near range tracking in both directions (see Figure 1.1 and Figure 1.2). Data from the individual sensors is transmitted to data fusion units at the gantry. There, the signals are fused and the result from this data fusion is send to the system's backend. The results from all gantries that are included in the system, are then fused, resulting in an estimation of the vehicles' velocities, positions, vehicle type, height and length. With camera and radar data, it is possible to detect turn signals, obstacles, construction sites, as well as occurring traffic scenarios. With the information from the data fusion, a prediction can be determined of future driving trajectories, traffic flow and upcoming traffic scenarios. All this information can be included in a digital twin of the highway, representing digitally the physical infrastructure. Within the project, two digital twins are created. One is a digital twin that contains the calculated results from the data fusion and prediction that can then be transmitted to vehicles to extend their perception of highway objects. The other digital twin visualizes the simulated traffic on the highway which can be observed by human users.

1.4.2 Use Cases and Scenarios

The traffic behavior predicted by Providentia can be transmitted to different user groups, for example, drivers of highly automated vehicles, the operators of the highway and teleoperators. Autonomous vehicles can use the infrastructure information to improve the driving performance. Drivers can inform themselves of the route by retrieving information from the infrastructure sensors. Thus, the infrastructure data can extend the user's knowledge of the highway situation. This combination of vehicles and smart infrastructure ensures a far-reaching view which informs traffic participants early on about critical situations or situations that are very likely to happen. The scenarios that can be detected by smart infrastructure are manifold, among others emergency braking or accidents. Also, vehicles with high speed or vehicles that are obscured can be detected.

In the following, two examples for warnings that are enabled by smart infrastructure, are given. First, if there is an abruptly braking vehicle further ahead, it can result in an accident or braking cascade. In case of a braking vehicle three vehicles ahead of the driver, the scenario would just be detectable by connected vehicles or infrastructure sensors. Then the sensors need to transmit this information to shorten the reaction time, and thus shorten the braking time of the driver or, in case of an autonomous driving vehicle, braking time of the vehicle. In this scenario, the smart infrastructure can prevent accidents and braking cascades by informing all vehicles and drivers about the events. The resulting effect is that traffic jams can be predicted and the driver can react to this information by changing the route.

In the second example, the driver can be warned by the smart infrastructure of an accident that is not yet well visible in case of bad weather on the highway, These are just some examples and use cases for drivers that are imaginable with smart infrastructure, others are explored in Chapter 3.

Vehicles that are connected to smart infrastructure, can react on traffic events earlier, adapt an ecofriendly driving style, improve the navigation or connect with other vehicles which all leads to safer



Figure 1.2: Picture of radars on the highway tracking the traffic. (Credit to Maximilian Schnettler and Philipp Quentin)

and more efficient driving⁵. By transmitting information from the infrastructure to the connected vehicle, the sensor data of this vehicle can be enriched with environment information from the infrastructure. The connected vehicle itself can then also send information of its own surroundings to the connected infrastructure which further adds to the data that the infrastructure collects on its own. The data that is provided by Providentia can not be captured by vehicle sensors alone. For example, autonomous trucks that are connected to other trucks via sensors can drive in a platoon on the highway. Infrastructure sensors provide a way to add sensor information to, and control the platoons, such as the distances between the trucks. With optimized traffic flow information, these platoons can get information about the route ahead to improve their route planning. Current projects that research vehicle platooning have the goal to maintain a constant distance between vehicles or perform evasive maneuvers such as emergency braking [19]. Through connected driving, traffic flow can be predicted and controlled to ensure smooth driving. Furthermore, to make driving more comfortable, data can also be collected to model realistic human driving behavior in traffic. This model can then be used to test automated vehicles and verify the autonomous driving algorithms and also make driving more comfortable. To sum up, smart infrastructure has much potential which has not yet been explored thoroughly.

1.5 Thesis Structure

After we introduced this thesis in Chapter 1, we present the overall structure of this thesis. This thesis is organized in seven chapters.

Chapter 2: Related Work: Designing for Autonomous Vehicles in Smart Infrastructures We give an overview of the related work and the resulting implications on this thesis. This chapter is

⁵ https://www.kassel.de/buerger/verkehr_und_mobilitaet/verkehrsprojekte/veronika.php, last accessed May 2020

Introduction

divided in six sections. Section 2.1 presents the most important insights in aspects of autonomous driving. Section 2.2 provides insights into smart infrastructure. Visualizations that are presented in related work with a focus on out-of-view visualizations is presented in Section 2.3. Since augmented reality is important for further studies in this thesis, we present related work in Section 2.4. The related work for design spaces is presented in Section 2.5. In Section 2.6, we summarize related work for explanations of artificial intelligence. Finally in Section 2.7 we discuss the gained insights. The thesis is divided in four main chapters:

Chapter 3: Stakeholder and Use Case Evaluation of Smart Infrastructure We conducted interviews and focus groups with infrastructure and autonomous driving experts. From the derived insights user groups and use cases are identified, which refine the users and use cases for this thesis. In Section 3.1 the conducted focus group and interviews are described that lead to use cases for drivers of connected vehicles. Section 3.2 focuses on emergency services and the conducted interview to derive use cases. In Section 3.3, we present a focus group with operators. At last, in Section 3.4 we summarize the lessons learned from the previous sections.

Chapter 4: Design Space for In-Car 3D Augmented Reality Applications In the next chapter, use cases were gathered for using, among others, infrastructure information for in-car augmented reality applications in 3D. With two focus groups we derive use cases that result in a use case set. This use case set resulted in a design space to enable designing for novel in-car applications.

Chapter 5: Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality Third, we consider visualizing information that is outside of the view of the driver and the different methods to present them. Visualizations in 2D, 3D, stereoscopic 3D and AR were implemented and tested in user studies. Those visualizations showed warnings for upcoming critical situations and prepared drivers for a take over request (TOR). With those visualizations the situation awareness was improved and the efficiency of the visualizations was evaluated. In Section 5.1 we explore how visualizations can be presented to drivers. Section 5.2 introduces a novel visualization of the highway and how this can improve the situation awareness of drivers. Visualizations that give hints on objects outside the view of the driver are presented in Section 5.3. The results are summarized in Section 5.4.

Chapter 6: Identifying the Needs of Drivers in Unexpected Driving Situations At last, we studied unexpected driving behavior and how explanations help the driver to understand the cause of it. Through three studies, we develop an understanding of how unexpected driving behavior of vehicles that might receive smart infrastructure information affects drivers and what kind of explanation should be provided to them to enhance their situation awareness. The chapter is divided into two sections, the first, Section 6.1, explores the needs of drivers in unexpected driving situations, and Section 6.2, presents a way to explain unexpected situations to drivers. These sections lead to guidelines which are further elaborated in Section 6.3.

Chapter 7: Conclusion Finally we summarize the results of this thesis and give an outlook on the future research direction in the conclusion.

2

Related Work Designing for Autonomous Vehicles in Smart Infrastructures

Research about autonomous vehicles is broad and touches many topics: hardware, software, sensors, infrastructure, communication or also the interaction between vehicle and driver. In this thesis we focus on the latter aspect of the interaction, between vehicles and drivers and the additional connection to the infrastructure. In previous work, these aspects were covered in different extents. In the following insight about the research context of this thesis is given, discussing autonomous vehicles, the interaction and relevant measurement factors. Then smart infrastructure and connected vehicles and the interaction with traffic participants is covered. Finally, insights about explaining decisions and reasons of intelligent systems to the user are given.

2.1 Autonomous Driving

To gain knowledge what autonomous vehicles are and how they process information, we first give an introduction to test fields, the level of automation and the main hardware and software components that are required to come to a driving decision. Over the last years, the vision of having a fully autonomous vehicle for the daily commute, started to become reality. There are several pilot cities and states in which autonomous driving is tested, such as Pittsburgh¹ or Munich ². Nevertheless, some claim that fully autonomous vehicles will not be reached for end users [191].

The strategy of the OEMs (Original Equipment Manufacturer) differ in the steps to reach fully autonomous vehicle. Some, such as Waymo³ or Lyft⁴ started with a fleet of autonomous vehicles that are just observed by an operator. Others, such as BMW⁵ or Tesla⁶ implement autonomous functions stepwise. First, they implement driver assistance functions. Over time they enhance and extend these functions until fully autonomous driving function is reached. Nevertheless, a study by Banks et al. [16] showed some problems with the semi-autonomous level of a Tesla, among others overtrust or mode error. The vehicles' driving abilities are standardized by the Society of Automotive Engineers (SAE) Level [130]:

[•] Level 0: No Automation, the driver drives completely manually

¹ https://pittsburghpa.gov/domi/autonomous-vehicles, last access May 2020

² https://www.bmvi.de/SharedDocs/EN/Documents/DG/projects-on-digital-test-bed-selection.pdf? __blob=publicationFile, last access May 2020

³ https://waymo.com/, last access May 2020

⁴ https://self-driving.lyft.com/, last access May 2020

⁵ https://www.bmw.com/en/stories/automotive-life/autonomous-driving.amp.html, last access May 2020

⁶ https://www.tesla.com/support/autopilot, last access May 2020

Related Work: Designing for Autonomous Vehicles in Smart Infrastructures

- Level 1: Driver Assistance, a driver assistance system supports the driver either when steering or accelerating/decelerating
- Level 2: Partial Automation, the system performs the steering and acceleration/deceleration, but the driver monitors everything
- Level 3: Conditional Automation, the vehicle performs in some driving modes all driving tasks, but the driver is still the fallback in a request to intervene
- Level 4: High Automation, in some driving modes, the vehicle is completely in control of the driving task
- Level 5: Full Automation, the vehicle is in full control of the vehicle's dynamic driving behavior in any condition

These levels classify the abilities of the autonomous vehicle. In this thesis, primarily levels 3-5 are considered, in which the driver either still needs to take over control of the vehicle or is driving fully autonomously. Since smart infrastructure aims to connect highly automated vehicles, the use cases of smart infrastructure are primarily covered with level 3-5 autonomous driving.

Today's highly automated vehicles rely on several sensors to observe the environment [82]. Autonomous vehicles use radar, lidar, GPS, camera and ultrasonic sensors to sense the outside world [82]. The overall characteristics of those sensors are accuracy, resolution, sensitivity, dynamic range, perspective, active sensor/passive sensor, timescale and output interface. With the accuracy of a sensor, the error between the true value and the sensor's reported measurement is described. The resolution is the minimum variance between two measurements. The timescale is the refresh rate of the sensor and the frequency of the measurement bandwidth time [277]. While cameras passively capture the reflected light of the environment, depending on the environmental conditions, lidar illuminates a target with pulsed laser light [82]. In this manner, a wide range of materials or structures can be detected with a high resolution, among others non-metallic objects, ice or road structures such as intersections [52, 315]. With radar, the range and velocity information of objects can be derived during daytime or nighttime and it can even collect data in humid conditions such as clouds [52]. The position of a vehicle can also be gathered with a GPS signal. Wing et al. [302] state that GPS performance varies according to the environment. Amani et al mention that the received position of neighboring vehicles or Road Side Units (RSU) improve the accuracy of the position of a target vehicle [5].

Even though road traffic is governed by rules, it is a chaotic system that includes many situations in which unambiguous rules can not be determined. In those situations the communication between drivers and other road users follows informal rules to guide the traffic [194]. Autonomous vehicles need to learn not just formal traffic rules, but also behavior conventions on the road. The vehicle gains situation awareness (SA) through the processing of gathered information. Reasoning on machine level can be done, for example by ontologies, which evaluate the context of a situation [11]. Golestan et al. [94] provide a taxonomy for situation awareness in connected vehicles. According to them, the main methods to get situation awareness and predict upcoming traffic scenarios are time, machine learning (ML), game theory (GT) and Probabilistic Graphical Models (PGMs). In connected vehicles, the individual situation awareness can be combined to a shared awareness of all entities involved. The last component included in their taxonomy is the interaction with human operators, to refine and evaluate the whole system [94]. From the presented related work, we narrowed down the automation level of this thesis. Furthermore, we gained the knowledge that autonomous vehicles process sensor information that can have different reliance, depending on the weather condition or daytime. The reasoning that determines the driving decision is based on complex algorithms. With the presented research in autonomous vehicles, we can build assumptions of the knowledge of autonomous vehicles of their surrounding, which will be relevant in the coming chapters that deal with the information interface of autonomous vehicles.

2.1.1 Situation Awareness of Drivers

The following section focuses on situation awareness while driving, to give an overview of the measurement methods and related work in the field. Situation awareness is in the following chapters an evaluation method of implemented visualizations. First, we define situation awareness and introduce the three levels of situation awareness by Endsley et al. [75] and the three levels of cognitive processing by Gugerty et al. [102]. Then we discuss how situation awareness can be measured and related work that assess the correlation between situation awareness and workload, trust and behavior measures.

Driving requires fast comprehension of the driving situation, traffic and environment. Situation awareness generally describes a person's ability to correctly asses a situation as a whole [226]. Maintaining situation awareness while driving requires perceptual and cognitive processes, such as event comprehension or task management [102]. Poor situation awareness often leads to poor driving performance [103, 272]. With a call for designing for situation awareness, Walker et al. [284] rise awareness of the importance of good vehicle design to reduce performance decrements due to lack of situation awareness. Previous work defined three levels that describe a person's situation awareness in a certain situation [75, 102]. The three levels of situation awareness are described by Endsley [75] as the perception of elements in a dynamic environment (Level 1), the comprehension of these perceived elements (Level 2) and the ability to anticipate the status of these elements in the near future (Level 3) [75]. While Endsley considers situation awareness rather as a conscious act, Gugerty et al. [102] argues that situation awareness is developed consciously and unconsciously. The author defines three levels of cognitive processing as "1) automatic, pre-attentive processes that occur unconsciously and place almost no demands on cognitive resources; 2) recognition-primed decision processes that may be conscious for brief periods (<1 s) and place few demands on cognitive resources; and 3) conscious, controlled processes that place heavy demands on cognitive resources" [102]. While driving is in some aspects performed unconsciously, in certain situations a conscious decision must be made. During a lane change, for example, for a short moment, conscious awareness is required. If the workload of the task increases, the driving decision is rather on a conscious process level [102].

With the levels described by Endsley [75], it is possible to assess the situation awareness of participants in a structured, comparable manner. By assessing if just the immediate environment is noticed or if a participant can also draw conclusions from the current traffic maneuver, the participant's improvement of perception can be measured. Many researchers use situation awareness to rate a take over request (e.g. [47, 86, 243]). In a hazardous situation it appears that drivers consciously assess the situation [127]. Horswill et al. [127] discuss that drivers that perceive hazards well, use a dynamic mental model to search for hazards actively. On an active level, a mental model, which is formed of the driving environment, is used to predict the upcoming traffic situation. Situation awareness is difficult to measure [102], but there are many measurement methods [284]. The Situation Awareness Global Assessment Technique (SAGAT) [72], which is used in studies to evaluate situation awareness, blacks out the scenario at arbitrary times. Then the participants are asked questions, aiming for the three level of situation awareness, in which the accuracy of the answer is used to determine the driver's situation awareness level. Another method is using eye tracking, to have an indication of the region and object, a participant looked at during a study. Sirkin et al. [257] present the method "Daze" that measures situation awareness while driving, without halting a situation, in-situ and in real-time conditions. Without halting a situation and during the study, situation awareness in a driving context can be evaluated without being affected of the evaluation thereof. With respect to smart infrastructure, connected vehicles or highly automated vehicles, situation awareness requires a look-ahead on situations that are not necessarily in close proximity of the driver's vehicle.

While driving as a passenger, situation awareness is lower compared to being the driver, when one has more active control over the driving task. Also, when the workload of the driver increases, the percentage of recalled vehicles in a traffic situation decreases [103], which indicates a lower situation awareness. Gugerty [103] argues that it could be a result from the driver focusing the cognitive load on a subset of vehicles. From this subset of vehicles, most were nearby, in sight of the driver, not in the blind spot and apparently hazardous, which motivates a higher attention on the vehicles [103].

The driver can be enabled to acquire high situation awareness, by presenting information about the situation and an instruction to react on the predicted traffic situation. This increases trust in automation [222]. Petersen et al. [222] argue that the different level of SA need to be included in driver assistance systems. For low SA levels, drivers are provided with status updates, whereas higher SA level provide projections about future events. Interestingly, the difference of providing low SA information ("Stopped vehicle ahead") compared to high SA information, in which the low SA information is combined with instructional information (e.g. "Take control now"), is insignificant on the trust assessment, but resulted in clear differences regarding behavioral measures. It can be argued that drivers think that their own judgment was sufficient [222]. From related work we gathered knowledge in how situation awareness is defined and that it can be measured by three SA levels. While some researchers measure situation awareness with eye tracking, in this thesis we assess SA with the three level of SA.

2.1.2 Take Over Request (TOR)

Since we narrowed down the automation level of this thesis to SAE level 3-5, take over requests are relevant for the transition of control. The control is shifted from manual driving, in which the driver is in control of the driving task to autonomous driving, in which the vehicle is fully in control. To gain knowledge how the control transition can be realized, we first give an introduction to TOR to understand what it is, then we give an overview of the times in TORs and then the effect of a TOR on the workload of drivers. Further, we discuss modalities that are used for TORs, and how context can be communicated during TORs. SAE level 3 and 4 of autonomous driving [130] require the driver to take over the driving task in case of system failures or system limitations. Take over requests have therefore been heavily researched in recent years. Flemisch et al. [84] described the cooperation of an automated system with a human as a system that can "lead to dynamic changes, e.g., of qualities like authority, ability, responsibility and control between the actors". TORs also exist in other domains, for example in aviation when the pilot needs to take over control of the plane from the autonomous

system. Funk et al. [90] identified problems relating automation issues for pilots. They reported that in case of monitoring automation for a long period of time, the pilots' perception and cognition is insufficient and errors resulting from this are likely. Wan et al. [285] studied the effects of the lead time prior to a TOR. They found the highest take over performance with a lead time of 10-60s that among others lowers the crash rate and time-to-collision (TTC). Nevertheless, with 15-60s lead time the driver's acceptance was higher. Other researchers provide a time of 7 - 10s that a driver needs to take over the vehicle and that the safety increases with more time [93, 197, 203]. Nevertheless, Kim et al. [147] state that in their experiments the reaction times of participants experiencing different take over events were significantly different. Therefore, there is no single TOR time that is adequate for all TOR events and human factors should be considered [147].

In case of high cognitive and visual loads, results indicate that TOR behavior is affected by it. A visual distraction in automated driving could affect the response time for a TOR. Choi et al. [47] experienced a decreased response time, but an increased steering angle variance after a lane crossing. They argue that a low situation awareness concerning the surroundings of the vehicle may have lead to the poor task performance in keeping the vehicle steady. A high cognitive load, tested by a N-back task affected the driving performance similarly, but in a different time course [47].

The effectiveness of non-visual warnings combined with visual warnings for TOR was stressed by Blanco et al. [22]. A combination of a visual alert and a haptic seat alert resulted in the fastest take over reaction. In case that drivers did not immediately need to take over control of the vehicle, their experiments showed that the drivers would delay the TOR until they needed to take over full control of the vehicle [22].

Since this thesis concentrates on TOR when infrastructure information is available, the TOR can benefit from the extra time and information that is provided by the look ahead on a possible dangerous situation. The extra time is gained by informing drivers earlier of a critical situation. When more time is given to the participant, situation awareness could be increased in reference to the upcoming traffic situation. Epple et al. [78] investigated a two step approach in which the driver was primed of a TOR before being alerted. Mostly drivers took over control of the vehicle before the alert step was initiated [78]. Eriksson et al. [80] also found that in case of non-urgent TORs, drivers took on average 2.5s longer to resume control of the vehicle when being engaged in a secondary task. Other research found that with a visual warning of a TOR, drivers failed to regain control of the vehicle while engaged in smartphone interaction or phone calls [313].

Those results provide similar insights as research by Bazilinskyy et al. [17] that queries the modality preferences for different level of urgency when taking over control of the vehicle. TOR that are realized visually are preferred in low-urgency scenarios compared to vibrations. Verbal messages were preferred and led to faster reaction times compared to abstract sounds [17, 86] and female voices were favored compared to male voices. In highly urgent scenarios, multimodal TORs were preferred [17].

In a study in which TORs are performed repeatedly, Roche et al. found indications that sounds alerting drivers to take over control might be preferred to TOR with visual-auditory modality [243]. This is in line with the research by Naujoks et al. [209] who showed that it takes drivers longer to put the hands back on the steering wheel with visual alerts compared to visual-auditory alerts. Other research mentions lead times of 5-8s for faster reaction and optimal effectiveness, when a verbal warning with the cause of the TOR was uttered [286]. The advantage of a verbal TOR over text combined with a tone was also identified by Epple et al. [78].

Communicating context information to the driver in case of a TOR, such as the location of a road block that needs to be avoided, was perceived as helpful and led to shorter reaction times and safer driving. Dynamic cues, indicating the direction of driving appears to create frustration, but are less demanding than static context cues [27]. When being primed with the appropriate action that needs to be taken, drivers were faster to take over control of the vehicle and had thus a higher TTC. By tracking the eye gaze it was found that drivers still observed the environment to check if the vehicle's driving decision is correct [28]. When communicating the context of a take over request, simplicity is key. Politis et al. [228] gave drivers either a simple countdown until the control needed to be resumed from the vehicle, a dialog in which the drivers repeated the information uttered by the system, a dialog in which the participant answered questions posed by the system or a multimodal interface. From all interfaces the countdown was the best with respect to TTC, Usability and Acceptance. Providing drivers with predictive warnings could also lead to negative effects of behavioral adaptation such as an increased intensity in secondary tasks, increased maximum speed or a decrease of minimum time-to-collision [210].

Lu et al. [183] subdivide transitions in which control shifts in four cases: driver-initiated driver control, driver-initiated automation control, automation-initiated driver control and automation-initiated automation control [183].

Parasuraman et al. [219] simplified the components by information processing and cognitive psychologists to four stages in a model of human information processing: information acquisition, information analysis, decision/ action selection and action implementation. Those stages mirror the input functions on which automation can be applied. Systems can include all four classes of automation application on different levels, from high to low automation [219]. Highly complex system such as autonomous driving systems include all four classes on a high level. Eriksson et al. [79] took those stages into account when designing a HMI that primes drivers of TOR. They increased the information acquisition by highlighting obstacles and hoped to improve the decision making process of participants by it. But the highlighting resulted in unnecessary braking. In this thesis, the control transitions were straightforward, initiated from the vehicle with the driver being in control of the vehicle afterwards. From the discussion of the related work, we assume that smart infrastructure information should be presented more than 10 s before the driver needs to drive manually again to call it early take over preparation. Using visualizations are preferred in low-urgency scenarios, which is the case when informing drivers early of an upcoming critical situation. Context cues are helpful for drivers in TOR situations. Overall, we derive many insights in the design considerations that are relevant for this thesis.

2.1.3 Trust in Automation

The consideration of trust is relevant for the acceptance in a system. Therefore, we introduce the relevance of trust in automation, especially in autonomous vehicles by first giving an overview why trust is important to consider, and then why we should not aim for undifferentiated trust in autonomous vehicles. At last we give examples of related work that research how trust can be influenced. When interacting with technology some level of trust in the decision, result or consequence of the use of the technology is build. Choi et al. [48] state that the adoption of autonomous vehicles can be predicted by the level of trust in them. Especially with the increasing level of personalization and including technology in every aspect of life, trust is an important aspect to consider. The understanding of the involved person for how the context affects the automation's ability also determines the appropriate trust [166]. Khastgir et al. [145] defined two forms of trust in the automotive context: Trust in the system and trust with the system. The latter is defined under the aspect that the user is aware of the system's limitations and thus is able to adapt the own "usage to overcome the limitations of the system". The goal of building trust is not just increasing it, but also calibrating trust, meaning having an adequate level of trust in automation. Thus, problems such as over- or under-trusting a system can be avoided [218].

One example of over-trusting automated systems, without having an adequate understanding of the systems' functions is the deadly accident of a Tesla driver, who drove assisted, but did not pay attention to a truck on the road 7 . Under-trusting an autonomous vehicle may lead to the driver taking back control from the vehicle unnecessarily and thus not using the vehicle to its full potential. Parasuraman et al. [218] phrased it as disuse of a system, so the system is not used even though it is capable to handle the situation. While trust has been defined as a belief, attitude or behavioral state of vulnerability [57, 123], Hoff et al. state it should be viewed as a mental state. Feelings of trust can vary with new experiences or information about the system or the ability of the operator to compare manual to automated performance [123]. Hoff et al. [123] describe three layers of human-automation trust and include dispositional, situational and learned trust in a model. Dispositional trust reflects the constant tendency of an individual to trust in automation. The external environment and the context-dependent characteristics of the operator are two sources for situational trust, which depends on the specific context of an interaction. Learned trust derives itself from the experience with the system [123]. Next to people's values and attitudes, emotions and moods are feelings that Jones et al. [138] relate to the experience of trust. When meeting new persons, people tend to build an impression of trustworthiness based on their feelings to the new person. When behavior is witnessed that is not in accordance to one's values, negative feelings may arise [138]. Hoff et al. propose design recommendations to create trustworthy automation. The five categories include appearance/anthropomorphism, ease-of-use, communication style, transparency/feedback and level of control [123].

Khastgir et al. [145] researched the influencing factors of trust. They provided the participants with different levels of knowledge about the system by reporting the true capabilities and limitations of the automated system, compared to no knowledge. The information was given to the participant in a lay-man language to ensure higher level understanding and to build on the existing mental model of the participant. With the knowledge about the system the trust of the participant increased, but in some cases the participants showed some signs of under-trusting the system. This was the case in very low capability and too much knowledge about the system's limitations and capabilities. Providing the autonomous vehicle with anthropomorphism features such as name, gender and voice seems to increase trust in the vehicle [290]. Lee et al. [166] gave five guidelines to increase trust by providing context, among others to reveal the context of the situations and how past performance depends on situations [166]. We can conclude that an interface should aim for a proper calibration of trust, and to communicate the abilities and limitations of a system honestly. Trust can be increased by several factors, such as anthropomorphism or context information.

⁷ https://www.tesla.com/blog/tragic-loss, last access May 2020

2.2 Smart Infrastructure and Connected Vehicles

To get an understanding of the current state of smart infrastructure we introduce several research projects and the related research focus. Since vehicle-to-vehicle can provide similar information as smart infrastructure if sufficient vehicles are connected among each other, we introduce relevant related work. Some applications that drivers could use would not work without 5G, fusion and prediction, therefore we provide an overview about these topics in the end of this section. There is an increasing research interest in connecting traffic participants digitally with the surrounding infrastructure [292]. The expected benefits include, among others, improved safety, driver comfort, efficient and anticipatory driving and reduced pollution⁸[120, 121, 152, 216]. Using sensors in infrastructure is considered for cities, urban environments, connected vehicles and other traffic participants. While in recent years the term smart infrastructure was used more prominently than intelligent infrastructure, both terms are used interchangeably [216]. Aktan et al. [4] defined it as a system that is able to "(a) sense its loading environment, as well as its own responses and any ongoing deterioration and damage; (b) reason by assessing its condition, health, capacity and performance needs and the actual performance that is being delivered; (c) communicate through proper interfaces with other components and systems, including human managers; (d) learn from experience as well as by interfacing with humans for heuristic and mechanistic knowledge; (e) decide and take action for alerting officials, diverting users, structural control, self-repair, closure etc." [4]. There is an increase of research in smart cities that use sensors to observe and control different sectors within the city. For example, a sewer system monitor is implemented in Indiana in the United States or an advanced analytics system predicts the location of future incidents, used by the Nashville Fire Department and social worker prioritize cases according to antisocial⁹. The so called "City Brain of Hangzhou"¹⁰, created by Alibaba, gathers information through cameras or GPS data on the location of vehicles and buses. Alibaba claims that the daily accident reporting increased and the response time decreased. The average travel time was reduced by three minutes with automated traffic signal control¹¹.

Nevertheless, not just installed sensors are used to observe the infrastructure. Other projects explore the potential of using drones to detect the traffic movement [151, 254]. Those examples show that smart cities explore undiscovered opportunities that could enhance the quality of living through shorter travel times or less congestion. Research projects such as *AutoNet2030* or *Autopilot*¹² also explore the potential of use cases, e.g. the aspects of co-operative driving of automated vehicles [54]. The idea of the projects *Diginet-PS*¹³, *KI4LSA*¹⁴ or *VERONIKA*¹⁵ is to use the existing infrastructure, such as traffic lights, and extend it with sensors. Those track the traffic in real time and thus improve the traffic flow, reduce noise and minimize pollution. Additionally, free parking spaces are identified

⁸ https://www.kassel.de/buerger/verkehr_und_mobilitaet/verkehrsprojekte/veronika.php, last access April 2020

⁹ https://www.ft.com/content/140ae3f0-1b6f-11ea-81f0-0c253907d3e0, last access May 2020

¹⁰ https://www.alibabacloud.com/de/solutions/intelligence-brain/city, last access May 2020

¹¹ https://edition.cnn.com/2019/01/15/tech/alibaba-city-brain-hangzhou/index.html, last access May 2020

¹² https://autopilot-project.eu/, last access May 2020

¹³ https://diginet-ps.de/, last access May 2020

¹⁴ https://www.iosb.fraunhofer.de/servlet/is/105505/, last access May 2020

¹⁵ https://www.kassel.de/buerger/verkehr_und_mobilitaet/verkehrsprojekte/veronika.php, last access May 2020

and the upcoming traffic flow can be predicted. In case of *VERONIKA*, emergency vehicles receive infrastructure data. Those smart infrastructure projects share the idea of improving traffic behavior by tracking the vehicle's behavior. Another large german infrastructure project is the *UR:BAN* project¹⁶. This project aimed to develop traffic management systems and driving assistance systems in the city.

The just mentioned research projects show that there is an increasing interest in connecting vehicles with each other and the infrastructure. The idea of gathering data from remote traffic data acquisition and providing real-time traffic forecasts and route guidance is not new and similar ideas were presented by Xu et al. [310]. Smart infrastructure connects the infrastructure with vehicles, nevertheless, the vehicle can get its information through other means as well. Vehicle-to-everything (V2X) means the connection between vehicles to any other entity that could affect the vehicles. This term aggregates vehicle-to-infrastructure (V2I), vehicle-to-vehicle (V2V), vehicle-to-network (V2N) and vehicle-to-pedestrian (V2P) connections [181, 182]. This communication and traffic infrastructure system, could extend the perception range of individual vehicles [164, 217] through cooperative sensing [121]. V2X is also investigated by Hobert et al. to identify potentials and core functionalities for cooperative autonomous driving. Within their work, vehicles cooperated with a V2X connection and infrastructure on intersections was taken into account [121].

The project ConVex¹⁷ defined some use cases for connected vehicles. Those included forward collision warning, a warning to overtake, a blindspot warning or the information to follow other vehicles with the same destination. This can optimize the taken route. They also consider transmitting personal information to other vehicles. For example, in case of a connected pedestrian, the transmitted information could contain, if the person is a child, adult or has a handicap, which could result in a more careful driving behavior. Gerla et al. [92] and Lu et al. [182] describe the Internet of Vehicle (IoV), a vehicular network of autonomous vehicles. The members of this Internet of Vehicles upload their sensor data, such as GPS location or road condition to the cloud. Human control is removed, which enables a smoother traffic flow. Flexibility in critical and unforeseen events must be ensured as well as safety against malicious attacks. The data processing and forwarding can be designed in different ways, there can be local information, containing warnings about a certain corner, but also information with a longer explicit lifetime. This flexibility needs to be integrated in the vehicular cloud. Some long time information should be kept in the vehicle data itself, instead of the cloud, to ensure scalability. Collaboration will be one strong asset, since data is provided to every connected vehicle without considering the party that provides the data. With this open access, the cloud is just as good as the willingness of every participant to provide their data to all other participants. On the driver side, this IoV could provide photos or videos of traffic scenes, for example, in case of an accident. Since the vehicles travel with high speed and optimized distances, the driver needs to be prepared to intervene and decide quickly [92].

Concerning the integration of smart infrastructure in the vehicle, Human Horizon presented the HiPhi 1 vehicle that is supposed to be connected to the smart city via 5G¹⁸¹⁹. Future 5th generation (5G) cellular networks promise disruptive changes for, among others, smarter devices. Device-to-device (D2D) connectivity, which includes the interaction between users via augmented reality or HD video streaming, requires local and non-local content and constraints of low-latency and high data rate [24].

¹⁶ http://urban-online.org, last access May 2020

¹⁷ https://convex-project.de/, last access April 2020

¹⁸ https://www.human-horizons.com/home/smart_car, last access May 2020

¹⁹ https://www.topgear.com/car-news/future-tech/human-horizons-hiphi-1, last access May 2020

5G is said to be an enabler for autonomous driving, therefore, the automotive vertical market is a key driver for these systems. The challenge that developers of autonomous vehicles face is the small distance and high speeds (up to 200 km/h) on the road [40]. Campolo et al. mention also a data rate of 10 Mb/s(downlink/downlink) with a latency of 1 ms [40].

The single sensor data is fused with sensor data from the network. This results in data about the velocity, location and potentially personal data of the connected user. With the data about the environment, map and user, a prediction about the next movement and the resulting consequences can be made. With the fused and collected data, a prediction about the further driving trajectory of the vehicles can be calculated [230]. Polychronopoulos et al. [230] optimized in 2007 the path prediction for vehicles and had results of being able to estimate the path for 3-4 seconds. The accuracy was less than 50 cm. Mirus et al. [201] used an approach that takes the surroundings into account. Their results are promising, but the prediction of future driving behavior remains challenging. Since datasets of highway driving consist mainly of straight driving vehicles, the training for lane changes is difficult [201].

The emerging technologies of 5G, autonomous vehicles, augmented reality or vehicle-to-X are intertwined and a combination of these technology could enhance the experience of drivers of autonomous vehicles. Within this thesis the focus is on exploring the use cases, user groups and resulting benefits of smart infrastructure on the user. The main focus is on the highway, since this thesis is set in the problem space of Providentia, a highway infrastructure project. Nevertheless, some use cases can be applied for an urban setting, and some studies are primarily set in the city. Throughout the thesis, we take the ideas of having a look ahead and detecting traffic participants outside the drivers sight, and explore the potential resulting use cases.

2.3 Out-of-View Visualization

Smart infrastructure offers the possibility of providing drivers with information that is outside the view of the driver. To gain an overview of relevant techniques to visualize this information, we first introduce the topic, give then an overview of general visualization schemes and in the end put it in the context of automotive interfaces. Since the development of cheaper sensors, it became more relevant and possible to observe the context of a system [58, 251]. Autonomous vehicles have sensor systems that detect the immediate environment of the driving situation in close proximity of the vehicle. Infrastructure can extend this context awareness even further. Traffic situations that are not necessarily in close proximity of the vehicle, which might need the driver's attention, need to be communicated to the driver. Visualizations and hints on other objects are relevant in various scenarios, spanning from gaming, augmented reality, aviation and automotive applications [10, 141, 314]. In those applications, clues are given to the user of the system to direct the attention in a certain direction [100, 245]. The next level display technologies, such as head-up displays (HUD) allow the presentation of information or warnings within the driver's windshield [109, 240]. In potentially dangerous situations, the attention of the driver shifts from the immediate environment to the dangerous scenario that is not necessarily yet in the view of the driver. Research shows that priming drivers before the actual take over request, results in safer hand-over scenarios [116, 299].

Visualizations aim at making users comfortable and aim for an easy use of interactions [95] There are interface schemes that provide the user with a wider context than just a primary visualization

screen. Some of the more known techniques are overview+detail, focus+context and cue-based techniques [50]. Overview+detail combines a view which gives an overview of a situation with a detailed focus on an aspect of the overview. In the scheme of focus+context, the idea is similar, a contextual view is displayed and a zoom is set on a focus region. One example for this scheme is a fisheye view, in which a certain region is zoomed, but the surroundings and context is shown as well [50]. Cuebased techniques selectively highlight or suppress items within the information space [50]. Known primarily from computer games and aviation, minimaps are similar to environment maps. They combine the initial aspect of focus+context views, by visualizing out-of-view objects with focus regions in a single screen, while at the same time relying on overview+detail elements. Other popular visualization solutions are Wedge, a technique that conveys direction and distance to the user [106] or arrows that point at a point of interest [252]. Gruenefeld et al. [99], used a visualization of out-of-view objects by a radar-like interface in augmented and virtual reality (VR) devices.

Visual warnings are suitable for non-urgent information, complex messages, discrete and continuous information and spatial information [39]. Some visualization schemes are more effective than others to prime users for the intended information, but the intended use needs to be considered when implementing an out-of-view visualization. For the automotive domain, visualizations might help with increasing situation awareness and with getting a faster reaction and better driving performance. Animated stimuli get a faster reaction from the driver and the visualizations help to locate other road users [101, 270]. Peripheral awareness displays can increase safe driving performance by combining a light strip with vibro-tactile seat feedback [156]. While reducing the time the driver spends glancing away from the road with this situation awareness display, simultaneously the situation awareness can be increased. Lenné et al. [168] implemented an advanced warning device to warn drivers about approaching emergency vehicles. The interface consisted of LEDs and an auditory warning, which improved the drivers preparation for the emergency vehicle. This is in line with the work of Popic et al. [231]. They compared a 3D HMI with a 2D HMI to assist driver in anticipatory driving. The differences of the results of the 2D compared to the 3D interface were small, whereas the 3D visualization was also continuous while the 2D visualization prompted new information upon approaching a critical situation. When using stereoscopic 3D (S3D) in the vehicle's dashboard, the combination of color and S3D increases the perceived urgency. Furthermore, it increases user experience [37]. Situations in which the driver needs to be alerted of situations or other traffic participants that are outside the view, are for example intersections, where traffic participants are obscured around corners or generally vehicles or traffic that are obscured by other vehicles or objects. The blind spot can also provoke situations in which drivers do not see other vehicles. With smart infrastructure information, the radius in which other vehicles or objects are detected is extended and therefore there are even more situations out of the line of sight of the driver. Werneke et al. [293] investigated the presentation of collision warnings at intersections. They compared an early warning, in which the possibly critical vehicle is not yet in view to a late warning with the vehicle in plain view of the driver. The early warning is efficient in preventing a collision accident with the critical vehicle. Early on, the drivers drove more carefully by waiting longer and accelerating slower. In a multi step approach, drivers are warned of critical situations in multiple steps. In a first phase, the information phase, drivers are primed of a critical situation. The next step is a pre-warning in which a multimodal approach could be chosen by additionally warning the driver acoustically [237, 303]. A multi step approach showed to be preferable in critical situations, drivers preferred to be informed at 450 m before the critical situation. In a less critical situation drivers preferred 340 m [303].

Related Work: Designing for Autonomous Vehicles in Smart Infrastructures

From the presented related work, we learn that visualizations are beneficial when the goal is to convey complex and non-urgent information [39]. Animated visualizations are more effective to gain the drivers attention and early warnings are efficient in preventing accidents. In the course of this thesis, the aspect of how smart infrastructure information influences the presentation of a TOR visually is considered, therefore, with this overview we discuss how take over requests can be presented to drivers.

2.4 Use of Augmented Reality in Vehicles

In this thesis we use augmented reality to support out-of-view visualization to increase situation awareness. Therefore, it is important to get an understanding of the technology and the advantages, disadvantages and best practices. Even though augmented reality cues for objects that are in view are an important field, the focus for this thesis is rather on augmented reality cues for vehicles and dangers that are outside the view of the driver. Therefore, following section is a short summary of work that gives cues about vehicles and traffic situations in the view of the driver. Possible AR visualization schemes for context information and for informing drivers of possible scenarios outside their periphery can build on existing research of AR cues in the driver's field of view. With regard to this research focus, the last paragraph summarizes the findings of using augmented reality to communicate context or out of view information.

In today's vehicles, augmented reality is used to add virtual objects or information, such as trajectories, speed information, hazards to the view [205] (see Figure 2.1 for exemplary visualizations from related work) Augmented reality is defined by Azuma as any system that combines real and virtual content, is interactive in real time and registered in three dimensions [13]. The displays that can be considered when using augmented reality applications are manifold, ranging from handheld displays, Head-Mounted Display (HMD) or desktop. There are many domains in which research is conducted using some sort of AR display, for example education, medical, interaction or entertainment and gaming [59].

Augmented reality use cases within the vehicle's cockpit are widely under research, with a special focus on acceptance of AR navigation cues [59, 266]. HUDs use AR elements to display velocity and navigation cues. It can also be used to display vehicles' driving behavior and information about the vehicle's sensor perception. While HUDs are constrained to a small area on the windshield, windshield displays (WSD) use the whole windshield to present information.

Application cases range from overlaying a certain area with color information, highlighting the edge of the street, enhancing night vision or highlighting dangerous objects on the road [165, 241]. Combining this with other modalities leads to novel interaction experiences. For example at CES 2016, BMW presented HoloActive Touch [56, 247], a haptic touch interface floating in midair. The interface is projected in the vehicle cockpit by a mirror plate to create the effect of floating objects. So far this interface is a prototype, displaying buttons and no real-time information.

Morrison et al. [204] compared augmented reality cues combined with a physical map to a 2D digital map. They found that combining the physical map with AR features fosters collaboration and discussion. When AR is being used to work in autonomous cars, this insight could help the development of new interactions to give an opportunity to foster collaboration via AR. Pervasive AR that "augments the physical wold digital with information registered in 3D, while being aware of and responsive to
the users context" [98] could be valuable when applied in vehicles. Information overlays for example aim for a context-aware experience [98].

Devices that create mid-air 3D AR objects include HMDs, 3D Tabletops and Holographic projectors. A particular advantage is the high mobility of modern AR HMDs, with integrated natural input possibilities such as speech or gesture (e.g., Hololens²⁰). Previous work on floating 3D objects inside vehicles focuses on communication [155], new input possibilities [247] or dashboard visualization [37, 174].

In Dikmen's [62] study, augmented reality cues highlighting other vehicles, does not increase situation awareness [62]. This is in contrast to Lindemann et al.'s work [173] in which AR overlays significantly improved SA while driving. They dynamically included information about destination distance and time, traffic priority, confidence information, traffic regulations a primary driving panel, navigation, traffic light and high priority information in the WSD interface.

In TOR situations, AR does not influence the reaction times of TOR [159]. Langlois et al. [159] also did not find indications that AR cues increase the behavior of checking the surroundings before taking over even though the awareness for a lane change seems to be improved by AR. This could also be the effect of adding the AR visualizations in the driving scene itself. Another study, also investigating AR visualizations in take over requests, provides the study participants with two augmented reality conditions. The first condition displays in red the area which should be avoided by the driver whereas the second condition shows in green the right lane. Lorenz et al. [179] investigated the influence of the conditions on the take over process. Similar to Langlois et al.'s results, they did not find a significant effect of AR on take over times. The take over behavior of the participants brake harder, which could be explained by the alarming effect of that color. Nevertheless, this take over behavior did increase the lane change errors [179]. The HUD with augmented reality elements is appreciated and preferred compared to augmented reality elements in the instrument cluster [255].

Tönnis et al. [270] explored the difference of visualizing environment information either with a bird's eye view with an indication on an alert position or an arrow pointing at a point of interest. Their results showed that the reaction time was faster for the bird's eye view. The error rate was also higher withe AR HUD. This shows that using AR to indicate objects on the road needs precision and is not always the best solution. Presenting information fast and accurately is efficient when using AR in the HUD [89]. Others used augmented reality cues to improve the left turning behavior [205, 271] or direct attention on roadside hazards [248]

Even though many AR applications already exist in today's vehicles, there are not many concepts that do not use the windshield or dashboard as interaction possibility. We explored new concepts of visualizing a preview of the highway in stereoscopic 3D [299]. Weidner et al. [291] used a similar approach and used different location to provide the participants with representations of the traffic scene and dangerous scenarios. They found that in two conditions those visualizations led to safer TORs compared to TORs in which no visualizations were provided. Furthermore, when compared with perspective displays, stereoscopic 3D performed better with regard to safety in TOR [291]. To summarize the findings, augmented reality is mainly used for navigation in HUDs, WSD or the infotainment display. It is used by overlaying areas with color information [165, 241]. Using HMDs is beneficial due to the high mobility that can be reached. In some cases AR overlays improve situation

²⁰ https://www.microsoft.com/en-us/hololens, last access: May 2020

Related Work: Designing for Autonomous Vehicles in Smart Infrastructures



Figure 2.1: Examples of augmented reality in the literature:A: A hint on a critical situation after a turn by Werneke et al. [293], B: A concept of a stereoscopic 3D solution by Continental-Automotive^{*a*}, C: A projection of a path by Tran et al. [271], D: Several highlighted objects, paths and persons by Park et al. [220]

^a https://www.continental-automotive.com/en-gl/Passenger-Cars/Information-Management/ Display-Solutions, last access April 2020

awareness of drivers [173]. The take over times do not seam to be effected by AR [159], which means that AR cues does not seam to have a negative effect on the performance of drivers during a TOR.

2.5 Design Spaces for Automotive Interfaces

With a structured approach to derive new designs of applications, we can effectively construct novel applications that especially aim for drivers in vehicles that are connected to smart infrastructure. In the following section an introduction to design spaces is given, by first starting with the overall definition. Then the advantages of exploring existing design spaces is shown, followed by giving concrete examples of them in the automotive domain. The importance of design is stressed by Norman [214], who provides principles that focus on the need of users interacting with things. The design process can be described as "iterative problem solving" [14] with designers working on a product, which should be mutable, effective, manageable, learnable, needed and aesthetic. Ultimately, the final de-

sign should understand what the users need and want [6]. But of course, design works also towards meeting the business goals and it should fulfill the usability principles [14]. Gould et al. [96] for example formulated three usability principles: Early Focus on Users and Tasks, Empirical Measurement and Iterative Design. With those principles, users should be included as early as possible in the design process, use prototypes to test a low fidelity prototype and iterate over the designs regularly. To achieve a good design, designers are provided with tools, such as design spaces. According to Westerlund [294] all design solutions, provided that they work, are included in a design space. Design spaces originally provided a cartesian space in which individual designs were mapped with design variables [66]. Design variables are defined by the constraints that are defined by rules, principles, conventions and laws. Therefore the design variables define the possible adjustments that can be made within the design space [97]. Therefore, it can have predictive and analytical properties. Design spaces can be described as "instances of conceptual spaces" and can be set up by possible chosen values that build up quality dimensions [66]. In research, the term "design space" is interpreted varyingly, spanning from conceptual tools to network graphs [66]. The main benefits of design spaces is to provide members of a team, working on the same design, with a communication tool to get a common understanding and idea of the final design. Furthermore, through the representation of the framework it helps them to reason and explore the constraints of the design space [187], considering alternative designs and recognizing trade-offs resulting from decisions [256].

The exploration of existing design spaces holds several advantages. As an example, exploring the space of different designs and deriving a structure that represents the network of those designs is the idea behind design space exploration. Providing designers with such a tool can help to find suitable designs and to find the optimal solution to their problem space [308]. Woodbury et al. [308] propose properties and structures of design spaces to formalize design space exploration. Designs are in general partial, meaning that they can and should be changed by adding and removing components [308]. Design space dimensions are not uncorrelated from each other, and therefore, setting constraints can influence several dimensions at once. For example, setting constraints for interactions may influence display-technology constraints, but also modality settings [66]. Once the design space has been created, it is not final, but thrives on change. Dove et al. [66] discuss the benefits of reflecting on the design space of projects by introducing the concept of design space reflection. They propose to revisit the initial design space by questioning it and considering alternatives. With this procedure, the alternatives can reveal constraints on other dimensions and opens up new areas for investigation. That aids and simplifies the identification of influences of constraints in one design dimension on another dimension. They compare this procedure to the Design Space Analysis (DSA) with the Questions, Option, Criteria (QOC) notation of MacLean et al. [187]. Since there is no strict method or procedure to create a design space, MacLean et al. consider the processes involved in generating DSA. Using the QOC process, an existing design is analyzed by posing questions to understand how and why design decisions have been made. Thus, options represent the alternative answer to those questions and evaluation criteria are used to make a choice among the options [187]. Lim et al. [172] used a design space to filter design dimension by creating a prototype and exploring the yet incomplete design. Throughout this process, the advantage of gathering cumulative knowledge is stressed. Additionally, they identified the method as a "critical thinking guide" that helps in making differentiated choices.

Next we give concrete examples of automotive design spaces. The tool to support the design process was provided in previous work in different forms of design spaces. For example, Müller et al. investigate the required design elements for public displays by exploring the mental models of the user and interaction modalities [206], the design space for shape-changing interfaces was presented by Rasmussen et al. [234] or Card et al. set up an information visualization design space [41]. In the automotive domain, another example is the design space provided by Kern et al. [144]. They present a design space that is focused on manual driving and the classical dashboard input and output modalities. Their main interaction areas included windshield, dashboard, center stack, steering wheel, floor and periphery. In their design space, they map those interaction areas with different input and output modalities, such as hand/foot or audio output. In case of autonomous vehicles, in which the driver could be faced away from the front cockpit, the interaction area that defines today's cockpits is obsolete. Since vehicle cockpit faces a change from a focus on manual driving to a focus on autonomous driving, there are many opportunities to include new technology in the cockpit. This opens the already existing design space and new designs need to be explored. For example, mobile devices could be integrated in the vehicle, adding to the existing interface within the vehicle. This could then have different advantages, from adapting the music to the preferences of the driver to acting as multi-touch input [60]. Since it acts as additional monitor, touch input and output can be combined with existing interfaces, but it can also stand on its own.

Since the introduction of head-up displays in the cockpit, they are used to display information such as navigational cues in augmented reality. Endsley et al. introduce design heuristics regarding human factors, ergonomics and user experience of AR devices [76]. They establish eight design heuristics, among others "Adaptation to user position and motion" or "Fit with user environment and task". Especially the increased dynamic positioning and attention direction is explored. This clearly differs to fixed interfaces that are implemented in today's vehicles. In vehicles, the users are confined to a limited space, nevertheless, in autonomous vehicles the interaction area is likely to be more flexible compared to today's vehicles. In autonomous vehicles, drivers are not obliged to be faced towards the windscreen or have their hands on the steering wheel, forcing them to a confined area of interacting with the HMI. Instead drivers are more flexible in their movements, do not need to have their eyes on the road at all times, but can also be immersed in leisure activities. This trend of introducing augmented reality to everyday activities is also reflected in current research and the resulting design spaces. Those focus on the challenges emerging from autonomous driving and possibilities of integrating AR technology in the vehicle [108, 223]. The work by Tönnis [269] and Häuslschmid et al. [108] investigate head-up and windshield displays and their design dimensions. Different to the design space of the classical vehicle cockpit interior, those design spaces also include for example different dimensionality (2D and 3D Information Presentation), content registration or frames of reference. With regard to design dimensions for 3D spaces and Head-Mounted Displays (HMDs), Ens et al. [77] explored the so called "Ethereal Planes", the design space of planar 2D information spaces in 3D mixed reality environments. They identified three groups Reference Frame, Spatial Manipulation, Spatial Composition with the seven dimensions Perspective, Movability, Proximity, Input Mode, Tangibility, Visibility and Discretization.

In this section, we did an excursus on the definition of design spaces, on the advantages of exploring existing design spaces and on concrete examples of automotive design spaces. From this we learned that design spaces help to explore possible designs for a product or prototype. They are constantly changing and help identifying and refining constraints of concrete designs. Automotive design spaces introduce input and output interaction possibilities [144] or the visualization and features of windshield elements [108]. Those design spaces focus on existing manually driven vehicles in which the driver is still in control of the vehicle.

Since design spaces are set up by exploring existing designs and identifying common design dimensions and constraints, design spaces for emerging technologies do not exist. Augmented reality is used in the vehicle primarily on the windshield for navigation cues [59, 266]. Nevertheless, autonomous vehicles promise to change the vehicle interior. A design space for applications that do not use windshields, windows or the exterior of the vehicle, so for applications that could be very likely be included in future autonomous vehicles, is missing. In this thesis we explore this knowledge gap to be able to introduce new applications and designs, for autonomous vehicles connected to the infrastructure that use augmented reality and are not limited on the windshield or window.

In Chapter 4 we describe the process of deriving a design space for mid-air 3D AR applications inside the vehicle. To derive it we applied a similar approach as Häuslschmid et al. [108]. In contrast to Häuslschmid et al., we aim at a design space for a field that is just emerging, with a limited number of existing concepts, prototypes and consumer products. Therefore, the aim is to identify design dimensions for 3D in-car AR applications in a technology field in which 5G, smart infrastructure and AR applications are on the edge of being in end user products.

2.6 Explainable Smart Systems

Autonomous vehicles are intelligent systems that use artificial intelligence (AI) techniques such as Machine Learning (ML) to percept the environment and make decisions based on those elements.

However, most of these systems are black boxes, it remains unclear to the user of the system, why a particular decision has been made [104]. In this section, we give an overview of explainable AI methods and related research, by first introducing the topic, then provide insights of the benefits of XAI. Next we introduce method on how XAI can be reached and then what kind of explanation other researchers introduced. The dependability of the output on the input or the weight of features often remains unclear to the individual user [190]. This can negatively impact user trust [67, 186] and acceptance of system suggestions [53]. This has raised concerns about fairness, discrimination and opacity [68]. While this individual intelligibility remains important, the focus shifts to the need for understanding algorithms on a macroscopic societal accountability [1]. Fair, accountable and transparent (FAT) algorithms [169] become a legal obligation in europe through the general data protection right (GDPR [235]) and the USACM released a statement on Algorithmic Transparency and Accountability ²². This call for transparency addresses all industries and domains.

With the introduction of explanations for the decision of the intelligent system, this call can be answered by making the system understandable. Explainable Artificial Intelligence (XAI), Transparent AI or Interpretable AI are techniques which focus on increasing the understanding of humans for the intelligent system [1, 65]. Interpretability is the ability of a system to explain or present in understandable terms to individuals [65]. With global interpretability, the whole logic of a model is understood, while with local interpretability, only specific decisions is understandable [104].

According to Adadi et al. [2], using XAI systems provides the required information to 1. justify results, particularly when unexpected decisions are made; 2. immediately identify and correct errors in low criticality situations; 3. continuously improve the model; 4. explain its learned strategy (knowledge) to the user [2].

²²https://www.acm.org/binaries/content/assets/public-policy/2017_usacm_statement_algorithms. pdf, last access April 2020

Related Work: Designing for Autonomous Vehicles in Smart Infrastructures

DARPA (Defense Advanced Research Projects Agency) launched a XAI program [105] that enables users to cooperate with the intelligent system and correct it. However, the field is still working towards finding suitable metrics for assessing the quality of explanations (see [65, 104, 124] for exemplary approaches). For example, Doshi-Velez and Kim [65] describe a taxonomy with for one, *application grounded evaluations*, in which real users use real applications. Then, other metrics are *human-grounded metrics*, in which a simplified task is performed by humans and *functionally-grounded evaluations*, which do not involve humans and use formal definitions of interpretable explanations, such as decision trees. The authors argue that users' expertise in a task plays an important role when designing explanations.

To clarify algorithm decisions, various visualization approaches have been researched. Bojarski et al. analyze a convolutional neural network (CNN) which is trained for autonomous driving by unveiling the learned image features [25]. Other explainability techniques include decision trees, LIME [238], rule lists or saliency map [2]. Nevertheless, even when producing visualizations, some fail in making the AI algorithm understandable [2].

In a concrete example, to provide users a system that is intelligible, Lim et al. [170] provided users with different explanations, among other "What if the factors are different, what would this inference be?". Participants spend significant time with it, when used, since it was interactive and they liked it. Nevertheless, the authors claim that their system is not excessive enough since it was used to the depth that was provided to them with their system. This can result in a call for explanations of deeper contexts instead of shallow application context [170]. Other attempts to explain AI systems include persuasive engagement, which emphasizes to take context and purpose into account to increase trust and satisfaction of the user [260].

Aiming for algorithms that explain AI systems is important to justify the algorithms decision and to provide a fair process. Visualizations are explored to make algorithms understandable.

2.6.1 AI Design and Mental Models

Human-AI interaction is very difficult to design. The challenges designers face, range from the understanding of the AI's capabilities, to AI prototyping until the inclusion of AI experts in the design process. For example, the AI system behavior or potential effects of AI are difficult to anticipate and therefore, to prototype. Other problems are with the interaction and shared control between users and AI. Explaining the behavior of the AI system or the constant evolvement is also challenging. The following section first introduces design processes and approaches. Next we discuss related work that explores mental models and finally we present design guidelines from related work. Yang et al. [311] used a user-centered design process and aligned design challenges related to AI, such as "how to bring a human-centered view to AI". They identified that the challenge of prototyping for AI systems lies in the complexity of the output and the capability uncertainty. According to design complexity, four levels of AI system exist: Probabilistic Systems (Level 1), Adaptive Systems (Level 2), Evolving Probabilistic Systems (Level 3) and Evolving Adaptive Systems (Level 4).

The interaction between users and systems is an extensive field and there are multiple design approaches to form the interaction. Rosson et al. describe a user interaction scenario as a sketch of use, to capture the essence of an interaction design [244]. For scenario-based designs, the focus for a suitable design representation is how people succeed in their work, by using narratives [42].

Scenario-based design methods of XAI focus on the use and anticipation of possible deployment scenarios at an early stage of system development [307]. Wolf [307] aims to change the design process from a less technology- or solution-first perspective, that rather focus on what an AI system is capable of explaining, to one that provides an intimacy with the problem space by asking what type of explanations users need. In explainability scenarios the perspective is on what types of explanations users need while using AI systems, to react on the system's output [307].

To measure the effectiveness of an explanation, Gunning [105] suggests to explore mental models and user satisfaction. The theory of mental models is one of the fundamental theories behind explanations and user perception. Tullio et al. [273] define mental models as a "hypothetical construct defined as a mental representation of a real or imagined situation". We argue that it is important to understand how users perceive intelligent systems to be able to gain their trust [273]. Mental models are always individual, but can be altered by explanations and change of the perception of the user of a system [46]. They are an important part of the theoretical basis for Human-computer interaction (HCI) and they help to understand how people use systems and can provide ways to improve them. Negative behavioral adaptation can occur when drivers have an inaccurate or incomplete mental model of vehicle functions. The behavioral adaptations include neglecting required supervising, when assistance functions are activated [261]. Sullivan et al. [261] provide approaches to improve the behavior through rewarding, driver monitoring, and not letting the driver out of the loop of the driving control task. Design guidelines for AI [7] include 18 criteria during different stages of interaction, e.g. "When wrong" or "During interaction". When wrong, the guidelines state to "Support efficient dismissal", "Support efficient correction" or "Make clear why the system did what it did" [7]. Those design guidelines are important aspects to consider when designing an AI application, especially supporting efficient correction or making clear why the autonomous vehicle did what it did. For autonomous vehicles, in which the scope is to not have the driver involved in the driving task, some design guidelines, such as "support efficient dismissal", described as "Make it easy to dismiss or ignore undesired AI system services" [7] might need some refining.

Concerning autonomous vehicles, Carsten et al. [43] states six crucial HMI design elements: "Provide required understanding of the AV capabilities and status (minimize mode errors). Engender correct calibration of trust. Stimulate appropriate level of attention and intervention. Minimize automation surprises. Provide comfort to the human user, i.e. reduce uncertainty and stress. Be usable." [43]. Since their work focuses on designs that are in current vehicles, the potential of an autonomous vehicle that is connected to smart infrastructure is not taken into account. Similar to that, Lee et al. explored the user needs in autonomous vehicles [167]. They identified in expert interviews an focus group interviews twelve user needs in autonomous vehicles, among others accessibility needs and personalization and customization.

From this section we can derive that the process of designing an interaction interface with an AI application is challenging. Effective explanations take the mental model of users of the application into account, and how the perceive the system to gain their trust. Finally we explored design guidelines, in which the challenges of designing for AI is taken into account.

2.6.2 Perception of Explanations of Reality

Explainable systems touch the field of human interaction researchers and psychology. Formulating explanations understandably while at the same time conveying technical information is a known

Related Work: Designing for Autonomous Vehicles in Smart Infrastructures

problem. Technical advances require the adaptation of explanations in intelligent systems. Therefore, we explore the field, how explanations can be conveyed on a societal level. First, we introduce constructivism. From this we can derive, how situations can be interpreted and finally we discuss the influence and failing of common sense on explanations. One theory on how humans perceive the world is provided by *constructivism*. Constructivism is an approach to solve the problem of knowledge and knowing. The concept states that the thinking person constructs all knowledge on the basis of their own experience. What we construct from experience makes up the only world in which we live in.

Von Glasersfeld [279] says that reality is like a black box, since there is just a hypothetical connection between the own experience and the reality. Since a model tries to depict the black box, a good model is one that has, with the same input, the same output as the black box. It depicts the constraints given by the situation, nevertheless, it is not said that this model is unique and that no other model could produce the same results. To interpret situations and statements of others, we base the own interpretation of that situation on the own experience [279].

From the previous section on situation awareness in vehicles, we derive that especially in safety critical systems such as vehicles, situation awareness is important to quickly comprehend the driving situation. On this comprehension we build an informed driving decision. Logical behavior is therefore essential for autonomous vehicles. People believe the first explanation that is approximately conclusive when confused [287]. Watzlawick [287] discusses that paradoxically, common sense and "logical" behavior tend to fail, while "illogical and unreasonable [actions] succeed in producing a desired change." [288]. Furthermore, trust in intelligent systems can be influenced by placebic explanations [70]. Therefore, programmers need to keep the user needs in mind since they tend to design explainable algorithms without focusing of users [199]. Communication and behavior anticipation guides also the behavior of road user [194].

From the exploration of explanations from the perspective of psychology we gained insights, how explanations can be perceived. The experience of drivers affects how situations are assessed and perceived. In case of confusion, we believe the first explanation that is rudimentary conclusive [287].

2.6.3 Explanations and Technology in Autonomous Vehicles

In the automotive context, different research aims for explaining the highly automated vehicle's behavior. In the following we give examples of related work that investigates, what an explanation should include to increase understanding and situation awareness, and what aspects should be considered when aiming for high usability. Finally, we discuss the impact of uncertainties and predictions on explainability. Koo et al. [148] investigated the content of an explanation message in the automotive domain. Explaining to the driver how a vehicle behaved led to poor driving performance. In contrast, providing a reason for the driving behavior was preferred and led to better driving performance. Providing both information led to safest driving behavior, but increased negative feelings. The explanation was provided verbally [148]. Studying the user experience while driving autonomously, Frison et al. [88] found four factors that are mainly important to keep in mind when designing for UX: the subjective feeling of safety, pleasure of driving, perceived travel time and personalized user interfaces. Ribera et al. [239] puts an emphasis on the importance of the user, the explanation is targeted to and to follow human conversation which is thoroughly studied. Eiband et al. provides a methodology to derive a target mental model, that identifies the common features of the mental model of experts and user [71].

When driving autonomously, sources can provide information of a driving decision of the autonomous vehicle. The sources may be unknown to drivers such as smart infrastructure or V2V communication. This information is transmitted by other means and not just perceived by the autonomous vehicle [119, 212], and therefore is beyond the common observation horizon of a driver. This not just opens up new possibilities to inform the driver about future driving decisions of the vehicle, but also the need to explain driving behavior of the autonomous vehicle based on traffic scenarios out of the perception range of the driver arises. Since most highly automated vehicles nowadays base their driving decision on their own sensor systems [185], the information that also other systems could form the driving decision needs to be communicated to the driver. Then the mental model of the autonomous vehicle can be adjusted to consider the implication of having a new information channel.

Especially when having infrastructure information, the prediction horizon can change and uncertainties play an important role in the driving decision. Providing uncertainty information or information about the vehicle's state can lead to faster reactions and safer driving [18, 107]. When communicating reliability, the driving level and the corresponding secondary task should be taken into account [81]. Especially in take over requests, in which the driver takes over control of the vehicle, it is important that the driver is quickly fully immersed in the driving task. When presented with an explanation of the reason for a TOR, drivers have a stronger feeling of understanding the system [150]. But in the study of Körber et al. [150], the explanation for a TOR had no effect on acceptance and trust. The reasons for a TOR were explained by missing GPS signal, missing lines or roadworks [150]. Providing explanations before an event however, could build more trust [114]. Another approach is presented by Fridman et al. [87], who argues that the human operator can be involved, when two AI systems disagree over a decision to be made. While it is a proof of concept, it could help drivers of oncoming difficult situations [87].

From related work, we derive that the driving performance can be influenced by providing drivers with explanations. When designing for high user experience different aspects need to be considered, such as safety aspects or the driving pleasure. The prediction horizon with smart infrastructure can change.

2.7 Lessons Learned

From related work we acquired examples of how applications can be integrated into the vehicle, how other researchers have realized out-of-view visualizations, how autonomous system behavior is explained and how to approach a design process. Nevertheless, the research focusing on the benefit and the application of smart infrastructure information in the vehicle leaves open research questions. Therefore, the problem space can be explored further from different angles.

The connection of autonomous vehicles to other vehicles and smart infrastructure improves safety. When integrating smart infrastructure in autonomous vehicles, many elements are intertwined and need to work together. The infrastructure information can be added to the information from the own vehicle, but in a sensible manner, depending on the level of automation of the vehicle. Showing context of a driving decision can increase trust and situation awareness [166, 222]. While

Related Work: Designing for Autonomous Vehicles in Smart Infrastructures

TORs without an extended view are time critical, taking back control from the vehicle with smart infrastructure information can change this interaction. TOR performance can be improved when executed with priming drivers and adequate modalities [22, 28, 78]. A TOR can be designed according to the purpose, for example warning or priming drivers, in these situations. In this thesis, most TOR were issued in the studies as early preparation, therefore, the urgency to take over the driving task was low.

In the exploration of related work, we gained an overview of related infrastructure and autonomous driving projects, and gave an introduction to the sensing and decision making elements of autonomous vehicles. A measure of situation awareness can be used to rate a system and by an increase of situation awareness can improve safety, especially in TOR situations [93, 197, 203]. Next, we gained insights how TOR are designed in related work and that control transition are critical events while driving in a highly automated vehicle. An early preparation of a TOR holds the potential of informing drivers earlier of upcoming TOR.

Preparing a TOR visually is adequate when the urgency is still low and the information to convey is complex. Related work provided drivers with context information, which could help in TOR situations. An appropriate calibration of trust is important, since drivers should be able to assess correctly, if the automation can handle the situation or if an intervention in the driving situation is necessary. Trust can be influenced by several factors, for example psychological aspects, such as an increase of trust in a system that has anthropomorphic features [290].

Some applications, for example AR or VR in the vehicle could be enabled by $5G^{23}$. Predictions of the future traffic flow could predict critical events and therefore, the driver could be warned early of those.

Design Spaces are a powerful tool for designing novel applications. Exploration of existing designs is important to develop new design guidelines for the interior of fully autonomous vehicles. Not all use cases and their design dimensions are fully explored, which results in a lack of knowledge how to design for novel applications that take smart infrastructure information into account. Through lacking tools and a structured approach to address the challenge of designing for autonomous vehicle interiors that takes the environment into account, the design process is challenging. Not all dimensions and therefore limitations are known, which might result in loosing time through implementation errors that are caused by the lacking knowledge of the requirements for an application. Therefore, we further explore design dimensions and use cases for highly automated vehicles that are connected to smart infrastructure.

Augmented Reality extends the vision. Even though AR is still an emerging technology, it enables many application areas in the vehicle. These possibilities however currently are not explored to its fullest. Situation Awareness can be enhanced when choosing the right AR visualization. For complex and non-urgent information visualization is the most suitable presentation method [39]. With AR, new dimensions can be explored, especially when visualizations are not confined to windows, such as HUD or WSD. Current research focuses on visualizations in HUDs, WSDs or infotainment displays, yet HMDs offer the potential to visualize in new spaces, such as the interior of vehicles.

²³ https://www.zdnet.com/article/how-5g-will-affect-augmented-reality-and-virtual-reality/, last access May 2020

Related work explores a limited number of use cases. AR in vehicles offers the possibility of diverse and novel use cases. Therefore, we aim to explore and extend existing use cases of AR in vehicles.

One should aim for transparent AI algorithms that can provide explanations for their driving decision. Explaining a situation in autonomous vehicles consists of numerous components, ranging from just visualizing surrounding objects, giving context information and explaining the whole traffic situation. When explaining autonomous driving decisions, different aspects need to be considered, for example, how certain information is.

Following the definition of Doshi-Velez et al. [65], our research follows a *human-grounded metric* (real humans, simplified tasks), as explanations need to be adapted to human needs in autonomous driving. Therefore we follow their recommendations to evaluate our application in our context and primarily focus on the needs of the human and the context while using the system. Designing transparent AI algorithms is difficult and providing explanations remains challenging. While some users might have an understanding of what the algorithm takes as a feature to come up with a driving decision, others mental model might need more and different explanation. Explanations are perceived differently by users and most base their decisions on their own experience. Therefore, for users to trust and adopt an autonomous vehicles, their mental model needs to be taken into account, by providing adequate explanations. This will result in an improved driving performance and ultimately enhance safety.

Stakeholder and Use Case Evaluation of Smart Infrastructure

This section is partly based on the following publication:

• Gesa Wiegand (2019, March). Benefits and Challenges of Smart Highways for the User. In IUI Workshops.

Please refer to the beginning of this thesis for a detailed statement of collaboration.

Smart infrastructure can connect, inform, warn and help traffic participants on the road and while planning a route. Governments and industry undertake an effort to increasingly integrate smart infrastructure in the existing traffic and infrastructure landscape. The previous chapter presents research projects that integrate sensors and network communication into the infrastructure to create a connection between different traffic participants. The initiated projects aim to improve communication and information throughout the whole highway and across borders¹²³. The process of integrating sensors on the highway, building up smart infrastructure and connecting it with traffic participants, is complex. The advantages and the potential of setting up smart infrastructure for different stakeholders is still under research. For example, warnings about construction sites are communicated to drivers of connected vehicles through installed road side units (RSU) on highways crossing international borders⁴. Other use cases that can be realized by smart infrastructure are, for example, warnings about wrong way driving or co-operative merging assistance, in which involved vehicles and RSUs negotiate and cooperate with each other [143]. There are different research directions for smart infrastructure projects. Communication protocols, hardware setups, 5G network slicing, data fusion and prediction are important topics that are explored in different research projects and in the end aim for improving the journey of a driver connected to smart infrastructure. Concerning the users of the systems, the projects focus on vulnerable road users, vehicles and the general traffic management system.

Depending on the quality of the data collected by the smart infrastructure, the resulting use cases for smart infrastructure differ. For example, in case of data fusion results, for which the accuracy for position estimation is good enough on a centimeter scale, precise warnings in critical situations can be given. The prediction, which gives an estimate about the possibility of an event or the course of a path, can be used to inform the driver of a possible accident or an upcoming traffic jam. Depending on the precision of this forecast, take over requests or warnings can be designed. The cause of a take over request might be interesting for drivers of vehicles, but also for the maintainer of the road or in case of an accident, the police, ambulance or firefighters. For one occurring event, it can be therefore necessary to inform different stakeholders.

¹ https://c-its-korridor.de/, last access April 2020

² http://testfeld-a9.de/en/home-2/, last access April 2020

³ https://www.mobility.siemens.com/global/de/portfolio/strasse/connected-mobility-solutions. html, last access April 2020

⁴ https://c-its-korridor.com/?menuId=42&sp=en, last access May 2020

In the course of this chapter we examine the question who the stakeholders for smart infrastructure are and what the use cases for smart infrastructure are. In an informal case study [163], we collected initial responses of potential users about use cases. We follow the approach of an informal case study to collect insights in interviews and focus groups. Through ethnographic research, we gain insights in a field in which we do not have much expertise. The participants were selected through cooperation with the research project Providentia and Interview enquiries.

Since numerous traffic participants can be involved, the use cases are manifold and will be extended with the development of sensor, communication and interaction technology. For highway infrastructure, the main use cases and stakeholders are still under exploration. Smart infrastructure information can be included in future vehicles and the potential advantages for drivers need to be explored. Apart from drivers of vehicles, other traffic participants also benefit from connected infrastructure, such as operators of highways or emergency services. Emergency services such as firefighters need to plan their operation as efficient as possible to save time. With smart infrastructure information the path planning could be optimized and detailed information about the scene could be provided. Operators of the highway need to observe the highway and make changes to the route or velocity of the vehicles, therefore, they could benefit from smart infrastructure by adding smart infrastructure information to their system. Identifying use cases and user groups is beneficial to argue for setting up smart infrastructure which is still costly and faces many challenges. Thinking beyond highways, in cities, with multiple building regulations and a limited amount of space for smart infrastructure, the motivation for changing or extending the infrastructure with sensors needs to be sound. Identifying the right direction of research is important to increase safety, reduce emission and improve the overall traffic flow.

To explore the problem space of use cases for smart infrastructure and define the further course of research, we collected statements from different stakeholders of a smart infrastructure system. The focus in this chapter is on exploring the users of smart highway infrastructure, even though some use cases can be applied in urban environments. In this way, we identify benefits of smart highway infrastructure systems. In the following we refer to scenarios in the meaning of critical or alerting traffic situations. To understand the problem space of this thesis and identify stakeholders, benefits and use cases we conducted interviews, focus groups and studies with different possible stakeholders of smart infrastructure. From the focus group, described in the section Driver of Vehicles, we derived the other user groups, such as highway operators and emergency services. We focused on following three main groups:



Drivers of Vehicles⁵

In Section 3.1 we describe an interview and two focus groups that we conducted to identify use cases for drivers of vehicles that are connected to the infrastructure. In an interview with a leading expert for future vehicle design, we discuss the effects of smart infrastructure on the drivers. To identify the main benefits of smart infrastructure for drivers, we conducted a focus group (N = 9). Scenarios that are detected and transmitted by smart infrastructure are collected in another focus group (N = 4, m = 3, f = 1).

Emergency Services⁶

Firefighters are a potential user group of smart infrastructure data. Being the first responder in case of emergencies, it is utmost important to ensure a fast and smooth operation. In Section 3.2 we describe the results of an interview with a representative of this group in which we discuss the current main pain points during operation planning and execution. The results reveal that route optimization or exact location information can be beneficial for operations.



Highway Operator⁷

Operators maintain the highway and provide driving safety and efficiency. Accurate data from the highway and the resulting prediction of upcoming traffic events such as traffic jams would help operators to improve their traffic flow management. In Section 3.3 we describe the process of acquiring use cases of smart infrastructure for highway operators. In a focus group (N=8) consisting of operators of the highway (N=5) and smart infrastructure experts (N=3) potential benefits were collected.

3.1 Drivers of Connected Vehicles

Vehicles that are connected to smart infrastructure can provide users with numerous information and warnings. Depending on the level of automation, traffic environment or the overall purpose,

⁵ Icon made by monkik from www.flaticon.com

⁶ Icon made by Freepik from www.flaticon.com

⁷ Icon made by Freepik from www.flaticon.com

the notification can be adapted. In case of autonomous driving, the driver could be provided with rather leisure information. In case of manual driving, warnings would be more beneficial to the driver. Considering the traffic environment, when also considering traffic environments apart from highways, it depends if the traffic area is rather urban or if the vehicle is on the highway. Combined with the overall purpose, drivers could be then notified about an offer in a shop they are passing by or of an upcoming traffic jam on the highway. While drivers mainly refer to active drivers, in the case of autonomous vehicles, drivers are rather passengers of the vehicle. In the following we do not distinguish between vehicle occupants that are actively driving or being driven by autonomous vehicles. To collect possible use cases, focusing on a vehicle, connected to smart infrastructure on the highway, we conducted an interview and two focus groups.

3.1.1 Interview: OEM of Autonomous Vehicles

We conducted a semi-structured interview with an expert of an OEM in UI/UX Research. To include the look ahead of infrastructure in future vehicles, it was interesting for us to talk to a decision maker of the vehicle of tomorrow. By discussing the potential inclusion of infrastructure information in future vehicles a further research direction is identified of how drivers benefit from smart infrastructure in the vehicle. Therefore, in the interview, a focus was put on the cockpit information that one could enrich with infrastructure information: what is the effect of information about scenarios in a distance, which modalities could make sense, which technologies should be used for the output, how can uncertainties be communicated to the driver and what level of abstraction should be used when displaying the information. The interview was conducted in a briefing room at the expert's office. The interview started with the following question:

What is the consequence of the distance to critical scenarios, on smart infrastructure visualizations in the vehicle?

After posing this question and following up in a semi-structured way, the next question that considers different technologies and interaction options was asked:

Which other modalities and technologies would be useful to alert drivers of critical situations?

The categories that should be considered when including infrastructure information in the cockpit are distance, visualization, modality, technology, automation level and uncertainties.

Distance The interview resulted in insights about the impact of distance on potential warnings in the vehicle. In a larger distance (approximately more than 100 meters or if the scenario is out of the view of the driver) to a potential dangerous scenario, drivers could be observed. For example by identifying the mental load it can be assessed if they are capable of processing further information. Then, the information level should be adapted according to the mental load level, to not overload the driver with too much information. In case of the driver already being informed about the situation or being able to assess the situation correctly, it might irritate the driver to get even more information, an acute warning should be the only presented information. This should be visualized in the HUD, since other devices, such as the mobile phone, are distracting and thus it could influence the safety.

Visualization In situations, which can be handled by the automation itself without the need of a control instance, drivers should not become involved. For example, if the vehicle in front brakes, the autonomous vehicle should be able to keep a safe distance to the vehicle. In case of take over scenarios, the driver should not be shown even more information to not increase the workload in this situation. According to the interview, there is no need for the driver to understand the context of a situation, as it is also distracting and potentially dangerous. The visualization of the environment information needs to be abstracted as much as possible. For example warning symbols could be put in the driver's direction of sight. The visualization could be displayed in 2D or 3D when driving autonomously. An open question is, what detail level should be used for the visualization. Another challenge that needs to be solved, is if the visualization needs to comply with ethical guidelines.

Modality Concerning the modalities, drivers could be informed by different modalities, such as light, about critical situations. In very critical situations, it is imaginable to support the driver with brake assistance. While the already implemented steering wheel vibration is used to warn drivers about leaving the lane, it distracts the driver from the actual task, for example braking. Since every user has their own preferences, there should not be an exclusion of a modality. Every user should choose their modality according to their own preferences.

Technology The expert was convinced that HUDs should be used. If the mobile phone is used as a navigation system, it could also be used within the driver's sight. Information about more distant scenarios could be displayed in the instrument cluster or the central information display (CID). In the CID it is imaginable to display the situations according to their distance in a list. Another option would be to visualize it first as an information overview with the option to go more into detail for certain events.

Autonomous Driving The level of autonomy has an influence on the activities drivers can do and the mental load of the driver. This has an influence on the interface and information that can be provided to the driver.

Uncertainties When displaying information about a situation that is several kilometers away, uncertainties exist. On the one hand, the system could collect unreliable information, because the sensors are not good enough. On the other hand, the situation could resolve itself before the vehicle arrives at the scene. To accept the forecasting information of the system, the uncertainties need to be reliable and accurate. It is imaginable to visualize it, assuming that it is possible to communicate them accurately. Overall following conclusions could be gained from the interview:

Impact of Distance	The information transmission should depend on the workload of the driver in that situation
Visualization	Visualizations should be as simple as possible to not distract the driver
Modality	All modalities should be an option for the driver, when driving manu- ally it should not distract the driver from the actual driving task
Technology	The driver should be able to see everything in the HUD when driving manually
Automation Level	It is possible to visualize more complex scenarios when driving autonomously
Uncertainties	The communication of system uncertainties needs to be accurate and dependable, to give drivers a good approximation of the estimated travel time or urgency

3.1.2 Focus Group: Smart Infrastructure Experts

After getting a first input how the vehicle interface could change with integrating smart infrastructure information, we next identify concrete use cases for drivers. For that, we conducted a focus group (N = 9). The focus group consisted of sensor, infrastructure, and data fusion experts, mainly working in smart infrastructure or autonomous driving projects. In a first brainstorming session all participants were asked to identify use cases and user groups that potentially benefit from a smart highway. In a second session, the participants were divided into three groups. Each group was told to brainstorm use cases for different user groups. The first group thought about the driver and co-driver of a manually driven, highly automated or autonomous vehicle (SAE Level 2-5 [130]), the second of use cases for teleoperated drivers and the third about operators of the highway. The results for the operator of the highway are reported in Section 3.3. The collected results were clustered in the groups according to common themes and then presented by the group to all participants. After the presentation the results were discussed.

Driver of Manual, Semi-autonomous and Autonomous Vehicles The smart infrastructure information can be provided to different user groups of the system within a vehicle. Those are: Drivers of manual, semi-autonomous and autonomous vehicles (SAE Level 2-5 [130]). Manually driven vehicles nowadays have some sensors, but with a limited range. Therefore, those vehicles can detect objects up to 250 m away⁸. They could benefit from infrastructure information by brought-in devices. In case of drivers using their phone for navigation, infrastructure information can be provided to their phone. Additionally, warnings of dangerous traffic situations or route information can be transmitted via the phone, even in case of a vehicle without higher technology. In semi-autonomous vehicles, it is still necessary to take over the control from the vehicle. This take over situation can be supported through infrastructure information by giving drivers early on information about the upcoming traffic situation. Autonomous vehicles that drive without input from the driver, can include infrastructure information in their system and treat it as additional information. The smart highway is a redundant sensor system that can be used by the algorithms of autonomous vehicles to validate its own sensor information or extend the sensor information of the vehicle. Planning and reacting on upcoming traffic situations can then be improved. Drivers of autonomous vehicles benefit from smart infrastructure

⁸ https://www.bosch-mobility-solutions.com/en/products-and-services/passenger-cars-andlight-commercial-vehicles/driver-assistance-systems/left-turn-assist/long-range-radarsensor/, last access April 2020

rather indirectly as they do not distinguish between the sources of information that the vehicle needs to drive.

Teleoperator of Autonomous Vehicles Autonomous vehicles will likely face limits in operability in certain situations in which their sensors break down or there is not enough environment information to securely operate the vehicle. In those situations, a teleoperator, who is steering the vehicle remotely, might be able to steer the vehicle to its destination [265]. By providing teleoperators with information about the surroundings of the vehicle, they are able to steer the vehicle from the distance, for example by joystick [177]. In case the vehicle is connected to smart infrastructure the surroundings of the vehicle are also monitored by infrastructure sensors. Even if the sensors of the vehicle itself do not work, infrastructure sensors do not face the same limitations as the vehicle sensors and can serve as a redundant sensor set. Therefore, the lost information of the vehicle can be replaced. By connecting the teleoperator with the smart infrastructure, situations could be prevented in which vehicle's passengers are not able to continue their journey.

3.1.3 Focus Group: Traffic Scenarios

While we now have identified users within the vehicle that could benefit from smart infrastructure, we now identify use cases and traffic scenarios that could be detected by smart infrastructure. In the course of a focus group (N = 4, m = 3, f = 1), a scenario catalog was set up of traffic scenarios that drivers of a vehicles could be warned of or in which they could benefit from smart infrastructure information. This scenario catalog was initially set up in a brain storming session (N = 4, m = 3, f = 1) and extended over the course of the Providentia project. The following incomplete list contains some examples of these traffic scenarios that were determined as most important and easiest to realize and which reoccur in studies in the next chapters. The rest of the scenarios are described in Section 5.3, Figure 5.1.

• Standing Vehicle

If a vehicle that breaks down on the highway, other vehicles can be warned of that vehicle.

• Ghost driver

A driver that mistakenly enters the highway in the wrong way, can be identified by the smart highway and it can warn other vehicles on the highway of the ghost driver.

• Bad weather

During bad weather, the orientation on the highway might be difficult for the driver. In this case, the surrounding traffic and the distance between vehicles can be provided to the user. By combining sensor data from the vehicles on the road and sensor data from the smart infrastructure, the accuracy of detected objects could be increased. The data from, for example cameras that use deep learning-based object detection and radars that provide object detection can be fused in a data fusion unit. By combining the strengths of the sensors, for example cameras are good classification sensors because they can extract individual objects while radars provide distance measuring data to determine velocities [194], a system that covers all weather conditions is used that is more robust than the sensors of individual vehicles [120].

• Early Warning

In case of a smart highway with sensors that are distributed along several kilometers of the

highway the driver can be warned of possible accidents or dangerous situations along the route early on [299]. The infrastructure system has the advantage of having an overall overview of accurate information about the traffic on the highway. Today's advanced driver assistance systems (ADAS) detect objects and traffic situations in close proximity and warn the driver of braking situations in the surroundings of the own vehicle. Smart infrastructure though is able to detect braking scenarios that lead to brake cascades. A brake cascade originates from a fast braking vehicle and can lead to traffic jams and accidents of following vehicles. An early warning of fast braking vehicles can, therefore, result in smoother traffic flow and a faster and safer journey.

3.2 Emergency Services

Emergency services such as firefighter, police or ambulance are the first who have to be notified of emergencies and critical situations. They are connected to a control center to get information about their deployment and the location of their vehicle is tracked. In the control center all information is collected and processed further, which is a highly demanding process [309]. With cameras and detection algorithms, it is possible to accurately detect emergency vehicles in traffic and control the traffic to provide emergency vehicles with a more efficient path through traffic [246] Smart infrastructure can improve the efficiency of navigation by changing traffic lights or controlling the traffic [213]. It is imaginable to provide parties from the outside, such as the control center or a firefighter unit on the way to an operation, with a video stream from the critical event to improve planning and coordinate resources. To explore the problem space of emergency services we conducted an interview with a firefighter that coordinates deployments.

Interview: Firefighter To get information from a rescue person communicating with infrastructure operators, we carried out a semistructured interview with a Fire Chief Superintendent. The interview lasted about two hours. In the course of the interview we visited the firefighter station (see Figure 3.1) and observed the control center. Within the control center the responders answer calls, plan operations, distribute resources and supervise active vehicles. In the middle of the room is a large monitor with several interfaces that control and operation. The initial questions were:



Figure 3.1: Firefighter Station

What are the current pain points during a deployment?

How can smart infrastructure improve the deployment?

In the course of the interview with the firefighter, representing emer-

gency vehicle operators, it became clear that the information from the infrastructure needs to be integrated seamlessly to get quickly to the place of operation. In Germany, the fire fighters have an assistance period of ten minutes to get to the place of operation, from the beginning of the emergency call. Within one minute, it has to be decided, what emergency personnel and vehicles are needed for

¹⁰ https://www.muenchen.de/rathaus/Stadtverwaltung/Kreisverwaltungsreferat/Branddirektion-

Muenchen/Brandbekaempfung-und-Technische-Hilfe/Feuerwachen/Feuerwache-4.html, last access May 2020.

the deployment. This calls for accurate and swift information. To get to the place of operation the trip needs to be optimized to get there as quickly as possible. Emergency vehicles are allowed to use roads that other vehicles can not use (e.g. one-way streets or pedestrian roads). It could be also help-ful to notify other vehicles or road users of the upcoming emergency vehicles. Currently, due to data protection, the exact position information of the emergency is not transmitted during an emergency call from the phone. Additionally, temporal changes in the route are not updated dynamically in the navigation information. To conclude, following use cases are imaginable for smart infrastructure for emergency staff:

- Optimization of the route to the place of operation
- Exact traffic jam information
- Notifications to other vehicles
- Exact localization of the emergency
- Warning of temporal change of routes
- Connection of emergency vehicles

While the exploration of the pain points of firefighters revealed many points that should be addressed in future work, it does not lay in the scope of this thesis and could not be further investigated.

3.3 Highway Operator

In the previous Section 3.1.2 we collected use cases of smart infrastructure that could benefit operators. Those results already provided some insights in the advantages of smart infrastructure for highway operators. The operator of the highway needs to maintain the highway, know about damages and accidents to redirect the traffic flow. Currently, the traffic on the highway is directed manually by changing speed limitations or by indicating that the emergency lane can be used by vehicles. The additional information collected by smart infrastructure can be transmitted to the operator to improve traffic planning. Emergency vehicles can be supervised from far to find the best and fastest way to an accident. Traffic flow management could be enabled by smart infrastructure by predicting the vehicles



Figure 3.2: Highway Operator Room ©Siemens¹²

trajectories and behavior. By providing the operator with an accurate traffic density on the road, the average speed of the vehicles and predicted maneuvers part of the traffic flow control could be automated. The attention of the driver can be directed to critical situations on the highway and as a consequence reduce the reaction time. An automatic detection of hazardous situations would be desirable and realizable with smart infrastructure. It was mentioned that especially hazardous situations in tunnels through standing vehicles or persons should be detected fast so that the operator of the tunnel can quickly react on it.

¹² www.siemens.com/presse, last access May 2020.

Stakeholder and Use Case Evaluation of Smart Infrastructure

Focus Group: Highway Operator In a focus group with highway operators (N=8) the potentials and challenges of smart infrastructure information were highlighted. The highway operator monitors the highway, changes speed limits with speed indicators, controls a release of the hard shoulder when no vehicle is on the hard shoulder. In addition, tunnels are monitored when vehicles are stationary. The monitoring is carried out by cameras on masts and gantries.

Within the focus group two questions were answered:

What challenges do you currently face in changing and influencing traffic flow control?

What challenges do you face at road works?

Some ideas were already mentioned in the first focus group with smart infrastructure experts. Smart infrastructure can support highway operators with data to improve traffic flow management. The potential of an smart infrastructure is to provide traffic data, road works data and supervise tunnels. With accurate environmental data and the possible inclusion of environmental data from vehicles, potentially critical areas (e.g. slippery or uneven roads) can be identified and the operator can be notified. By giving highway drivers precise reasons for the cause of a speed limit, the acceptance of the speed limit could be increased and overall safety improved. Accurate sensor data can influence following traffic after road works or traffic jams. Forecasting the flow of traffic can improve traffic management. In addition, the control of road users as a collective can prevent traffic jams. At present, the legal framework hinders the personalized information prompts in the vehicle, e.g. through the radio, but it is conceivable to provide improved lane recommendations through displays. By determining times of low traffic and the current traffic flow, road works could be better planned. An automated detection of dangerous situations would be desirable. Especially in current dangerous situations, such as vehicle breakdowns or people in tunnels, the operator could react immediately to this situation with automatic detection.

Concluding, the following use cases of an smart infrastructure for the operator of the highway can be identified. The use cases belong to the categories *Informing in Critical Situations*, *Traffic Flow Management* and *Informing Drivers*:

Critical Situations

- Automatic detection of critical situation, e.g. in tunnels
- · Supervision of emergency vehicles to make navigation more efficient
- Identification of dangerous rough roads

Traffic Flow Management

- Accurate and smart traffic flow management through driving behavior prediction of traffic participants
- Identification of low traffic times to improve construction site planning
- Identification of the optimal driving lane
- Identification of traffic participants

Informing Drivers

• Redirect attention of drivers on critical situations through accurate and informative displays

3.4 Lessons Learned

We approached the exploration of use cases and user groups from diverse angles and with diverse methods. Many ideas and visions can be derived from the focus groups and interviews and a good overview of the topic can ge gained. Three main stakeholder for smart infrastructure were primarily mentioned by the participants of the focus group: Drivers, Highway Operators and Emergency Services. In an interface that provides the user with information about the infrastructure, additional information about traffic situations in a distance, can be communicated. Therefore, some reflection is needed on how to integrate intelligent infrastructure information into the interface of the stakeholders. In the following we present more insights on how this could be realized for each user group.

Driver of a Semi-autonomous and Autonomous Vehicle While design spaces for windshield applications and driver-based automotive user interfaces exist [108, 144], there is no design space for interfaces of autonomous vehicles in general. The cockpit of autonomous vehicles will gradually change with the automation level. As mentioned by the focus group of smart infrastructure experts, brought-in devices might get integrated in the vehicle interface. As mentioned by the autonomous vehicle expert, in manual vehicles, the focus is on keeping the distraction of the driver to a minimum. In autonomous vehicles, the driver can focus on other tasks than driving, and therefore the focus is more on leisure activities [223]. To put an emphasis on infrastructure interaction, the design dimensions need to be adjusted accordingly, including different indications of information from far or close to the vehicle or explaining the cause for a changed driving behavior. In theory, information from the whole length of a highway can be collected with smart infrastructure on the highway. Therefore, from the interview of the autonomous vehicle expert (see Section 3.1.1) it can be concluded that the information *what* has happened, *where* it has happened and *how reliable* that information is, are important aspects that can help drivers to form an informed decision for the route planning.

The information *what* has happened is crucial to understand the context of the information. Nevertheless, in a situation that requires a fast reaction of the driver, too much context information could be distracting and lead to dangerous situations. Given traffic situations that are further along the route from the own vehicle, and the driver does not need to react immediately on, more context information could be given. The same can be applied for the automation level of the vehicle. In case of higher automation, the driver does not need to pay attention to the road at all times, therefore other activities can be considered. Information about a scenario, so a traffic situation a driver needs to be informed about, also needs to be accompanied by distance information. As mentioned in the interview of the autonomous vehicle expert (see Section 3.1.1), the notification modality could differ, if the scenario is close by, compared to the modality of notifications of scenarios in a far distance. Otherwise, the drivers can not estimate if immediate action needs to be taken or if the scenario might change over time. The criteria, *how reliable* the information is, depends on the certainty of the information. Maurer et al. [194] discusses three types of uncertainty that can be accounted for: the *state uncertainty*, which describes the uncertainty of the physical measured variables such as speed or position. Then there is the *existence uncertainty*, if a detected object actually exists. And then there is the *class uncertainty*, which describes if the assigned class is correctly semantically assigned [194]. This information is then sent to the vehicle. Since the scenario could be in some distance, the scenario could already change, until the vehicle traveling on that highway arrives. The traffic jam could have been resolved, or the standing vehicle could already have continued with its journey. Therefore, the information how certain the system is of information should be communicated as well and need to be accurate and reliable (compare with Section 3.1.1).

Predicting whether a scenario will change until the vehicle arrives at the scene, could help the driver make an informed decision about changing the driving maneuver or route. In case of an accident, the algorithm could calculate that it might take some time to clear the highway. Therefore, the prediction could be very certain that the driver will lose time on the route. In case of fog in the morning, in far distance, the prediction could communicate to the driver that the fog will probably lift itself before the vehicle arrives at the scene.

Considering all this, the interaction possibilities and the displayed content will change with smart infrastructure information involved.

Teleoperator of Autonomous Vehicles Teleoperation, as mentioned by the focus group of smart infrastructure experts, is needed in several contexts, e.g. drone control, robot control in space operations or medicine. In automotive vehicles, the design of teleoperation interfaces is under research [29]. Teleoperated driving requires a network that has a high uplink data rate and a much lower downlink data rate according to Boban et al. [23]. Teleoperated driving is preferably latency free, therefore a minimal delay between the steering decision of the teleoperator and the movement of the vehicle is desirable [91]. To get an experience similar to that of a regular driver of a vehicle, several sensors (two or more cameras and other sensor information) need to transmit their information to the teleoperator interface. Berggren et al. test a teleoperated bus on a test bed [20]. To transmit the environment information and steering relevant information of the bus to the teleoperator, a driving simulator interface is used. Operators see a camera image of the cockpit of the bus on a screen in front of them. The input control can be manipulated by a steering wheel that is handled by the remote operator. The bus speeds up to 20 km/h. In another study by Georg et al. [91] teleoperated driving with head-mounted displays is compared with teleoperated driving with conventional computer screens. Even though they did not find significant differences between the two output modalities, the participants thought, the top down view of the vehicle surroundings was helpful. As mentioned by the participants of the focus group, with smart infrastructure and map information, this top-down view and all other visualization angles could be realized in teleoperated driving. Situation awareness would then increase if the whole environment would be visualized. Passengers of a vehicle that is steered by a teleoperator might not have transparency of the system that is in control of their vehicle. To regain control of the vehicle this information might be necessary though. Therefore, a clear and transparent communication of who is in control is necessary to gain trust of the user. Therefore, the accuracy of the sensors and the classified and identified vehicles on the road must be high enough to make accurate steering decisions. The participants of the focus group (see Section 3.1.2) do see the potential of smart infrastructure for autonomous vehicles. It is challenging to test and integrate teleoperated, connected autonomous vehicles in the infrastructure and for it to drive safely, technology needs to be enhanced.

Emergency Services From the interview with the firefighter, we identified six use cases for smart infrastructure information. Emergency services such as police or ambulances have many applications for connected traffic, such as surveillance, optimized traffic flow or identification of vehicles. Currently, the use cases are limited through the General Data Protection Regulation (GDPR) in Europe [235]. For example, number plates can not be tracked, which could be used by the police to

track speed offenders. From what we learned from the discussion of the potential of smart infrastructure for firefighter departments (see Section 3.2), there are some benefits. There is the restriction that all information needs to be efficiently integrated and easily available. The information available through sensors in smart infrastructure, can primarily be used, to make the navigation more efficient. This information is processed in the control center, where the information about available routes is available. Since the decision for a certain route needs to be made quickly, it is necessary to have an automated route optimization algorithm that takes all eventualities into account. In case of a daytime construction site or an accident, the information for a better route needs to be calculated instantly. Exact information is key for an efficient navigation. Therefore, the localization of other vehicle, traffic situations, temporal changes of the route and the exact localization of the emergency needs to be accurate and efficiently communicated to control centers and emergency personnel.

Highway Operators From our ethnographic research we learned that highway operators observe and control the highway. We identified eight needs of highway operators. First, we discuss the current workplace of highway operators before we elaborate the findings and the resulting conclusions of their needs. The velocity of the traffic can sometimes be regulated and the emergency lane can be blocked or authorized for traffic¹³. The design of control rooms consists of several working desks that have screens to show control applications or videos of the highway [131, 175]. One challenge of those large screen applications is the loss of orientation on the screens. Looking for the mouse cursor on large displays creates high physical demand. One possibility of improving control operators input techniques is eye tracking as suggested by Lischke et al. [175]. Some control rooms, such as nuclear power industries, have a strong focus on safety and performance of operators [180]. Therefore the control rooms need to be designed in a way that operators have no spatial constraint and have fast interaction possibilities. The importance of traffic flow management was stressed by the focus group participants (see Section 3.3). With smart infrastructure, operators of the highway could improve their planning, control and information management. Therefore they need an effective way to communicate and process the given information. In the automotive domain especially tunnels need to be observed. In case of an emergency, they need to be closed for incoming traffic right away. Therefore, the operator needs information of dangerous scenarios on the highway right away. Then, emergency vehicles can be informed and traffic can be controlled.

Challenges The benefits of the system promises unlimited autonomous driving, early warnings of take over requests, early traffic information or ecological driving. Nevertheless, there are some challenges that need to be addressed. Most presented ideas and concepts require the information to be fast and seamlessly integrated in existing technology. Drivers, as well as highway operators need to react to that information. In case of using the system as redundant sensor system for teleoperators or drivers, the system needs to be spread to the whole length of the considered highway. It is imaginable though, to focus the system on a limited stretch of roads with higher accident potential. To prevent others to harmfully interact with the system, safety is also an important topic that needs to be addressed. If wrongful information gets distributed, the algorithm of the vehicle might make a decision that can result in an accident or undesired behavior. The statements of the stakeholders led to numerous open questions. In the course of the thesis, the driver of a vehicle with smart infrastructure

¹³ http://www.stmb.bayern.de/vum/strasse/verkehrsmanagement/verkehrssteuerung/index.php, last accessed April 2020

connection, visualizations of scenarios outside the periphery of the driver and the interaction with a vehicle with unexpected driving behavior due to infrastructure information is researched.

Through the case study on stakeholder of smart infrastructure we gained in depth knowledge about pain points and possible use cases. Since the scope of this work is set on drivers, this work provides an overview of open research points for future work.

Design Space for In-Car 3D Augmented Reality Applications

This section is based on the following publication:

• Gesa Wiegand, Christian Mai, Kai Holländer, and Heinrich Hussmann. 2019. InCarAR: A Design Space Towards 3D Augmented Reality Applications in Vehicles. *In Proceedings of the 11th International Conference on Automotive User Interfaces and Interactive Vehicular Applications* (AutomotiveUI '19)

Please refer to the beginning of this thesis for a detailed statement of collaboration.

Innovative and emerging technologies such as connected vehicles, 5G [24] and smart infrastructure will change the way our traffic system works. In autonomous vehicles, these technologies could enhance, for example take over requests, by extending the vehicle's knowledge of the surrounding environment and infrastructure information [27, 28, 299]. The change within the transportation domain leads to visions of radically different vehicles in the future¹². Concepts of vehicles in which passengers are turned towards each other, or passengers watching movies on big screens while driving autonomously have already been envisioned by industry. The focus from the HCI perspective is on drivers that do not need to pay attention to the road in fully autonomous vehicles anymore.

With autonomous vehicles, a change in public mobility towards vehicle sharing is likely [176]. In this case, multiple people share the same vehicle, interface and sometimes the ride. A new driver might need to adapt to new vehicle functions and interfaces that were adapted by the previous vehicle sharer. It is therefore challenging to provide an intuitive user interface that quickly provides all necessary and personalized functions.

Augmented reality glasses or applications could offer a solution for fast changing and customizable interactions. AR glasses become more usable, smaller, have a bigger field-of-view and can be integrated in everyday activities. Head-mounted displays (HMDs) that display AR objects and support user input, are very likely to shrink in their size and might become as small as contact lenses [258]. There is no security risk of wearing HMDs inside an autonomous vehicle in which the driver does not need to pay attention to the road anymore, therefore, even today's technology can become relevant for a wide range of use cases. Due to the high transportability of compact HMDs a fixed physical interface might become obsolete. Vehicle manufacturers [56, 304] and researchers [37, 38] have already presented concepts to use AR for the design of novel in-car interfaces, e.g., augmented buttons floating in the air. To sum up, autonomous vehicles, infrastructure information and augmented reality interfaces offer the opportunity to create innovative automotive cockpit systems.

Nevertheless, this call for a change in the future vehicle's cockpit design is lacking a design space that clarifies the design criteria and constraints for new augmented reality applications. There is no

¹ https://www.volvocars.com/intl/cars/concepts/360c, last access May 2020

² https://www.youtube.com/watch?v=-8wWwYhBGhE&feature=youtu.be, last access May 2020

systematic approach to explore design alternatives for in-car 3D AR applications. Lacking tools and experience in designing new systems is time consuming, poses the risk of not using design dimensions to their full extend and thus missing out on important aspects. The problem remains how to integrate AR applications into highly autonomous and connected vehicles.

Design spaces for cockpit elements and windshields [108, 144] are tools to design and optimize applications on the windscreen or the cockpit of vehicles. Therefore, we propose a design space for 3D AR applications in highly automated vehicles that also considers use cases of vehicles that are connected with the environment. Those use cases go beyond safety and navigation to include entertainment and interaction functions [154, 223]. The strong advantage of design spaces is an overarching view of elements to consider, when designing new applications or demonstrations [142]. We build on existing automotive design spaces [108, 144] and develop them further, by investigating the use of AR in the highly automated vehicle's interior. In contrast to related work, presenting design spaces based on implemented examples, we present a design space to a recently emerging field of research. It is important to provide designers and researchers a design space in this early stage of development to give guidelines and support the research and development.

Our design space supports the concept development of in-car AR systems, serves as a communication tool, enables objective comparison of systems and can be used to formulate standardized requirements descriptions for building prototypes. We focus on SAE Level 4 [130] since the use of augmented reality could pose a distraction on a driver in control of the vehicle. Nevertheless, in control transition cases, it could also be helpful to use augmented reality to introduce a take over request.

To derive this design space we conducted a literature review of scientific publications, patents and commercial products. Two focus groups (N = 3; N = 5) with experts from the AR and automotive domain provided 84 additional novel use cases. Combined with the literature review, 98 examples for in-car AR are identified and clustered (Table 4.1, 4.2 and 4.3). By analyzing the design dimensions of the use cases and combining them with existing design spaces and taxonomies, we derive a design space for in-car AR applications. Hence, we contribute to the emerging field of in-car AR by transferring previous insights into a new context and drawing attention to possible differences. To motivate the feasibility we explore the design space by sorting in existing concepts and present a fictional usage scenario. By deriving a design space for in-car AR, designing applications in AR for in-car usage can follow a structured approach.

Within this chapter, we answer the questions what possible use cases and design dimensions for incar 3D AR applications in highly automated vehicles exist when 5G and smart infrastructure enable communication, entertainment or information about the environment. Furthermore, we answer how designers of new applications can be supported with the design space, by giving an example of a design process and by analyzing the designs of this thesis. In this way, we experience the benefits of using the design space to find new designs in a structured approach.

4.1 Methodology

To define the categories describing the design space for in-car AR, we followed four steps:

1 – Literature Review: *Related Work*

We reviewed related work on related design spaces and design categories for the development of user interfaces, namely public displays [206], head-up displays [108, 144] and windshield displays [108].

2 – Literature Review: *In-Car AR Use Cases*

To verify existing, remove unnecessary and add important categories found in Step 1, the literature review focused on existing in-car AR applications (Section *Literature Review*). Those existing applications were the first examples in the use case set.

3 - Focus Groups: In-Car AR Use Cases

We identified examples for in-car AR usage by conducting focus groups with experts from the field of AR and autonomous driving (Section *Focus Groups*). This way, we avoid missing application cases that are not yet published or developed, e.g., due to technical limitations. In the focus groups the novel use cases were sorted into the use case set of Step 2 while critically reflecting upon them.

4 – Building the Design Space

Based on the resulting use case set (Table 4.1, 4.2 and 4.3) we went step by step through the dimensions of related design spaces collected in Step 1. When going through these dimensions and discussing the applications, missing dimensions or unsuitable categories and characteristics became obvious. Some design categories were removed, enhanced, split or extended. To get a better impression of this process, we show in Figure 4.2 use cases, which are exemplarily fitted into the final design space (Figure 4.1).

4.2 Creation of the Design Space

This section describes the systematic procedure for developing the design space. We describe the previously introduced individual work steps such as literature review, focus groups and building of the design space of the methodology in detail.

4.2.1 Literature Review

We conducted a literature review using the following search terms: *hologram, holographic display,* 3D representation and S3D and combined them with the terms: *augmented reality, AR* and *auto-*

Design Space for In-Car 3D Augmented Reality Applications

	Entertainm	ent, Communication and Work						
Commercials	Commercials for products or ser-	Display products to support buying decision and to draw attention to offers						
	vices, e.g. promotions of stores	of products (P)						
Economy &	Economical driving and costs dis-	Show optimal trajectory in 3D world (D)						
Costs	play							
Work &	Information about general tasks,	Display route/map to visualize topography (P)						
Tasks	activities or (office) work tasks	Calendar with interactive elements (P)						
		Visualize weather like clouds/ current weather in vehicle (P)						
		Virtual keyboard (P)						
Driver Mood	Track the driver's status to pro-	Display basic requirements of single persons as a bubble over head (stres						
& Status	mote a mood or physical state	tatigue,) (P)						
Education	The driver can gain knowledge or	After Take over: Show Status of traffic situation/vehicle before take over						
	learn a specific behavior	(D,C)						
		Highlights within vehicle interaction elements that are needed (D, C, P)						
		360° view of the last 30 seconds to recapitulate past traffic situation (D,						
		C, P)						
		Sightseeing objects and information about them are displayed that one is						
		driving past (D, C, P)						
		Certain body movements are augmented to train the body (C, P)						
Gaming	Game play alone or together with	Board games in AR (P)						
	others	Cooperative AR games with other passengers (P)						
		Extended world, the horizon of the visualization extends over the bound-						
		aries of the vehicle (P)						
	~	Virtual play partner for kids (P)						
Multimedia	General information; music or	Entertainment interface (D, C, P)						
& Web	video player; access to the Inter-	Locked interface wherever the passenger is turned to, to manipulate enter-						
	net or news; specific for passenger	tainment depending on the view direction (P)						
	entertainment and waiting times	3D world to look at (P)						
		Extended interface cooperating with laptop, so that contents can be						
		dragged and dropped. (P)						
		3D movie (P)						
		Ploating text (P)						
Anto & Dho	Distura or arts presentation appli	JD AK VIdeo Calling [155]						
tography	actions that anable drawing and	In Air Drawing (P)						
tography	taking photos	raking pictures of situations outside and sharing them with outers in the						
	taking photos	Taking picture of environment as 3D picture (P)						
Atmosphere	Creation of a different atmosphere	Blend out dangers outside (P)						
runosphere	in the vehicle by displaying differ-	Relaxation/reduction of information like counting sheen (P)						
	ent surroundings or ambient lights	Personal spaces during ride sharing (P)						
		3D curtain to block out distraction from other passengers (D, C)						
		Visual Overlay of other vehicle passengers (P)						
		Relaxing abstract forms that adapt to the vehicle movement (P)						
		AR sound curtain [55]						
Observation	Observation of a person, normally	virtual huts/routes in the mountain (P); other chosen vehicles relative to						
	relatives, or an object and its state	destination (D, C); Prediction and visualization of vehicle trajectory (D,C)						
Social Inter-	Social interaction with other par-	Teleport friends in vehicle (P)						
action	ties than drivers	Easier conversation with back seat, like showing kids virtually where						
		snacks are (P)						
		Virtual personal assistant (P)						
		Telephone partner in AR (P)						
Driver2Driver	Communication with other drivers	3D avatars (vehicle, person) to communicate with other vehicles/people						
Communica-		on the road (P)						
tion		Shared interface during convoy driving to plan trip/ choose music (P)						
Internet of	Access to or control of things	Smart Home: model of house and easy interaction, see status of connected						
Things		devices/house (P)						
Health	Encourage and support healthy	AR trainer shows movements to follow (P)						
	behavior in the vehicle	Virtual Sport: Sailing or Golf that adapt the movement of the vehicle (P)						

Table 4.1: The use case set for entertainment applications. Each category features entries accompanied by a short title, description and examples from either related work or focus groups. The entries from related work are cited and the results from the focus group are labeled with the primary user group (D = Driver, P = Passenger, C = Co-Driver).

motive. We specifically investigated literature regarding in-car usage, hence, applications within the vehicle. To that end, we searched on the following platforms: Google Scholar ³, Springer Link ⁴, ACM DL ⁵ and IEEE Xplore DL ⁶. We excluded all papers that use windows and windscreens to place applications.

4.2.2 Focus Groups

We organized a first (N = 3) and second (N = 5) focus group to find possible applications for the use case set that resulted from the literature review. In our focus groups, we had one female and seven male participants between the ages 25-35. Four were AR experts, two autonomous driving experts and two professionals working in both domains. In the first focus group, use cases were generated by a Brainwriting session [276]. To prepare for the writing, participants were introduced to the topic and several examples from related work were presented and discussed. They were told to focus on highly automated vehicles from Level 3 to Level 5. Additionally, we specifically told them to just think of use cases that do not need windows or dashboards. After finishing the writing, resulting ideas were sorted by the participants into the existing subcategories – e.g., Subcategory View Point in category Safety (see Table 4.2) – of the use case set of Haeuslschmid et al. [108]. The first focus group came up mainly with ideas in the category *Entertainment and Communication*. To create a richer set of possible use cases, the initial question in the second focus group was: "What is important for you during a journey in the vehicle during the start (navigation), the driving on the highway/city (safety and entertainment) until the end (looking for a parking lot)?". Additionally the following scenario was outlined: "Imagine you live in the year 2040. You are a designer for vehicle interiors. Your new task is to design the next vehicle's interior with focus on floating 3D AR interfaces. You have the following information: vehicle status, vehicle2vehicle information, information of the surroundings, infrastructure information and passenger information." We then asked the participants brainwriting ideas for AR systems that address a particular user group in the vehicle. They should imagine a "driver" that is conducting a take over request. Then to think of the "attentive co-pilot" that also pays attention to the road and helps the driver. Finally they wrote concepts for a "passive passenger" in the vehicle. The other categories were not explained to not bias the participants towards the existing use case set. The examples from the brainwriting were discussed and similar ideas were put in relation to each other by the group. After the focus group two study supervisor compared the created clusters with the existing use case set. Further we sorted the resulting ideas in the subcategories of the use case set independently, again to find possible gaps. The focus groups lasted about 120 minutes each.

4.2.3 Building the Design Space

We identified 14 use cases for our problem space through the literature review. The focus groups resulted in 84 novel use cases for floating 3D in-car AR applications (see Tables 4.1, 4.2 and 4.3). AR applications are dominantly imaginable in the category *Entertainment and Communication* with

³ https://scholar.google.com, last access: May 2020

⁴ https://link.springer.com, last access: May 2020

⁵ https://dl.acm.org, last access: May 2020

⁶ https://ieeexplore.ieee.org, last access: May 2020

Design Space for In-Car 3D Augmented Reality Applications

Safety							
Vision Exten-	Extension of the driver's view by	Using 3D AR to visualize occluded objects on the road (D,C)					
sion	displaying occluded objects	Visualization of approximate traffic situation (D,C)					
		Visualization of Countdown until dangerous situation (D)					
Vision En-	Enhancement of the driver's vi-	Visualization of surrounding to aid with decision making in uncertain sit-					
hancement	sion in bad viewing or lighting	uations (D,C)					
	conditions						
View Point	Display of the views of virtual or	Camera image of accompanying drone in vehicle (D, C)					
	real cameras; often replacing mir-						
Count of a							
Spatial	derstanding of the appage around	Enlarge vehicle interior with virtual objects to see dangers further on (D)					
Awareness	the vahiale	comparison between venicles 5D sensor model and reality by overlaying					
	the vehicle	for action (D)					
		3D AR avatar that reacts on the environment [107]					
		3D representation of the highway to prepare take over request [299]					
Monitoring	Safety-relevant information about	Birds-eye view of possible dangerous situation to inform about dangers					
Surround-	the environment	and limitations and give information about route and alternatives (D)					
ings		Adaptive virtual map with dynamic objects of surroundings and informa-					
_		tion about other drivers on the road (risk and aggressiveness) (D)					
Driver Moni-	Driver performance and physical	Inform passengers about status (attentiveness, tiredness, aggressiveness)					
toring	state observation	of driver with a 3D overlay over the driver (C)					
Breakdown	They help the driver in breakdown	Help of virtual assistant of breakdown service (D,C,P)					
	situations	Block doors with virtual signs to prevent passengers from leaving through					
		wrong door (D,C,P)					
a 10 a	~	3D tutorial what to do (D,C,P)					
Specific Sup-	Systems directed at a specific	3D Highlight to focus gaze of driver on situation that requires his/her at-					
port Systems	party, support several tasks or pro-	tention (D) Civing action accommon dations (D)					
	vide various information	Show where hands and feet should be to steer vehicle (D)					
		Visualization of trajectory planning and showing alternative routes in pos-					
		sible situation (D)					
		Guiding of drivers attention on highly prioritized external factors (D)					
		Co-driver guides attention of the driver on specific situation by manipu-					
		lating virtual arrow (C)					
		Visualization of intention of other drivers when they are crossing my way					
		(D)					
		Using an avatar of other drivers to communicate with them (D, C)					
		AR Warning [174]					

Table 4.2: The use case set for safety applications. Each category features entries accompanied by a short title, description and examples from either related work or focus groups. The entries from related work are cited and the results from the focus group are labeled with the primary user group (D = Driver, P = Passenger, C = Co-Driver).

43 examples. The use cases were clustered into five categories: Safety (see Table 4.2), Entertainment, Communication and Work (see Table 4.1), Navigation, Convenience and Vehicle Monitoring (see Table 4.3). The use cases identified by the literature review were in the category *Safety* (e.g. an AR warning [174], a 3D AR avatar [107] or a 3D representation of the highway to prepare a take over request [299]), *Vehicle Monitoring* [174, 227], *Entertainment, Communication and Work* such as a 3D AR video calling interface [155] or a sound curtain [56] and in convenience, as alternative interfaces to the dashboard [35, 36, 37, 56, 247, 304]. The category *Navigation*, includes path finding, planning and public transport. We identified the need for a category regarding *Convenience*. A similar category is also described in the work of Brandt [31] in which it is linked to the category Entertainment and Communication. The focus of the category "Convenience" in Brandt's paper is on telecommunication and radio. For a clearer separation of the use cases, we decided to separate the categories into *Convenience* and *Entertainment and Communication*. The new subcategories in the category *Convenience* are: Well-being Aid, Load and Interface Enhancement. It also describes alternative interfaces to the dashboard [35, 36, 37, 56, 247, 304]. Exemplary use cases are related to level out bodily effects of the ride, storage of objects in the vehicle or providing alternative interfaces on surfaces. *Entertainment and Communication* relates to leisure activities in the vehicle. *Vehicle Monitoring* combines the status of the vehicle, its supervision, maintenance and fuel or battery status [174, 227]. The methodical approach of sorting exemplary use cases in the design space is clarified in Figure 4.2 and a design process is described in Section 4.4. The colored lines indicate the design categories that are used or should be considered in designing the example application. The visualization can be used to not forget about certain design alternatives during the creative process.

Defining the Targeted User Groups

The user group benefiting most of this design space are drivers of SAE Level 4 [130] vehicles or higher, attentive co-drivers and passive passengers. Drivers of SAE Level 0-3 vehicles need to observe the road and should not be distracted by floating 3D in-car AR applications [111, 126]. *Drivers of SAE Level 4* vehicles may benefit from in-car AR before the situation of a take over request occurs and while driving manually. *Attentive Co-drivers* support the driver in navigational, safety and dashboard control tasks within their range of control. *Passive Passengers* is the umbrella term for the driver who is driving autonomously, the co-driver who does not pay attention to the road and other passengers.

4.3 Design Space

This section presents the outcome of our methodological approach described in the previous section which resulted in a design space (see Figure 4.1). In the following, we describe the dimensions (e.g., *User*), categories (e.g., *User Mode*) and their characteristics (e.g., *Single User*).

4.3.1 User

We identified three user groups which are defined in the section above (*Defining The Targeted User Groups*): Drivers of SAE Level 4, Attentive Co-drivers and Passive Passengers. Resulting from those target user groups, there are three categories: User Mode, Observer and Actor.

User Mode The user mode contains *Single User* and *Multi User* as the systems can either be used alone or in cooperation with others.

Observer Users observing the system are part of our identified user groups: the *Driver*, *Attentive Co-driver* and *Passive Passenger* of a vehicle. Additionally, observers are only in this role if they are focused (either with their gaze or mind) on the application.

Actor An actor is defined as a user manipulating the interface [108]. All observers might be actors. Additionally, a remote actor can interact with the interface. For example, a service transmitting information to maintain the vehicle or a non-present colleague cooperating via a shared interface with actors inside the vehicle.

Design Space for In-Car 3D Augmented Reality Applications

		Navigation						
Path Finding	Support in finding the way to the	pport in finding the way to the Digital representation ("digital twin") of highway to visualize real-time						
	target	route information (D,C)						
Vehicle-	Support when following a vehicle	Compass that points towards vehicle to follow (D,P)						
Following		Visualization of distance and status of other vehicle (D)						
Traffic &	Display of traffic signs currently	Representation of current speed limit sign (D)						
Street Signs	applying; street signs and names							
Points of In-	Additional on-route information	Visualizing map with parking spaces (D,C)						
terest	to find people, shops or services	Visualization of POI along the road in map (D,C,P)						
		Preview of route highlights in AR (D,C,P)						
Public Trans-	Support for commuters who	3D model of vehicle park/train station with visualization of walking routes						
port	switch to public transportation	(D,C,P)						
		Visualization of time of arrival of train by displaying train model in dif-						
		ferent locations depending on time of arrival (D, C,P)						
Path Plan-	Support in planning the route and	Activity planning on interactive map that notify all passengers of their						
ning	adjusting the route while driving	tasks (C, P)						
	when external influences force a	Support to plan route when several passengers need to pick up by using						
	route change	avatars in interactive map that can be dragged & dropped (C, P)						
		Convenience						
Wellbeing	Aids to prevent motion sickness	Abstract 3D forms and pictures to visualize vehicle kinematics (P)						
Aid		Inform passenger visually about routes ahead to prepare him/her for the						
		next section of the ride (P)						
Load	Everything that is related to load-	Luggage Tetris to visualize best packing strategy (P)						
	ing the vehicle.							
Interface En-	Alternative interfaces	Interface elements in 3D (D,C,P)						
hancement 3D highlight of individual interface elements and own								
	(D,C,P)							
		3D AR Dashboard [35, 36, 37, 56, 247, 304]						
		Multimodal Feedback with AR [247]						
		Vehicle Monitoring						
Vehicle Sta-	Information about vehicle parts	3D model of vehicle with status information and eventually statistics of						
tus	and the momentary status of the	vehicle functions (D,C, P)						
	engine	Miniature 3D model of vehicle with luminated defective parts (D,C,P)						
		AR owner manual [227]						
		AR vehicle status [174]						
Supervision	Support in supervising the vision	Visualize breakdown of sensors in virtual digital twin of vehicle sensors						
	of the vehicle's sensors; most of	(D,C,P)						
	them aim to increase trust							
Fuel & Bat-	Information about the current fuel	Status information of fuel/battery in sight (D,C,P)						
tery	or battery status							
Maintenance	Information of the overall status of	3D avatar of vehicle with a character that indicates that everything is ok						
	the vehicle	(D,C,P)						

Table 4.3: The use case set for navigation, convenience and vehicle monitoring applications. Each category features entries accompanied by a short title, description and examples from either related work or focus groups. The entries from related work are cited and the results from the focus group are labeled with the primary user group (D = Driver, P = Passenger, C = Co-Driver).

4.3.2 Context

The *context* of use is defined as the nature of the users, tasks and equipment. Furthermore, it includes the physical, social and cultural environment in which a product is used [98, 259]. Within this dimension, there are the categories Application Purpose, Information Context, Driving Mode, Level of Automation, Privacy and Travel Time.

Application Purpose The *Application Purpose* contains the characteristics *Safety*, *Navigation*, *Vehicle Monitoring*, *Entertainment*, *Communication and Work* and *Convenience*. These characteristics are defined by Brandt et al. [31]. *Work* was added to the category *Entertainment and Communication* because of the possible importance of this category in autonomous driving. We added one category since we believe productive (office) work will become a relevant aspect of highly automated driving [49, 162, 229, 249]. *Navigation* includes information about the trip or path and in contrast to previous definitions, we include route planning as well.

Information Context This dimension describes factors influencing the situation in the vehicle. *Environment* includes e.g., the size of the available space inside a vehicle, the amount of seats and their orientation or people who are present. *Vehicle* considers the vehicle and its purpose, e.g., information about the engine or the vehicle type. *Personal* includes all information about a person, such as position, age or physiological data. *Time* describes the time of the day, date and season.

Driving Mode The driving modes are based on Häuslschmid et al. [108] and include *Driving*, *Waiting* and *Parking*.

Level of Automation We focus on SAE Level 4 [130] of automation. Hence the level of automation can be *Manual*, *Semi-automated* and *Autonomous*.

Privacy The privacy settings of an application can be *Public* (e.g., sports results), *Personal* (available for family or business insiders) and *Private* (only accessible for a single person) as proposed by Häuslschmid et al. [108].

Travel Time We added this category to distinguish between *short*, *medium* or *long* duration of a drive. Depending on the travel time different side effects might occur. Especially long rides require planning breaks or considering energy loading stations.

4.3.3 Visualization

In this dimension we describe relevant categories (Level of Augmentation, Registration, Placement Strategy, Field of View Position, Presentation and Graphic Design Factors) related to the visual presentation and perception of in-car AR.

Design Space for In-Car 3D Augmented Reality Applications

	User Mede	Single User					Multi User					
User	Observer	Driver (SA	-	Attentive Ca Driver			Multi User					
	Observer	Driver (SAE Level 4)				Attenuive Co-Driver			Passive Passenger			
	Actor	Driver (SAE Level 4)		At	Attentive Co-Driver		Passenger			Remote Actor		
Context	Application Purpose	Safety	Safety Navigation		Vehicle Monitoring		Entertainment, Communic and Work		mmunicati ork	tion Convenience		
	Information Context	Environment			Vehicle			Personal			Time	
	Driving Mode	Driving				Waiting			Parking			
	Level of Automation	Manual				Semi-automated			Autonomous			
	Privacy	Public				Personal			Private			
	Travel Time	Short duration				Medium	Medium duration			Long duration		
Visualiza-	Level of Augmentation	Reality		 Augmented Reality 				Virtual Reality				
	Registration	Unregistered				2D registered			3D registered			
	Placement Strategy	None			Binary			Linear Exp			onential/Higher Order	
	Field of View Position	Foveal				Central			Peripheral/Ambient			
uon	Presentation	Symbolic					→		Naturalistic			
	Graphic Design Factors	Color		Transpar	rency	Si	ize		Motion		Depth	
Interaction	Input Modality	Touch & Control	Ge	stures		Gaze Speed		h	Behavior		Physiological	
Interaction	Multimodal Feedback	Haptic/Tactile		Auditor		Olfactory		Sens	ense of Balance		Temperature	
Technolo- gy	Image Generation	Glasses-Bas Displays	ed	Head		Worn Displays		Autostereoscopic 3D Displays		Н	Hand-Held Displays	
	Size	Full Space]	Full Dimen		Static Area			Situated		Dynamic	
	Depth	2D				Pseudo-3D			3D			
	Display Factors	Color Depth Resolution			tion	Contrast I		В	rightness Transparency		Transparency	

Figure 4.1: Design Space for In-Car AR.

Level of Augmentation In contrast to previous design spaces we do not include discrete categories such as AR and VR. We base the definition of the *Level of Augmentation* on Milgram's Mixed-Reality Continuum [198]. The Mixed-Reality Continuum is an axis with real world content on the one end and exclusively virtual content on the other end. The idea of the continuum is that an AR system can be anywhere along the axis. A designer can use this continuum to decide for the degree of virtual content presented to the user. Furthermore, Mann's Mediated Reality continuum [189] adds the idea of filtering to Milgrams concept [198].

Registration Unregistered visuals are placed unrelated to an physical object within the vehicle. 2D registration is defined as being registered to an object, but not meeting its depth. 3D registration means a positioning of elements with regards to a physical objects position and its depth. Different to previous definition [108], we do not include gaze-dependency in this category. We argue that the definition of AR registration describes a positioning of virtual objects relative to the physical world. In case of a visualization that follows the gaze, the 2D or 3D registration changes based on the focus of the gaze. Therefore the gaze-dependency is a result of a change in the 3D registration based on the users *input* via *gaze*.

Placement Strategy Previous work lacks in a description of the change between two points of registration, therefore we add the new category *Placement Strategy*. The placement strategy describes animations, needed to change positions of an AR object or the registration state, e.g. from 2D to 3D registration. We came up with categories based on work by Lauber et al. [161]: *None* (the former visualization stays untouched, a new one appears), *Binary* (e.g. disappearing on one spot and emerging at a new one), *Linear* (e.g. following the head movement) and *Exponential/ Higher Order* (e.g., inertia on an element when turning the head, it follows slowly, then catches up).

Field of View Position Häuslschmid et al. [108] propose the characteristics *Foveal*, *Central* and *Peripheral/Ambient* which we are using.
Presentation The Presentation of augmented reality objects can be *symbolic* (abstract) or have a *naturalistic* (real-world) appearance. There are no discrete distinctions between the characteristics, but a smooth transition with infinite states in between.

Graphic Design Factors In addition to the definition of Häuslschmid et al. [108], we add *depth* as characteristic to the category *Graphic Design Factors*. Many of the examples found in the focus groups rely on having objects in different depths available. Zooming, panning, moving and interacting with objects benefit from providing a depth dimension.

4.3.4 Interaction

Interaction with 3D AR applications can be designed by altering the following categories.

Input Modality Inputs can be triggered in an implicit or explicit manner. Implicit actions describe a reaction of the system without a conscious decision of the user, whereas explicit inputs include only informed actions of the user. *Touch & Control, Gestures* and *Speech* are normally used for explicit input. *Gaze* can be used for explicit input, but might also be executed without any intentions. *Behavior* and *Physiological* are mostly used to define implicit input. *Behavior* describes facial expressions, eye movements, body position and posture of the driver [206]. To avoid implicit *Gestures*, they should differ from natural movements and thus be learned. However, gestures can reflect movements used in natural interactions, such as unscrewing a bottle by turning the hand [224].

Multimodal Feedback In this category we include *Hapticl Tactile*, *Auditory*, *Olfactory*, *Sense of Balance* and *Temperature*. The use of *tactile* feedback is well established in the domain of smartphones and could be transferred to in-car AR objects. The *sense of balance* can be activated via movements of the vehicle chassis. For example, to support virtual sports experiences such as sailing (use-case "sailing" in Table 4.1). *Temperature* has been used as a feedback modality [301]. It could indicate points of interests on a map (e.g., feeling warm spots indicating restaurants on a map).

4.3.5 Technology

The technology used to create 3D images in a vehicle is based on an effect achieved by showing two slightly different images for each eye [184]. We refer to Billinghurst et al. [21] for more detailed information of AR technology.

Image Generation Based on previous work [35, 274, 275], we identified four suitable classification categories for in-car AR display technologies: *Glasses-Based*, *Head-Worn*, *Autostereoscopic 3D* and *Hand-Held Displays*.

Current *glasses-based* commercial products e.g., the "3D Vision 2 Wireless Glasses" by NVIDIA ⁷, combine a display with glasses worn by the user. The glasses separate the two stereoscopic pictures for the left and right eye respectively and often implement active shutter systems [30, 184]. Other

⁷ https://www.nvidia.com/object/product-geforce-3d-vision2-wireless-glasses-kit-us.html, last access: May 2020

Design Space for In-Car 3D Augmented Reality Applications



Figure 4.2: The lines visualize the possibility to sort in and compare exemplary use cases from the use case set (Table 4.1, 4.2, 4.3). Lines crossing a field vertically indicate that it belongs to the use case. In yellow, underlain with a shadow, the line for the implication example (Section 'Example Scenario') is shown.

glasses use interference filters [140, 184] or polarization [184], to produce the 3D effect. Especially for *glasses-based* AR applications, the field of view becomes an additional factor for consideration. Current technology, such as the HoloLens or Vuzix Blade ⁸ enable a field of view of about 30°x17.5°(HoloLens). Designers need to understand if their chosen technology is able to comply with their field of view requirements. *Head-worn Displays* include two optical micro displays to produce a 3D experience [125, 221, 233]. Benefits are independency of the user's position and headtracking possibilities. They can support the full spectrum of the *level of augmentation* [135, 262, 263]. *Autostereoscopic 3D Displays* show 3D images without additional devices [64, 125, 184, 274]. Well-perceivable 3D-visual effects created by such systems are restricted to a specific point of view. Additionally, they can suffer from distortion, lead to fatigue or display flipping images [9, 139, 158, 196]. Broy et al. [37] implemented a stereoscopic 3D interface in the dashboard of a vehicle and showed that it enhances the presentation of spatial information (e.g., navigational cues) compared to 2D presentations. *Hand-Held Displays* are a useful tool for implementing optical see-through applications to augment reality [202, 281].

Size The AR content can make use of the *Full Space* of the vehicle interior. Alternatively, it could cover a *Full Dimension* in one of the three spatial axes limited by the vehicles interior dimensions. AR objects could also fill any geometric *Static Area*, e.g., floating circularly inside the interior. Additional AR features could be placed contextually *Situated*. Hence, information regarding the status of the vehicle could overlay parts of a physical instrument cluster. Furthermore, the image size could actively adapt to the displayed content by using a *Dynamic* range.

⁸ https://www.vuzix.com/products/blade-smart-glasses, last access: May 2020

Depth Images for in-car AR features could render 2D views e.g., for a digital speedometer, a web browser floating inside the vehicle or the top-view of a map. *Pseudo-3D* imitates depth with perspective 2D images in a way that objects do not appear flat, but rather cubical (e.g.,[146]). Real 3D images show a spatial illustration which has visual information in three dimensions analog to real world objects [15, 51, 128, 208]. Thus, these representations contain an explicit value for depth perception. Kulshreshth et al. present active 3D stereoscopic rendering techniques adapting to the displayed content [153].

Display Factors The attributes *Color depth*, *Resolution*, *Contrast*, *Brightness* and *Transparency* are crucial attributes for a well perceivable illustration. The source of the image has to meet specific requirements to present a satisfying in-car user experience. Changing light conditions (day / night) and rapidly passing-by shadows are potential aspects of everyday drives and need to be considered in the design process.

4.3.6 Differentiation From Other Design Spaces

We refer to our suggested classification system as design space, although we introduce more than two dimensions and no x- or y-axis to locate content. We consider our work as a tool to enhance future developments of in-car AR, therefore it is more than a taxonomy. Also, it is not structured hierarchically. Hence, our work extends the usual appearance of a design space and could be perceived as a meta-design space with influences of common taxonomies. Our proposed design space builds upon previous work from the automotive domain and therefore has some overlapping factors [108, 144, 206]. However, we did not find similar work focusing on in-car AR. In contrast to the previous work there are manifold additional opportunities for positioning, interaction, use cases and users. With our design space we add the novel categories *travel time* and *placement strategy*. We have also significantly redefined the categories *level of augmentation*, *registration*, *graphic design factors*, *multimodal feedback*, *observer*, *actor* and *image generation* to meet the needs of in-car AR. To distinguish our design space further from others, we discuss the particular differences below:

Positioning The position of the interaction element plays a significant role for its use case. Augmented objects on the windshield force the user to actively turn towards a window. 3D AR objects can be boundlessly adapted to the user's field of view and orientation in three spatial dimensions. Therefore, our suggested design space expands beyond multiple positioning options and includes placement strategies.

Interaction Floating 3D objects are manipulated with different gestures than 2D interfaces [224]. Direct touch interaction is not recommendable for windshields applications as it would obstruct with the vision [108]. Public displays need to count on attracting the passers-by attention and need a different interaction design [206]. In-car AR demands a consideration of interaction strategies, which is reflected in this design space.

Use Cases and Users The use case set of 3D in-car AR and windshield displays both include a majority of entertainment and communication applications. However, use cases for 3D in-car AR applications are not limited to a position (e.g., windshield) and can address all passengers of a vehicle without occlusion. Especially, if the vehicle design changes in the future, the windshield might not

be a point of interest anymore. For example, all seats could face each other. Use cases for in-car AR include every possible user inside of a vehicle and furthermore, could be used to digitally augment passengers or even replace empty space with people who are not present.

4.4 Usage of the Design Space

In the following section we suggest a way how to use the Design Space and motivate its validity. We intend the design space to be a tool during the creative design process that (1) helps to keep all dimensions in mind, (2) provokes an objective and thorough discussion of each design category, (3) supports identification of critical design categories. In the following example we will refer to these numbers, wherever the story addresses one of the prerequisites. We present a fictional scenario that showcases the ability of the design space to achieve these goals. As input we introduce personas and a user story. The aim of the fictional example is to develop a new concept for a system that enables joint planning of the drop-off point at the destination.

4.4.1 Personas and Use Case

We introduce two personas that are different in their background and abilities, still they are very likely to work together in a UX design process.

Laura is a 29-year-old UX designer that just graduated from university. She has some background in the automotive domain from lectures at university and is a specialist in the application of UX methods and user-centered design. However, she does not yet have a proper overview on technological trends and user needs when designing in-vehicle applications.

Peter is a 45-year-old computer scientist with 15 years of experience in developing for the automotive domain. He is responsible for the implementation of technology for the human machine interface in the vehicle. Throughout his working experience he gained some knowledge about UX design, however his main interest is in making things work.

Use Case Due to the current success of AR technology and the ongoing progress in the development of automated vehicles Laura and Peter are asked to think about how these two domains can complement each other. On a higher management level it was decided to work on concepts that explore collaborative route planning. When several persons drive together in a vehicle, the route is defined early on. If spontaneous alterations in the route occur due to the needs of the passengers, for example due to a sudden changed meeting point, the drop off location changes as well.

4.4.2 Example Scenario

Peter feels overwhelmed by the problem. Although, he knows a lot about the traditional driver centered interface design, he does not know yet how to approach in-car AR solutions. Laura got to know the design space for in-car AR (Figure 4.1) in her studies and introduces it to Peter. Peter has a look at it and finds a number of familiar words and categories drawing his interest. They agree to focus on the *multi user* aspect of the problem due to the prerequisites of their management. The request is to built something for a *passive passenger* as an *observer* and *actor*. To stay aware of the characteristics, they

mark it in the printed out design space, which makes it visible to everybody (final result: Figure 4.2, yellow). Next they discuss the application purpose to be partly *navigation* and *communication* between the collaborators. Peter knows that navigational tasks can benefit from switching view points, e.g., from a birds view to a perspective view, as not everybody is able to read a 2D map. Peter then starts with a lecture about interaction design for a 2D map. Laura, agreeing with the idea of changing the viewpoint, interrupts Peter. She reminds him to stick to the higher level of the previously decided multi user aspects (1). They decide that the tool is used while driving in the autonomous vehicle. They make a note on the side of the design space to keep in mind the communication of the vehicle state to passengers, as there is no *driver* performing this action. In the category *privacy*, they get stuck in their discussion. Peter argues that *privacy* is never an issue as people in the vehicle know each other and are willing to share information. Laura highlights that they do not really know yet about the usage of the vehicle. What if the users are sitting in an autonomous taxi that they share with strangers, a *public* context. Peter and Laura agree that this is a critical, not yet decided design factor and needs clarification (3). As they are short on time, they decide to design for a group of friends going to a festival willing to share *personal* data. Two friends want to go to the tent area first while the other two want to go directly to the concert. To take full advantage of 3D AR, Laura suggests that festival organizers could already provide a 3D map of the campsite. By exploring the amenities and the distance to the concerts, discussion between the friends can be improved. The *travel time* will be *short*, requiring a quick and easy system setup and strong decision support. To make it fast and easy to use, it should be part of the vehicles infrastructure to prevent connectivity problems and a setup of external devices. To speed up the decision making process, Peter is thinking about an application that has the *personal* preferences of the single user. The system also knows the time to arrive at the destination and the input of one user that requests a change towards a specific drop off location. Based on this information the system will provide three alternatives for planning the route and parking, which reduces the negotiation time. After having decided the structural boundaries for the system, they proceed with designing the implementation by starting with decisions on the *visualization.* Augmented reality would be ideal, as the passengers can still see each other and discuss on a common map. Virtual reality, Peter explains, could provide advanced spatial knowledge about the destination, but would hinder interaction as it covers the wearer's face (2). A 2D registration is ideal to position a map *centrally*. A shared display influences interior design decisions because it requires users to be able to rotate towards each other. To ensure that everybody is able to see the display the system might need information about the *environment* and add a tick to *information context* (1).

A *placement strategy* might be used to blend between the 2D representation of a map and a 3D point of view image to support spatial knowledge for the desired destination. The *presentation* style should have two levels as it is common from other navigational tools. To make a clear distinction they plan to change from a symbolic visualization to more a naturalistic one in a *binary* way. All *graphic design factors* should be optimized for best viewing conditions in the diverse automotive everyday context. In particular, motion and color could be used to support the group awareness about who requested a change and the three alternative destinations the system suggests.

They decide for *touch* and *gestures* as *input modality* and use well known interaction metaphors. Laura had sorted in other concepts for in-car AR experiences beforehand (Figure 4.2). Hence she suggests to distinguish the input modality from competitors, by adding *behavioral* input (2). For example, as the time frame for the decision is short, the system could detect the single users attention by tracking gaze or head orientation. Peter doubts the practical usefulness of this input modality and suggests to collect more information on this design category. They speculate that *haptic* feedback will

Design Space for In-Car 3D Augmented Reality Applications

be used to support single users interaction with the display. In contrast, *auditory* sound can be used to communicate system states through space that are important to everyone. However, they postpone the detailed work on the interaction and underline it red in the design space (Figure 4.2) (3).

To support spatial awareness at the destination, they want to have the possibility to present 3D images. Peter says that *autostereoscopic 3D displays* are feasible however, they are limited in the number of possible users (2). Therefore they should focus on the integration of *head-worn displays* within a *static area* of the vehicle. The *display factors*, he says are not of major importance for a possible prototype.

Laura is happy with their progress, as it provides a focused description of the system specifications. In a further step she will process the results and create some storyboards showcasing possible design solutions, within the decided characteristics of the system (Figure 4.2, yellow).

4.5 Experience with the Design Process

In the course of the thesis different applications were designed for which the design space was relevant. In Section 5.1 we designed an application that is displayed in AR glasses that shows a warning to prepare the driver of an upcoming take over request. With a floating stereoscopic 3D representation of the highway in Section 5.2 drivers can inform themselves of an upcoming critical traffic situation. The next visualization in Section 5.3 shows hints in a HUD on out-of-view objects. Finally, the design in Section 6.2 presented an explanation to the driver on the windshield. Further information and pictures on the designs can be found in the corresponding chapters. While not all fit in the problem space of the design space for in-car 3D augmented reality applications, the design dimensions are also suitable for applications that use windscreens to display information. All design dimensions of the corresponding design are sketched in Figure 4.3.

While most categories are the same for all designs, there are some outliers. One design (Explanations in Unexpected Driving Situations(Section 6.2)) is aimed for autonomous vehicles, therefore the application purpose differs. Another design (Trust (Section 5.1)) is implemented in AR glasses, therefore the registration or image generation differs. While using the design space, we experienced that through the design process, the purpose was clarified and the limitations have been explored. Nevertheless, designing applications on paper can face complications. For example, when designing for the AR glasses, it was crucial to know the possibilities regarding the field of view. We expected the application to have the potential to visualize in a dimension, similar to windscreens. Since the dimension of the visualization was much smaller than expected, we had to change the design accordingly. Especially when designing for new technology, it is beneficial to first explore the possibilities through practical testing.

Discussion

	Visualization A	À		Vis	sualiza	ation B		C	Visua	lizat	ion C		Vi	sualizat	ion D)	
	User Mode				Single User								Multi User				
User	Observer		Driv	ver (SAE Level 4)					entive C	o-Driver	Driver F			assive Passenger		
	Actor		iver (SAI	E Le	vel 4) Attenti			tive	Co-D	r	Pa	Passenger			Remote Actor		
Context	Application Purpose		Safety		Navigation Vehicl			le N	le Monitoring			ntertainme a	ertainment, Communi and Work			ion Convenience	
	Information Context		Environment			Vehi				le				Personal			Time
	Driving Mode	Driving					Waitinç				ing	1			Parking		
	Level of Automation	Manual				ıl		Semi-auto				omated	ated			Autonomous	
	Privacy			F			Persor				onal				Private		
	Travel Time		Short duration							edium o	luration	<u> </u> L			Long d	uration	
Visualizatio n	Level of Augmentation	Reality					·	Aug	Intented Reality					→ Virtual Reality			
	Registration		Unregistered					2D registered						3D registered			
	Placement Strategy		Noi	ne			Binary			Linear				Exponential/Higher Order			er Order
	Field of View Position	Foveal								Central				Peripheral/Ambient			
	Presentation	S	Symbolic 🖌					+				Naturalistic					
	Graphic Design Factors		Color	Color			Transparency				Siz	ze		Мо	Motion		Depth
Interaction	Input Modality	Fouch & Control			Gestures 0			Ga	aze	ze S			Speech B		Beha	havior Physiological	
	Multimodal Feedback	Ha	Haptic/Tactile			Auditory				Olfactory			S	Sense of Balan			Temperature
Technology	Image Generation	Glasses-Base Displays			ed Head-V			Wo	/orn Displays			Auto: 3D	Autostereoscop 3D Displays		ic	Hand-	Held Displays
	Size	F	Full Space			Full Dimension					Static	Area	a Situa		ated		Dynamic
	Depth	2D								Pseudo-3D				3D			D
	Display Factors	Color Depth				Resolution				Contrast			Brightness			liansparency	

Figure 4.3: Classification of the design spaces of the visualization implemented in this thesis: Visualization A: Explanations in Unexpected Driving Situations (Section 6.2), Visualization B: AR Warnings in a Take Over Request (Section 5.1), Visualization C: Out-of-View Visualization (Section 5.3), Visualization D: Visualization in Stereoscopic 3D (Section 5.2). The colored lines indicate the category that is suitable and most important to consider for the corresponding implementation.

4.6 Discussion

Although research of in-car AR is very limited, previous work on automotive interface design gives us a strong foundation to base this work on. Thereby, we were enabled to transfer and expand insights from other design spaces into an AR-centered design space. We took great care in creating a use case set covering a large field of applications. To this end, we conducted focus groups with experts from AR and automotive domains. The design space can be used as a source of inspiration and to discover gaps of ideas in the current development. Although some use cases can be implemented on other devices, we aim to encourage new interactions with new technologies. However, some design dimensions might not be complete yet. In the course of time, new ideas develop during the practical use of these systems, which lead to applications that may not yet be known to us. For example, bodily effects such as motion sickness in context of AR are still under research. We do not restrain the creation of ideas by reporting current technological limitations, but foster the curiosity to search for the latest trends. The design space is an initial step towards supporting the ideation for novel prototypes. However, we are aware that it might need to be expanded in the future, since design spaces are partial and are subject to change with future technological developments. Due to the limited related work, the design space is mainly based on examples derived from focus groups. The approach of setting up a design space, based on theoretical application possibilities, is exceptional, therefore, this methodology extends the concept of a design space.

4.7 Lessons Learned

The main motivation for this chapter was the exploration of designs for applications for driver and passengers of autonomous vehicles with the possibility to connect the vehicle to the infrastructure or other vehicles. The considerations of existing design spaces and the identification of a design space gap for in-car 3D AR applications, led to further exploration of this knowledge gap. Autonomous vehicles hold the potential for new interactions and unexplored interaction areas. The target group interacting in certain areas could change and thereby the interface changes accordingly. By providing a design space for in-car 3D AR applications, the design creation for applications in autonomous vehicles can be supported and developed in a structured way.

We derive the design space from a comprehensive literature review and 84 novel AR application cases from focus groups. Our results are based on previous knowledge reported in taxonomies and design spaces from the design of vehicle interiors and AR systems. The work on use cases for applications in 3D AR showed that many of those applications need information from the infrastructure, connected vehicles or a stable and fast network connection, but also travel companions, traveling with each other. In our opinion, providing this to autonomous vehicle users is essential to enhance the experience when using in-car 3D AR. This work helps people from all professional backgrounds to develop, analyze and report in-car 3D AR applications.

In this work, we discuss similarities and differences between our use case set and previous work on user interfaces, HUDs and windshield displays and we identify the design parameters that we expect to be of primary importance for in-car AR applications. Using a fictional application example, we validated the design space and show the advantages and the applicability of it when used in a real world design process. Nevertheless, we are certain that the design space will evolve in time when more AR applications are implemented and different categories emerge.

A lesson learned was that the consideration of different design dimensions was necessary and helpful to identify correlations and constraints between dimensions. The concrete development of the details of a design dimensions is best assessed by collaboration. By discussing about it, but also questioning the necessity, different aspects occur and become relevant. For example the necessity for the category *Convenience* became apparent by discussing the difference of it, to the category *Entertainment, Communication and Work*. Designing for an application is not straightforward and problems rather surface by discussing the different aspects of application designs. A design space helps to have a structured approach and to get back on track when deviation in the design processes occur. Especially the constraints of some design dimensions became apparent. If one wants to design a rather active application, that requires a high mobility, the limited degree of freedom must be taken into account, resulting in compromises that need to be made.

By being open about the potential applications, we did not limit the technological applicability, resulting in a very diverse use case set for different user groups. This provides enough space to potentially change and further develop the design space.

To conclude, the main lessons from this work were:

• Explore every possible design dimension with discussions

By design space reflection and exploration, the dimensions increase their validity and meaning.

• Work with the constraints of the design

The constraints of some designs can result in creative solutions. At the same time, by identifying the constraints, impossible or impractical designs can be avoided.

• Stay open for designs with technical limitations

Even though some use cases seem to be not yet relevant with the current technology, new ideas and design encourage solutions that overcome technological limitations. Therefore, in the first step it should not be prioritized to develop with existing technological solutions only.

• Use user stories to explore and reflect upon the design space

User stories help to think of every aspect of a certain design and to put yourself in the place of the designer. With user stories, an imagined design process identifies missing dimensions and constraints and verifies the existing design space.

Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality

Due to better sensors and smart infrastructure information, which gathers traffic data, it will be possible to inform drivers about critical situations that are outside their field of view. Out-of-view visualization is common in several fields such as virtual reality, gaming or aviation, for example with minimaps, object overlay or traffic collision avoidance system (TCAS) [10, 141, 232, 314]. Current highly automated vehicles have a limited sensor detection range. Drivers are informed in a display about the environment, vehicles in the surroundings, the current lane or navigational cues. Situations that are farther away, are not detected by the vehicle.

Smart infrastructure can have a significant impact on the information available to drivers in critical situations. It enhances the identification of the surroundings of vehicles and traffic situations that are not in the own vehicle's sensor range [92, 112, 207, 212]. Connected vehicles get traffic information that is not in close proximity of the vehicle, but might be even several kilometers away. Smart infrastructure can also detect occluded vehicles that are for example around the corner of a house, and thus drivers can be informed about their position, distance and velocity.

On the one hand information about situations that need an immediate reaction, for example in case of events in close proximity of the vehicle, can be provided. And on the other hand, smart infrastructure can provide information on which the driver does not need to react immediately on (non-urgent situations). Depending on the desired reaction of the driver and the automation level of the vehicle, the provided information can have different level of urgency. In both cases, imminent critical situations that require the drivers attention, and non-urgent situations, such as an upcoming flexible construction site, drivers can be informed of the location, context and focus of attention.

For non-urgent situations some considerations need to be taken into account. Drivers can be prepared by giving more information without the need for an instant reaction. In those situations, they have more time and mental resources to prepare the take over request, which leads to a more informed driving decision. When communicating context information, a particular focus should be put on increasing the situation awareness of a driver with it. With good situation awareness, the driver knows the status of the vehicle, where it is located in reference to other vehicles, the velocity of the own and surrounding vehicles, and can also predict the next steps of the surrounding traffic [72]. With good situation awareness it is therefore possible to drive adapted to the situation or to perform better in a take over situation [93, 278, 312]. For a safe take over request drivers have to shift their attention from being immersed in a secondary task to taking over control of the vehicle. Required information for taking over control, like the necessity for an emergency braking, has to be presented effectively.

However, in case of drivers not seeing the cause of a warning, they might not trust the information from the system, take the warning seriously or react adequately on that information. On the one hand, the driving quality depends on the time before a reaction is needed [93, 197, 203]. On the other hand, warning the driver too early, can result in the situation resolving itself or warning the driver of too many scenarios at once. This can cause frustration and loss of trust in the system. Therefore, it is important to find a balance between informing drivers early on, providing enough information about problems, but also to not overload the driver, since this could reduce the driving performance [188].

Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality

In this chapter, it is assumed that drivers still need to take over control of the vehicle, therefore SAE Level 3 [130] of autonomous driving is considered. The implications of fully autonomous vehicles are considered in Chapter 6. Within all sections in this chapter, the potential of visualizations that indicate critical traffic situations and the driver's benefit thereof are explored. Many researchers rather focus on the optimal take over time and getting the driver as quickly as possible back in control of the vehicle [93, 197, 203]. Different to that, we include augmented reality information of an early warning in a take over scenario. We implemented several visualizations that prepared drivers with information about a upcoming situation. First, we focused on how to visualize that information. We evaluated, if the performance improves, and if the situation awareness increases by providing the driver with a representation of the highway. Next, we conveyed information about the critical situation in a non-intrusive manner. Since secondary task become more likely when driving autonomously, drivers may want to stay immersed in this task. With this visualization it is possible to get hints about upcoming take over situations in a HUD.

We contribute with this chapter a concept of a 3D AR representation of the highway for smart infrastructure information to improve the take over performance and an evaluation of design concepts for early warnings on head-up displays to improve situation awareness.

This chapter is divided in the following sections:



Sketching Context to Improve the Trust in the Driving Decision

The first aspect to consider is how smart infrastructure warnings and information can be conveyed to drivers. In Section 5.1 we investigate how to sketch context information and to prepare drivers for upcoming critical situations. In an informal case study, consisting of one survey and two studies, we examined how to visualize and interact with situation information from the highway.



Preparing Take Over Requests in a Stereoscopic 3D Representation

In Section 5.2 the concept of providing driver with highway information before the actual TOR is investigated. We provided participants with a miniature digital twin in stereoscopic 3D which showed upcoming critical situations and tested the TOR for performance, workload and user experience. The representation enabled participants to react earlier and the performance of the TOR was improved.



Visualizations that Look Beyond

In Section 5.3 a visualization in the windshield that prepares drivers for upcoming take over maneuvers is explored. The visualization gives hints on upcoming possibly critical situations and other traffic participants in some distance. Three visualization concepts were tested for situation awareness, user experience and performance. Wedge visualization had the best situation awareness score, but overall a symbol visualization resulted in the best performance.

5.1 Sketching Context to Improve the Trust in the Driving Decision

In the following, we explore visualizations that inform drivers of situations that are outside their current view. In Chapter 3 we identified drivers as a user group for infrastructure information. We do not know yet how smart infrastructure information can be presented to them and how to convey context information. For example, when an accident is detected, this information could be communicated, including the number of vehicles involved, the lane on which the accident occurred and if a traffic jam is forming. Furthermore, it is not clear if participants use context information to make an informed driving decision. This information needs to be conveyed to drivers in an understandable and trustworthy manner. Building trust is important to establish acceptance for systems, and presenting information about upcoming traffic situations in AR can increase this trust. To explore the potential problem space, we first prototyped visualizations and explored the potential of presenting smart infrastructure information early on.

In an informal case study, we explore visualization schemes and presented participants implementations in AR glasses. This section is organized as follows: First, we implemented critical situations in a simulation environment. Then, we sketched possible warnings and distinguished between sketch, picture and video of the situation that could be displayed to drivers. In the next step, in Section 5.1.1, we implemented a warning using AR glasses to prepare drivers for a potential take over request.

Simulating Critical Situations In the initial step to provide users with context visualization of traffic situations, we first implemented possible critical situations in visualizations in a simulation environment (see Figure 5.1). This digital twin, a digital representation of the whole highway, is simulated with Carmaker¹, in which traffic objects, for example, trucks, vehicles, pedestrians or traffic signs such as velocity limitation or gantries can be simulated. The traffic objects are assigned with attributes, such as autonomous driving, length, velocity or braking behavior. In this simulation environment, traffic situations can be prototyped. With these visualizations, a better understanding of possible visualizations is build.

¹ https://ipg-automotive.com/products-services/simulation-software/carmaker/, last access April 2020

Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality



Figure 5.1: Example use cases and prototypes of the digital twin of the highway. A: Birds eye view with warning symbol, B: Construction site, C: Standing vehicle on the side shoulder, D: Beginning of a traffic jam, E: Simulated traffic scene, F: Real image of traffic scene in E.

Context Visualizations in a Sketch

This section is supported by the semester project of Antoine Poppe and Ahmed Zitoun. Please refer to the beginning of this thesis for a detailed statement of collaboration.

The use cases of the stakeholder (see Chapter 3) led to the conclusion that with smart infrastructure information, drivers could inform themselves about upcoming critical traffic situations long before it was necessary to react. Therefore, complex visualizations and information could be taken into account to inform driver. We chose the six scenarios identified in Section 3.1.2 to evaluate the type of context presentation: bad weather, merging on highway, ghost driver, fast vehicle from behind, standing vehicle and abrupt braking. Some established visualizations from those situations exist, for some we sketched the situation (see Figure 5.2). In an evaluation we asked participants (N = 8), if they understood those visualizations. We asked them, what happened in those situations and in what kind of distance the scenario took place.

Next, with the resulting feedback, we sketched possible context information on paper and asked participants (N = 20) which condition they preferred. We had three conditions: symbolic presentation, pictures combined with a symbolic presentation and a short video of the scenario with a symbolic presentation. The pictures and videos were taken from the previous implementation of traffic situations in the simulation environment.

Overall, the symbolic presentation was preferred by 46.72% of the participants. The video presentation was favored least (18.06%). The visualization of the ghost driver was not well understood, in

this scenario the symbol had the same preference rating as the video (28.60%), but the most preferred visualization for this scenario was the picture combined with a symbol. We can derive from this that overall the sketched context information have a good acceptance among the participants. We used those findings in the next step to implement a scenario of an accident in a simulation environment and implement a warning thereof in AR glasses.



Figure 5.2: Sketches for visualizations of critical situations: Sketch A: Merging, Sketch B: Bad Weather, Sketch C: Ghost Driver, Sketch D: Fast vehicle from behind.

5.1.1 AR Warnings in a Take Over Request

This section is supported by the Bachelor's thesis of Dominik Hiemer. Please refer to the beginning of this thesis for a detailed statement of collaboration.

 2 In a pre-study to derive a visualization that can later be used to warn drivers of an upcoming critical scenario, we just visualized different warnings in AR glasses. Eleven participants tested six warning visualizations of an accident in AR glasses. Two visualizations were in 2D, two visualizations in 3D and two visualizations were animated, one in 2D, the other in 3D. The not animated visualizations can be seen in Figure 5.3. The warning consisted of the information that an accident occured due to bad weather. In the animated visualization, the vehicle drove to the accident and then changed lane. The bad weather was animated rain. These sketches of scenario visualizations were then presented to participants in a HoloLens. We asked the participants how they would react when seeing this visualization. In the vertical visualization, 72.7% of the participants would change the lane. 54.5% of the participants stated that the animated vehicle did not help their understanding of the accident situation. 33% of the participants did not understand the animated rain. 90.9% of the participants found it harder to understand the visualization in 2D Overall, this evaluation confirmed our assumption that the best visualization to convey context information would be the 3D vertical visualization. We conducted a study to explore, if the visualization of an accident scene increases trust in the vehicle.

² This section is supported by the Bachelor's thesis of Dominik Hiemer. Please refer to the beginning of this thesis for a detailed statement of collaboration.

Method:

The independent variable were the visualizations in the AR glasses. The first condition was the baseline. The participant watched a video as a secondary task for 40 seconds. Then the video disappears and after 10 seconds, a TOR is initiated and the participant takes over control of the vehicle. In the second condition, the participant also drives autonomously for 40 seconds, then the 3D vertical visualization of the accident is displayed in the HoloLens for 10 seconds. Then a TOR is initiated. The dependent variable is the trust in the vehicle. 20 participants (f = 8, m = 12, 19-76 years) took part in the study. The study was conducted in a driving simulator with one monitor. The simulation environment was Unity 3D (Version 2017.3.0f3). Before the study, the participants were asked on a 6 point Likert scale if they generally trusted autonomous systems, if they would enter an autonomous vehicle and if they would hand over control to the autonomous vehicle. Then they got accustomed to driving in a driving simulator by testing it without the HoloLens. After the conditions, we asked six questions on a 6 point Likert scale (1-not at all, 6-extremely) based on the trust evaluation of Jian et al. [136]. The two questions that evaluated distrust were: I am suspicious of the system's intent, action or outputs and The system's actions will have a harmful or injurious outcome. The four questions that answer the trust in the system were: I am familiar with the system, The system provides security, The system is reliable and I can trust the system.

Additionally the participants answered for which system they trusted themselves most, to take over control of the vehicle successfully and safely. Furthermore, if their trust increased through the system, for which drive they had the feeling to take over control of the vehicle safest and if their driving behavior is improved through the visualization on the HoloLens.



Figure 5.3: Visualizations of warning drivers of accidents with context information in AR.

Results: Before the study, 50% of the participants generally did not trust autonomous systems (SD = 0.51). They preferred to keep the control of the vehicle. One participant was unsure if the system would react as well as a human driver. The participants that trusted the autonomous vehicle mentioned the advantage of saving time with autonomous vehicles. One participant mentioned that s/he would just trust autonomous systems if it was 100% reliable and that in case of an accident the liability for damages would be clear. The data for the trust score is not normally distributed, tested with the Kolmogorov-Smirnov test (p=0.98). The average scale for trust was with the visualization at 4.53

points (IQR = 2) and at 4.26 points (IQR = 2) without a visualization. A paired T-test revealed statistical difference in the trust scales between the two conditions (p<0.001). In the end of the study, 85% ($\sigma = 0.51$) of the participants indicated that they trust the system.

This study indicated that a visualization of a traffic scenario could improve trust in autonomous vehicles. We gained insights in visualizations of scenarios and the effects of AR glasses during a take over situation.

5.2 Preparing Take Over Requests in a Stereoscopic 3D Representation

This section is based on the following publication:

• Gesa Wiegand, Christian Mai, Yuanting Liu, and Heinrich Hußmann. 2018. Early Take-Over Preparation in Stereoscopic 3D. In Adjunct Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications (AutomotiveUI '18).

This section is supported by the Master's thesis of Hua Wentao. Please refer to the beginning of this thesis for a detailed statement of collaboration.

From the previous section, we learned that context information, such as a ghost driver on the highway, is difficult to visualize without overwhelming the driver. Take over requests (TORs) are issued in case of critical situations ahead or in case of system limitations of the autonomous vehicle. Due to sensor limitations and the detection of a critical situation, the driver needs to react quickly on an issued take over request. This section focuses on the driving situation before the TOR, so the preparation phase. A novel output interface is presented, to prepare the driver of a TOR. It is investigated, if a floating 3D AR representation (or S3D) of the real traffic situation within the cockpit, improves the driver's take over performance.

In industry, the goal to include a 3D representation in the vehicle is a comfortable and intuitive interaction with the vehicle³. By visualizing the traffic situation in a bird's-eye view, the driver is able to assess the appropriate driving trajectory or route. By displaying an as accurately as possible representation of the highway the driver gets more information about possible limitations on the road. We put a focus on conveying context information without limiting the amount of provided information. This means, drivers get an accurate and complete representation of the highway with the possibility to interact with it. This representation is called digital twin in the following. With this representation of the digital twin it is possible to change the viewpoint of the highway, jump to critical situations, get distance information of the critical event to the own vehicle and an overview of other vehicles on the highway. For construction sites, it would then be possible to get an overview of the construction site, its length and active construction site vehicles. Furthermore, it is possible to inquire if a traffic jam is starting, the extend of it and if it makes sense to change the route.

³ https://www.continental.com/en/press/press-releases/2019-06-11-3d-instrument-cluster-174836, last access May 2020



Figure 5.4: VR Cockpit with digital twin and representation of the own vehicle. The secondary task, which is used in the study is placed near the A-Pillar, above the steering wheel. The digital twin is floating above the gearshift in between driver and passenger. Within it, one can see the surrounding traffic situation. The green A indicates the autonomous mode.



Figure 5.5: Pre-Warning of accident on own lane. The mode indication turns yellow since a take over request is imminent.



Figure 5.6: Take over request with video task turned off, the proposed driving trajectory is display next to the manual mode indication.

By showing the driver an accurate traffic representation, the driver is able to get more information to plan the driving trajectory of the vehicle. The following study considers a vehicle with SAE Level 3 automation [130]. Most of the time, the vehicle drives autonomously, but the driver needs to take over the vehicle if the autonomous system fails.

We present a study to increase situational awareness and thus improving the driving trajectory planning. With this study, we assess the workload, usability and performance when using the representation of the highway.

5.2.1 Method

A study (N = 20), comparing the proposed system to a system without a possibility to foresee the situation, was conducted to gain insights into driving performance and workload of drivers. The digital twin appeared before the take over request. A possible reason for a take over request, such as

bad weather or an object on the road, is displayed in the digital twin. Drivers can interact with the digital twin before a take over request to inform themselves about the traffic situation on the highway.

To test the system there were two conditions that the participants experienced. One condition served as a baseline and no digital twin was displayed. The second condition (S3D Visualization Condition) included the AR representation of the highway. To eliminate possible learning effects due to this setup the participants were divided up into two groups. Both groups get the same introduction to the setup. Group 1 experienced the baseline condition first and then the S3D Visualization condition and for Group 2 the order is reversed. Both groups did a preliminary driving test to get used to the system. The study is designed as a within-subject study, so every participant performed both driving tasks. The system was evaluated by the NASA TLX [113] questionnaire and an acceptance scale questionnaire [157]. To evaluate the effect of the system on the driving behavior, the steering wheel angle, velocity and the position of the vehicle were tracked.

Participants

20 participants (f = 8, m = 12) with a mean age of 25.3 (STD = 2.2) took part in the study. Most (80%) have a driving license and one to three years of driving experience. 60% of the participants never used VR glasses.

Apparatus

To be able to simulate the S3D interface and for an easier implementation, the study was conducted using a VR head-mounted display based driving simulator. Within the VR environment, the driver sits in the cockpit of a highly automated vehicle. Inside the cockpit of the vehicle an AR representation of the highway is displayed above the gearshift (see Figure 5.4). The digital twin displays a sector of the highway the own vehicle is driving on. The driver can move the section in the digital twin forwards and backwards by pressing a button. They can center the digital twin again on their vehicle if they want to focus on the situation around the own vehicle. The distance of the current sector of the highway to the own vehicle is shown as a floating digit next to the highway. The virtual environment was set up using Unity 3D (Version 2017.3.0f3)⁴.

Procedure

Before the study, the participants signed consent to collect their data. Then they filled in a demographic questionnaire. During the introduction the AR representation of the highway and the functions of the setup were explained. In the next step, the take over process was explained. The S3D representation of the highway was displayed to drivers before they needed to take over the driving task (Figure 5.4). They could scroll through the whole traffic situation and change the view to get an overview of the current situation on the highway. With the distance information, the driver knew where the possibly dangerous situation was in respect to the own vehicle. After informing the driver of the highway situation with the S3D (Figure 5.5) the TOR itself is executed (Figure 5.6). The take over time was set to 10 seconds. Within the study there were four phases:

⁴ https://unity.com/, last accessed May 2020

Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality

1. Autonomous Driving During this phase the vehicle controls the steering. The driver concentrates on the secondary task. The secondary task consists of watching a movie and counting objects within the movie.

2. Information Phase This phase begins by a sound to inform the driver that the digital twin is displayed. Drivers can interact with it and thus inform themselves about the current traffic situation. The secondary task is not available to the driver anymore. This phase lasts 10 seconds.

3. Take Over Phase A take over request is issued when the own vehicle has a time to collision (TTC) of 10 seconds. A warning sound initiates this phase, and the driver is prompted to take over the steering wheel. At the end of the take over phase, the digital twin disappears.

4. Manual Driving After the driver regains control of the vehicle, the driver needs to decide on the driving maneuver. To avoid a collision with the objects on the road, the participant needs to change lanes.

After getting used to driving in the simulations, participants experienced the following conditions:

Condition Baseline	Condition S3D Visualization
The first driving test is the baseline of the study.	During the S3D Visualization condition the intro-
The driving phases consist of Phase 1 - Phase 3 -	duced system is used. Different to the baseline
Phase 4.	condition, an additional phase is added. The driver
	experiences Phase 1 - Phase 2 - Phase 3 - Phase 4.

After every condition, the dependent variables were measured through a questionnaire.

5.2.2 Results

In the course of this study, the performance during the take over maneuver, the workload of the participants and their user experience was assessed. In the following we present the results of the study. To assess the significance of the results a paired t-test, Friedman test [200] and Wilcoxon Signed-Rank Test [236] is used.

Take Over Performance

We compared the driven trajectory of the baseline and the S3D visualization. To assess the driven trajectory, the lateral lane position is recorded. As the driving task requires the driver to change the lane the objection was to drive on the lane without the obstruction on the road. In the S3D condition the lateral lane positions were generally closer to the lane without an object (Mean = -108.31, SD = 1.15) and results obtained by a t-Test revealed a significant difference between the baseline condition (Mean = -109.51, Sd = 1.08) and the S3D condition (p = 0.04). Additionally the steering wheel angle of the driving maneuver was recorded. This gives us information about the steering behavior of the driver, like uncontrolled or smoother lane changes. Though the S3D condition has overall smaller values with a mean of 10.24° and a SD of 4.82° and the baseline condition has higher values with a mean of 15.86° and a SD of 8.45° , the p-value indicates no significant difference between the two conditions. As a third evaluation measure the speed was taken into account. The speed in the S3D

condition was lower (Mean = 3.82 m/s, SD = 5.93 m/s) than the baseline condition (Mean = 8.63 m/s, SD = 10.72 m/s), which indicates that the user adapted the speed earlier and more appropriately with the S3D visualization. The p-value of a t-Test is 0.04, indicating significant differences in the speed values in both conditions.

Workload and User Experience

The workload of the system was assessed using the NASA TLX (see Figure 5.7) on a 20-point Likert scale. The temporal demand of the S3D condition is highest (Mean = 7.65, SD = 5.27), which indicates that the participants needed to spend time on the digital twin to understand the digital representation. The biggest difference between both tests is the effort value (Baseline condition: Mean = 11, SD = 1.02; S3D condition: Mean = 5.6, SD = 4.86). That could be an indication that the digital twin mentally prepares the driver better on the TOR and thus lessens the effort of the TOR. We tested the data on significance with a Friedman test [200], since the data is not normally distributed (tested with the Kolmogorov-Smirnov test [63]). The measurement scales of mental demand ($\tilde{\chi}^2(1) = 7.2$, p = 0.007), performance ($\tilde{\chi}^2(1) = 4$, p = 0.046), effort ($\tilde{\chi}^2(1) = 14.22$, p = 0.0002) and the overall measure ($\tilde{\chi}^2(1) = 11.84$, p = 0.0006) show a significance (mental demand (p-value = 0.007), performance (p = 0.037), effort (p = 0.0004), overall measure (p = 0.02314)). The S3D condition has a lower score in most measures and performs better than the baseline condition.

The acceptance scale is used to analyze the acceptance of the digital twin. The acceptance is measured by evaluating the usefulness and satisfaction scale. Participants have a positive attitude towards the digital twin with a mean of 1.06 (SD = 0.76) in the usefulness and a mean of 1.16 (SD = 0.80) in the satisfaction scale. One remark of a participant indicated that the understanding was better, when a TOR was issued and in assessing the proper point of time to take over the vehicle with the S3D visualization.

5.2.3 Discussion

The results show that a visualization of the digital representation of the highway can improve the driving performance. The driven trajectory was significantly improved by the S3D visualization compared to the baseline. Furthermore, the speed was significantly lower with the S3D visualization. Drivers adapt their speed earlier and more appropriately with a S3D visualization. Additionally, the workload is significantly lower regarding the mental demand, performance, effort and the overall measure. The participants generally have a high acceptance of the S3D visualization. Overall it can be concluded that a digital twin in S3D does prepare drivers of upcoming take over requests. This results in a more foresighted, safer driving.

Limitations In this study it is assumed that the driver has enough time to interact with the digital representation of the highway and that therefore, an overview of the traffic situation can be gained. While the overview completely represents all traffic objects, the placement of it forces the driver to look away from the road. If drivers need to take over control of the vehicle, first they need to turn towards the steering wheel again. Therefore, this visualization would just be beneficial if drivers have enough time to react on the critical situation. It is questionable that a digital twin would be very helpful in case of a high frequency of take over requests.



Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality

Figure 5.7: Workload scores of the NASA TLX. The error bars show the standard error of each category.

With an automated detection of critical events, drivers could still inform themselves about the critical situation, how the situation looks like and how the traffic is reacting upon it. Emergency vehicles could get information, if a rescue lane is formed, if other emergency vehicles are on their way and the location of the accident.

Nevertheless, for smart infrastructure representations for drivers, many questions remain. For example, if it is possible to increase the situation awareness of drivers in a more non-intrusive manner, without having a complicated real world representation. The digital twin is not abstracted and the context information needs to be derived from the representation itself. In the next section, we therefore look at abstract visualizations of upcoming situations that indicate that a critical situation is ahead.

5.3 Visualizations that Look Beyond

This section is based on the following publication (submitted to AutoUI'20):

• Gesa Wiegand, Beatrice Sax and Heinrich Hußmann. 2020. Looking Beyond the Visible: Exploring Out-of-View Visualizations in Cars. *Submitted to WiP Adjunct Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications* (AutomotiveUI '20).

This section is supported by the Bachelor's thesis of Beatrice Sax. Please refer to the beginning of this thesis for a detailed statement of collaboration.

In the previous section we explored to prepare the driver of a take over request with a digital representation of the highway. While this approach is valid in some cases, in other cases, in which the driver quickly needs to take over control, other visualizations are more appropriate. While environment information is implemented in most current vehicles, in which the own vehicle and surrounding vehicles are displayed, smart infrastructure offers the possibility to visualize vehicles that are further away.

The challenge that remains, is visualizing the vehicles in a non-intrusive way without loosing distance information or how the vehicles could influence the own driving behavior. When not using a digital twin to display all vehicles in the surrounding, out-of-view visualizations could offer the possibility to give hints on vehicles and possible critical events. The goal of this section is a visualization that delivers environment information in an understandable manner to improve situation awareness.

To identify, which visualizations are best to increase situation awareness in situations in which the cause of the warning is outside the view of the driver, we design a warning interface. That conveys the information of a dangerous situation outside the view of the driver. The hypothesis is that by informing the driver of the surroundings, situation awareness can be improved. In the last section we showed that driving performance is already improved by a digital representation of the highway, therefore, we hypothesis that situation awareness can be improved as well.

To test this hypothesis, we first set up a scenario catalog (see Table 5.1) to identify the possible use cases and constraints, such as the time the driver needs to be able to brake in time. To get an understanding of possible visualization concepts in situations in which the driver is warned early of dangerous situations, we analyzed existing prototypes in automotive and other domains. Furthermore, we conducted two focus groups (N = 9) in which we asked participants to sketch a visualization. Next, we implemented four different visualizations and conducted a study (N = 24). The visualizations were examined according to their improvement in situation awareness, workload via the NASA TLX [113], System Usability Scale (SUS) [34] and the braking performance of the participants.

5.3.1 Identification of Visualizations

Our objective is to find a visualization that informs about a dangerous traffic scenario out of the driver's view that increases situation awareness. In four steps we approached the identification of possible warning visualizations:

ID	Scenario Description	Short	Long
		distance	distance
	Letting vehicles merge	x	
2	Francijam on exit		X
3	Detailed traffic jam information		х
4	Warning of obstacles on the road	х	х
5	Warning of accidents	х	х
6	Early formation of a rescue lane	х	х
7	Warning of fast cars	х	
8	Instructions for zip merging	х	
9	Road condition	х	х
10	Improving awareness in bad visual conditions	х	
11	Exit planning	х	
12	Braking cascade	х	
13	Warning of a vehicle that suddenly shears out	х	
14	Early warning of slow driving cars in the middle lane	х	х
15	Foresighted driving to reduce fuel consumption and emissions	х	х
16	General anticipatory driving in relation to other road users	х	х
17	Warning of defective vehicle/non-reactive user/Safe Stop	х	х
18	A look ahead over the hilltop ahead	х	
19	Lane - Departure - Warning for legacy cars	x	
20	Wildlife crossing detection	x	х
21	Warning of vehicles on released hard shoulder	х	х

Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality

Table 5.1: Scenario catalog in which the scenarios are sorted in ones that are relevant for warnings in short and long distances.

х

х

х

1. Set up a scenario catalog

22

23

24

- 2. Collect visualization ideas through literature review and two focus groups (N = 9)
- 3. Test prototypes in a pre-study

Capturing information of mobile traffic signs

Persons (on bridge) throwing objects onto the carriageway

Early lane change in case of an obstacle on the road

4. Implement the visualizations and conduct a study in a driving simulator

Scenario Catalog A scenario catalog of 24 scenarios was set up, of highway scenarios that infrastructure sensors could detect and the driver could be warned about early on. We got these scenarios from two brainstorming sessions with experts, working in infrastructure projects (N = 4, m = 3, f = 1). The participants were between 25 to 35 years of age. The scenarios were sorted in those that the user can be warned of early on (in a distance of more than approximately 5 minutes) and those to which the driver should respond more directly (approximately 2-5 minutes). The scenario catalog gives an outlook, on the potential of the early warning visualizations, but is just a first draft and can be extended over time.

Focus Groups and Literature Review To develop a new visualization concept of objects and traffic participants that are outside the visual periphery of the driver, we conducted a literature review and two focus groups. The literature review revealed concepts to visualize objects that are outside the periphery of the user in augmented reality (AR) or virtual reality [26, 100, 252]. Most computer games use minimaps to give the player an overview of the surroundings and visualize objects in some



Figure 5.8: Sketches for the pre-study.

distance [178]. In the further course of the work we concentrated on the results and visualization ideas of wedges [106], EyeSee360 [99] and arrows [252]. In two focus groups we collected visualization methods for driving scenarios that are outside the view of the driver. One focus group (N = 6)consisted of non experts, the other (N = 4) of design students. We wanted to explore different ideas from experts and non experts and chose the focus groups accordingly. The participants were between 20 and 35 years of age. We started the focus group by showing a video of a vehicle that needs to do an emergency braking because vehicles in the front do an unexpected lane change. Next the participants made a sketch, how they would warn the driver in this situation, with manually driven vehicles in mind. Then we varied the distance of the scenario to see if the sketch would differ if the driver had more time to react on this situation with highly automated vehicles in mind. In the end there was an open discussion about possible different modalities, uncertainty visualizations and if drivers should even get the information about situations that are further away. The participants identified six cockpit elements(e.g. Steering Wheel, Sound or Light), nine different visual methods (e.g. arrow on vehicle ahead, ghost overlay on windshield display (WSD), bird's-eye view of scenario), 19 overall methods to warn the driver (e.g. colored WSD framing, red flashing or pump seat upright) and three different warning modalities (e.g. Vision). A warning symbol was mentioned most often. Most participants sketched some sort of environmental model or minimap in which the scenario can be visualized. Others sketched information on the whole windscreen or used the frame of the windscreen for displaying the location of the object.

Pre-Study In a pre-study (N = 9) we sketched possible visualizations (see Figure 5.8) to collect opinions on those. Five paper prototypes visualized a minimap, icons, wedge, EyeSee360 and a combination of the minimap and the wedge. Each paper prototype consisted of three frames, which should reflect a period of about 30 seconds to show dynamic functions of the interface components. The participants were asked to rate the visualizations on a 5-point Likert scale, where 5 was the best score. The minimap got the highest and the EyeSee360 visualization got the lowest score. The icons and the combination of minimap and wedge got good results (3.44 and 3.33 points). The wedge got



Figure 5.9: Implemented visualizations, from left to right: Baseline, Arrows, Symbol, Wedge.

a rating of 2.67. Due to the 2D inanimate visualization the participants had difficulties to understand the concept of the EyeSee360 visualization. Based on the feedback we got from the participants we decided not to use the EyeSee360 visualization in the following study, but to implement the minimap, wedges, a warning symbol and arrows. With those visualizations the distance of the scenario could be visualized, the urgency and in case of the wedge and the arrows the location of the other traffic participants.

5.3.2 Method

After defining and implementing the scenarios, we conducted a study to evaluate the visualizations. The dependent variables were workload, situation awareness, system usability scale and performance. We used a within-subject design with visualization as the only independent variable resulting in four conditions. The conditions are counterbalanced by a latin square design, so all participants had differently ordered visualizations. The study lasted about 30 minutes per participant. We concentrated in this study on scenarios that are in closer proximity of the driver, but are occluded or out of the view of the driver. We determined two key scenarios from the scenario catalog (see Table 5.1) that were used in the study: 1) Fast vehicle from behind 2) Braking cascade. We chose those scenarios since the participants can relate to those scenarios easily and since both scenarios result in many accidents or traffic jams on the highway. These two scenarios were implemented two times, but slightly differently to avoid learning effects. In the first two scenarios is a fast vehicle, invisible to the driver, either from behind or the front. The own vehicle would try to overtake the front vehicle resulting in an accident. Therefore, the participant needs to brake to prevent it. Those braking incidents were recorded. In the other two scenarios, there is a braking cascade which is caused on two different intersections. The participant would need to brake to prevent an accident. In the baseline visualization just the minimap is displayed. The minimap shows the environment of the own vehicle from a top down view with a radius of about 25 m. The own vehicle is marked with an arrow. In case of a dangerous scenario that requires braking action from the participant, the involved vehicles are colored red. Using this as the baseline of the visualizations, we add the different visualizations on top. First, we add arrows on the side of the minimap. They point towards the vehicles in the distance and change the color in case of a dangerous action of that vehicle. The next interface shows a warning sign in case of a dangerous situation. The last interface is similar to the visualization with the arrows and displays wedges that



indicate the position of other traffic participants. All participants drove through eight scenarios and tested four visualizations: baseline, wedge, arrow and symbol (see Figure 5.9).

Figure 5.10: Driving simulator in the study setup.

Participants In the study we had 24 participants (f = 11, m = 13). All participants had a driver license. 13 participants had much experience with video games and 14 participants had experience with VR glasses.

Apparatus The study was conducted in a driving simulator (see Figure 5.10). The simulation environment was Unity 2019.2.8f⁵. The scenarios were shown on three screens in front of the vehicle. The visualizations were placed on the windscreen in front of the driver. Thereby it is directly connected to the primary driving task such as maneuvering and reacting to critical situations [268]. The participant was able to brake in case of a dangerous situation.

Procedure Before the study, the participants filled in a data protection sheet and the questionnaire. Then the participants drove first through two scenarios with one of the four visualizations. The two scenarios consisted of one of the two variants of the scenario of a fast vehicle approaching and of one of the two variants of the braking cascade. All scenarios lasted until shortly before the time to collision was zero. Then just a black screen was shown. To assess the situation awareness we did not show the resulting possible accident. After driving through two scenarios with one of the four visualizations the participants assessed the system usability scale, workload and situation awareness. This was repeated for all visualizations.

⁵ https://unity.com/, last accessed May 2020



Out-of-View Visualization of Highly Automated Vehicles in Augmented Reality

Figure 5.11: Situation Awareness Levels and overall SAGAT score.

5.3.3 Results

The study was assessed according to the participant's situation awareness, workload, system usability and performance. We first tested the data on normal distribution with the Kolmogorov-Smirnov Test [63]. The data for the situation awareness, workload, system usability and the braking times is not normally distributed. Therefore, we used the Wilcoxon Signed-Rank Test [236] to test for significance in the results.

Situation Awareness Situation Awareness (SA) is assessed by using SAGAT, measuring each level of SA by two questions. We asked about the positions of the other traffic participants and the color of their vehicle (Level 1), about the distance to the other traffic participants (Level 2) and about the future driving behavior of the own vehicle and of the traffic scenario (Level 3). Altogether six questions were asked. From the participant we took the mean of the answer concerning the level of situation awareness. Then we had one value for each participant for each of the three levels of SA. We tested the significance of the differences of the conditions for each level (see Figure 5.11). For the third level, there are some significant differences: symbol (Mean = 1.71, IQR = 0.5) between baseline (Mean = 1.83, IQR = 0.5) and arrow (Mean = 1.65, IQR = 0.5) (p = 0.04) and wedge and baseline (p = 0.001) Participants were more aware of the surroundings and the future development of the scenario with the wedge visualization.

Workload We tested the workload (see Figure 5.12) for statistical difference between the conditions. The subjective workload was highest with the wedge visualization (Mean = 31.35, IQR = 27.5) and lowest with the symbol (Mean = 23.61, IQR = 24.17). This difference is statistically significant (p = 0.012). The baseline (Mean = 25.76, IQR = 17.08) was second highest, followed by the arrow



Figure 5.12: NASA TLX Scores. The error bars show the standard error.

visualization (Mean = 25.42, IQR = 23.75). Looking at the data, the cause for the statistical difference between symbol and wedge is revealed. The mental demand of symbol (Mean = 30.83, IQR = 45) and wedge (Mean = 41.25, IQR = 46.25) is statistical different (p = 0.03). Furthermore, the temporal demand between symbol (Mean = 26.25, IQR = 32.5) and wedge (Mean = 36.04, IQR = 47.5) is statistically different (p = 0.04). The effort of symbol (Mean = 17.92, IQR = 15) and wedge (Mean = 27.08, IQR = 31.25) is statistically different (p = 0.009). The physical demand between wedge (Mean = 19.38, IQR = 21.25) and baseline (Mean = 12.71, IQR = 10) is also statistically different (p = 0.04). Between symbol (Mean = 17.92, IQR = 15) and arrow (Mean = 24.58, IQR = 25) and wedge (Mean = 27.08, IQR = 31.25) and baseline (Mean = 20.42, IQR = 20), the effort is statistically different (p = 0.01 and p = 0.03).

The results indicate that the wedge visualization requires more workload from the driver to understand than with the symbol condition. The low NASA TLX value for the symbol can be explained by the previous experience of the participants of what to expect when seeing the warning sign. There is a trend towards a higher workload of the baseline compared to the symbol. That can be explained as upon seeing the baseline visualization, the participants still have to put some effort into interpreting the visualization. Since the baseline visualization did not show much more information compared to the observation of the environment it could be rather distracting to the driver to observe the Minimap visualization. This could lead to the conclusion that the Minimap should be more minimalistic to be less distracting.

Usability The usability, measured by the SUS, of the symbol was highest (see Figure 5.13). Since scores above 80% indicate good to excellent usability, the baseline visualization also has an excellent usability score. We tested the SUS Scores on differences between the conditions. The Friedman test [200] revealed significant differences between the conditions ($\tilde{\chi}^2(3) = 11.88$, p = 0.007).





Figure 5.13: System Usability Score.

There is a significant difference between symbol (Mean = 83.4, IQR = 8.75) and wedge (Mean = 68.8, IQR = 15.63) with p = 0.002 and wedge and baseline (Mean = 80.1, IQR = 17.5) with p = 0.05. This was also mirrored in the feedback of the participants who thought that wedges are not easy to use.

Performance The performance was measured by tracking the braking behavior of the participants. We calculated the braking time with the mean time of the two conditions with one visualization. If no brake was applied in one condition, just the condition with braking time was considered. We had to exclude the data from participants in which the braking time was more than 5 seconds before the optimal braking time or the standard deviation of the data was more than three times the mean. Then we assume the braking point, at which an accident can still be prevented. A negative number indicates that the participants braked earlier than the optimal braking point. The moment at which braking is required to prevent a collision is zero. The braking times were the following: baseline (Mean = 0.725 s, IQR = 1), arrow (Mean = 0.6 s, IQR = 1.13), symbol (0.35 s, IQR = 1) and wedge (Mean = 0.6 s, IQR = 1.13). We used the Friedman test to test for significance. The braking times are not statistically significant ($\tilde{\chi}^2(3) = 2.2931$, p = 0.51). Nevertheless, on average the point of time for braking is later than the optimal braking time. In the baseline condition the participants brake latest.

5.3.4 Discussion

The key findings of the study can be summarized by following: A wedge visualization had the best situation awareness score, indicating that better situation awareness can be achieved. Furthermore, established visualizations such as a warning sign overall perform best. By informing drivers of their surroundings we assumed that the situation awareness could be improved. There was a statistically significant effect of the wedge compared to the symbol visualization. So even if the score for wedge for the system usability was low, the situation awareness could be improved by indicating early on the location and distance to a dangerous situation. Therefore, further research is needed, but our results are promising that especially situation awareness can be improved by combining the environment map

with other visualizations. Although the SUS was quite low for visualizations in which the participants had to interpret the required action themselves, such as arrow and wedge, it would benefit the situation awareness to indicate a problematic situation on the road.

A warning symbol indicates that immediate action is required, but if the transmitted information is not urgent it would cause stress to get the attention of the driver with a warning symbol. Smart infrastructure enables information about scenarios that are further along the route and therefore do not need immediate attention. It would be even safety critical if a mere information would be misunderstood as a safety critical warning, resulting for example in sudden braking that could lead to an accident. Therefore an increased focus should be put on improving the system usability and workload of the visualizations that indicate location and distance of other out of view objects. Since there is much potential of displaying information in future vehicles, for example through windshield displays, new visualization concepts can help providing information about traffic situations outside the periphery of the driver.

5.4 Lessons Learned

The goal of the research presented in this chapter is to understand how information from a smart infrastructure can aid drivers to improve their driving performance during take over maneuvers. Thus, the fundamental underlying questions that we answer are:

Does warning with smart infrastructure information differ from warnings from the sensor system of the own vehicle? How does it differ? Is it better than warning the driver with just vehicle sensor information?

To answer these questions, in section 5.1 we explored visualizations that convey information from smart infrastructure or connected vehicles to the driver. There is no one-fits-it all solution for visualizing context information of traffic scenarios. It is possible that symbols for certain traffic situations need to be learned, as it is the case with many traffic signs. In section 5.2 the driver was informed of the current traffic situation and the possible reason to take over the vehicle in a stereoscopic 3D interface within the vehicle. The performance, workload and user experience was measured. The participants drove slower with a visualization of the upcoming traffic scene, and the lateral lane position was closer to the lane without an object. The overall workload of the system with visualization was lower and the participants had a positive attitude towards the visualization. In section 5.3 we collected possible scenarios in a scenario catalog. With this scenario catalog we categorized the scenarios according to the urgency of a warning. Then, we focused on the design of warnings of objects that are outside the periphery of the driver. We compared the warnings by measuring situation awareness, workload, system usability and braking performance. With that we can show that a symbol visualization resulted in the highest usability and the lowest workload, whereas using wedge resulted in the best situation awareness. Through the study we show that well known symbols outperform not established symbols. For a visualization to result in a high situation awareness as well as a high usability, both aspects need to be combined. The discussed attributes gave insight on the user's preferences of visualizations of situations that are outside the view of the driver and with the collected knowledge in the sections we can answer the fundamental questions from the beginning.

Warnings with smart infrastructure information are different to those that are based on the own vehicle's sensor information. Warning drivers with smart infrastructure information differs in some cases from warning them with information derived from the immediate surrounding of the vehicle. With the scenario catalog we explored the potential of the situations in which smart infrastructure warnings can be presented to drivers depending on the distance to the critical situation. From this, we can conclude that not in all critical situations the warning would differ from warnings with just vehicle sensors. In case a short reaction time is needed, the warning should warn the driver efficiently. In case the critical situation is further away, the warning can be presented differently. Drivers can get more context information, for example, information how certain the smart infrastructure is that this situation will not resolve itself by the time the vehicle arrives. Especially warnings with the color red should be avoided since participants are trained to react immediately on that color. With the smart infrastructure, the traffic situation on the highway can be rated. The number of vehicles and their velocities can be tracked. With much traffic, the situation can be estimated as more stressful for the driver therefore the warning and information can be adapted accordingly. Depending on the stress level of the driver the warning can be personalized with influencing factors such as the maneuver of the own vehicle or neighboring vehicles, weather and other environment factors.

Warning with smart infrastructure information differs regarding reaction time, urgency and content. Drivers that are warned with smart infrastructure information do not need to react immediately, but can first increase their situation awareness with the information provided to them. The warnings differ regarding warnings that need immediate action and can alert the driver in a more non-intrusive manner. With the preparation of the upcoming critical traffic situation, more context of the situation can be provided to the driver.

Warning drivers with smart infrastructure information is better than warning the driver just with vehicle information. The conducted studies that we performed showed that the take over performance can be improved by early visualizations of upcoming critical situations. With the adjustable degree of context information, drivers can choose for themselves, how much context information they need and want. For semi-autonomous vehicles in which it is still necessary to take over the control of the vehicle, the context information can also be provided to the driver, but the distraction from it can not lessen the driving performance. In this situation, the context information needs to be presented intuitively and easy to grasp.

6

Identifying the Needs of Drivers in Unexpected Driving Situations

In the previous chapters we explored the use cases and user groups for smart infrastructure and introduced a design space for 3D augmented reality. Furthermore, we presented concepts to increase the awareness of drivers of situations outside the periphery of the driver. From these sections, we identified how we can convey information about objects, or other traffic participants in the surroundings, to drivers of manual driven and autonomous driven vehicles. While those concepts concentrated on the design of applications and prepared the driver for upcoming take over situations, one question that remains is how drivers are affected if the driving behavior of the vehicle remains unclear. Unexpected driving behavior results in perceived risk [61]. This results from a study of Dikmen et al. [61], in which participants experience automation failure in a Tesla which resulted in unexpected driving behavior. Through the infrastructure or connected sensors, an autonomous vehicle might get information about the environment and react on that information. This vehicle might still be within the autonomous driving limits of the system and therefore does not issue a take over request. In that situation, without more information, the driver is not be able to comprehend which information the vehicle received and the resulting driving decision of the vehicle. For example, if the own vehicle receives information that an emergency vehicle will pass, it reacts on that information by braking. Without the information about the emergency vehicle, the driving behavior is unexpected to the driver. In that case, the vehicle drives unexpectedly without the driver knowing the situation context for this behavior. This can cause various problems, for example, drivers would not know if their intervention is necessary. This can result in drivers taking over control of the vehicle without the need for it and drivers that are not prepared to drive, resulting in critical situations. Additionally, the driver's trust could decrease resulting in lower acceptance of the vehicle and thus in a reduced usage of autonomous driving functions.

Explanations for autonomous system behavior can make the reasons for an algorithm to choose a certain solution clearer. Since autonomous vehicles will probably not immediately work perfectly and without unexpected driving decisions, it is important to keep traffic participants in the loop and informed about the behavior of autonomous vehicles and the context of a driving situation. However, in case of unexpected driving situations that occur often or in which the causes are obvious to the driver, explanations might annoy the driver. Therefore, it is necessary to find those driving situations that are actually unexpected to the driver and to then provide drivers with an explanation that is understandable and upon which the necessary action from the driver is clear.

Drivers in autonomous vehicles that do not have the possibility to intervene in the driving behavior have been taken of their autonomy to decide for themselves. When experiencing this situation, we explored what explanations could be presented in an by a user interface.

In this chapter, two driving simulator studies are presented. Both studies focus on situations in which the vehicle is driving unexpectedly and in which drivers do not know the cause for the unusual and different driving behavior. While one study explores the needs of drivers in unexpected driving situations, the other derives the content of an explanation. Based on the studies, we derive design

Identifying the Needs of Drivers in Unexpected Driving Situations

guidelines that are beneficial for explanation designs in autonomous vehicles. Afterward, we define features of what vehicles detect that are necessary to be mentioned in explanation interfaces for unexpected driving behavior.

This chapter introduces two driving simulator studies to answer research questions RQ4 and RQ5:



Influence of Unexpected Driving Behavior on the Driver of an Autonomous Vehicle

In Section 6.1, we explore the needs of drivers in unexpected driving situations. We identify situations in which those unexpected driving behaviors could occur and examine what drivers need in those situations. Therefore we searched for unexpected driving recounts and did a study (N=26) in which we collected qualitative results. We derive a codebook and design implications for autonomous vehicles.



Developing Adequate Explanations in Unexpected Driving Situations

In Section 6.2 we explore a possible explanation interface for drivers in unexpected driving situations. We first identified interface elements in a focus group. In a study (N=16), we let participants design their preferred interface, providing the reason for the unexpected driving behavior. From this we can then derive a target mental model that should be addressed when designing for explainable interfaces.

6.1 Influence of Unexpected Driving Behavior on the Driver of an Autonomous Vehicle

This section is based on the following publication (conditionally accepted for Mobile-HCI'20):

• Gesa Wiegand, Malin Eiband, Maximilian Haubelt, Heinrich Hussmann (2020, October). "I'd like an Explanation for That!" Exploring Reactions to Unexpected Autonomous Driving. *MobileHCI '20: 22st International Conference on Human-Computer Interaction with Mobile Devices and Services.*

This section is supported by the Master's thesis of Maximilian Haubelt. Please refer to the beginning of this thesis for a detailed statement of collaboration.

Fueled by considerable research and commercial interest, autonomous driving has seen major technological advances in the last years. Under certain conditions, it has become reality: Early rider programs have been launched by companies such as Google¹, Uber² or Aptiv³. Beyond the technical challenges, autonomous driving raises major questions for HCI [122]. Drivers, or when driving truly autonomously rather passengers, of highly automated vehicles can experience unanticipated and unpredictable vehicle behavior. Furthermore, decisions can be difficult to comprehend and can be different from those, a human driver would make. This can lead to anxiety as well as lack of trust in, and acceptance of, highly automated vehicles [118, 250]. It has been shown that providing *explanations* to drivers can mitigate these negative effects [148, 149]. Providing explanations can adequately support drivers in such situations. By addressing the question *when* and *what kind* of explanation drivers need for vehicle behavior can give them back the feeling of power over the driving decisions.

In this chapter we refer to the research question what drivers of autonomous vehicles need and how they want to interact the vehicle in that moment (RQ4). Furthermore, we determine what the situations are in which drivers need explanations of autonomous driving behavior. Therefore, we first conducted a systematic web search of blogs, video material and experience reports to analyze and classify situations in which the need for an explanation arises. Those unexpected driving situations describe real-life experiences with autonomous driving. To derive the situations, we clustered the experience reports into 11 themes to get a common understanding of the context of the collected situations in which drivers need explanations. Through filtering the situations to our problem space, we derived 17 scenarios. We implemented these experiences in a simulation and conducted a study, simulating a fully autonomous ride through the situations. Participants shared their thoughts through think-aloud [134] and were asked to indicate their need for an explanation by pressing a button. Thereby, we collected 2499 statements from participants and their indication if an explanation is needed. Through qualitative analysis, we derived 37 codes and six categories. The codes build up the codebook from which we derive nine design guidelines. The steps are shown in Figure 6.1. We found that some interaction options should be given to the driver of an autonomous vehicle. This is in line of the work of Walch et al. [282] who presented participants with an interface that enables cooperation with the vehicle. Furthermore, we found that autonomous vehicles should give users the chance to ask for explanations when unexpected driving behavior occurs. Participants request an explanation especially in situations with a long waiting time. When no obvious reason is visible there is a higher demand for an explanation.

The contribution of this chapter is a thorough and methodical approach to derive experience reports of using autonomous vehicles, a classification of stereotypical reactions of users when confronted with unexpected vehicle behavior and design implications for a user centered explanation interface.

¹ https://waymo.com/journey/, last accessed May 2020

² https://www.uber.com/, last accessed May 2020

³ https://www.aptiv.com/, last accessed May 2020

Identifying the Needs of Drivers in Unexpected Driving Situations



Figure 6.1: Work flow to derive unexpected driving scenarios and conclude design implications.

6.1.1 Extracting Scenarios

Similar to take over requests, in which the vehicle abruptly hands over control to the driver [132, 253], unexpected driving behavior can disrupt the ride. To explore the reactions of drivers of autonomous vehicles with unusual driving behavior, scenarios are required that identify as unexpected and strange. Unexpected driving scenarios can be implemented in a driving simulator to observe the behavior of potential drivers and collect feedback from participants. In papers from AutoUI (Automotive Use Interfaces) and CHI (Conference on Human Factors in Computing Systems), two premier conferences for HCI and automotive interfaces, studies were conducted that research take over requests and autonomous driving. We reviewed the chosen scenarios in papers from the years 2018 and 2019. Since autonomous driving is a fast changing technology we concentrated on the most recent years. We looked at studies that included take over requests or unexpected driving situations, to get an understanding of the used scenarios in autonomous driving studies. From seven selected papers just one indicated a real life event for their chosen simulated driving scenario [156]. In other cases [160, 195, 283], the chosen scenario was based on assumptions of the autonomous vehicle's limitations.

To collect scenarios, based on real experiences, we conducted a systematic web search. We searched for experience reports from videos, blogs or news articles. The derived scenarios were clustered using thematic analysis [33] and then relevant scenarios were chosen. The procedure to derive scenarios can be summarized as follows: First we conducted a web search, then extracted scenarios, clustered them through thematic analysis and finally filtered the relevant scenarios.

Deriving Scenarios

To find realistic scenarios that are based on real life experiences, we collected experience reports of users of autonomous vehicles. We conducted a web search to identify real life experiences of unexpected autonomous driving, preferably from users that use autonomous vehicles regularly and in everyday situations. We searched for experience reports using the following keywords on Google and YouTube: "Driving in an autonomous vehicle", "Waymo annoying" ⁴, "autonomous vehicle an-

⁴ https://waymo.com/, last accessed May 2020
noying", "Tesla error"⁵, "autonomous driving strange" and "self driving vehicle anecdote". Only results that reflect a real life experience with autonomous vehicles were considered. Since currently the vehicles are not always capable of driving fully autonomously, the collected experiences were only considered for this work if the vehicle was driving autonomously at the moment of the scenario. From the web search, we identified 25 potential situations, with unclear driving behavior, and performed a thematic analysis [33] with those scenarios to identify relevant scenarios. We clustered the situations first according to the vehicle's motion state: driving (N=23), approaching vehicle (N=3)and *parking*(N=2). We excluded the cluster *driving* as it is beyond the scope of this work. We continued by clustering the remaining situations into the following cluster: road type [highway (N=1), city (N=11), undefined (N=11) and influence factors [speed (N=5), robotic behavior (N=12), amount of other vehicles (N=13), personal preferences (N=4), error (N=6)]. We investigated if we can find a balanced amount of scenarios from different road types and influence factors, but there was a majority of city and undefined scenarios. From each of the influence factors the scenario groups we included scenarios in the final scenario group for the study. We excluded all situations in which the vehicle clearly drove against the traffic rules. Situations that occurred while not driving (e.g. "...we approached it [to get into the vehicle], [the vehicle] moved away from us repeatedly, it would not completely stop."⁶, describing a situation in which the driver wanted to board a vehicle, but it moved away) were also excluded. Scenarios that were vague without a concrete description of the situation were also excluded ("Volvo admits its self-driving vehicles are confused by kangaroos."⁷). Our scope just included the driver's perspective on these scenarios, but we implemented also scenarios that were observed by other road users than the driver. The process resulted in 17 scenarios (see Table 6.1).

Resulting Scenarios

We implemented the scenarios in carmaker ⁸, a software for driving simulators. From the web sources and videos we only have short descriptions of the situation or video clips without the complete situation and from different filming angles, which results in room for interpretation, how the scenarios actually happened. Therefore, we repeatedly discussed and interpreted the situations and decided how the situations should be implemented. Some situations could be interpreted in a different manner, hence we decided to implement one described situation in two scenarios. Scenarios 1 and 4 and scenarios 2 and 8 are based on the same source, but are implemented differently. The stopping times and distant of the vehicle to the pedestrian or intersection and the driving behavior varies between the scenarios. We tested all scenarios with two participants (male, 28 and 32 years old) in a pilot study. Based on their feedback, small changes such as vehicle speed or timing with other traffic participants were adjusted.

Limitations

The situations are based on the report of individuals, describing the situation on the internet. Since experience reports are subjective and not meant for interpretation for a driving simulator, the scenario

⁵ https://www.tesla.com/, last accessed May 2020

⁶ https://pbs.twimg.com/media/D2RVjLUUwAIvOFi.png, last accessed May 2020

⁷ https://www.theguardian.com/technology/2017/jul/01/volvo-admits-its-self-driving-cars-are-confused-by-kangaroos, last accessed May 2020

⁸ https://ipg-automotive.com/products-services/simulation-software/carmaker/, last accessed May 2020

Identifying the Needs of Drivers in Unexpected Driving Situations

Scenario	Description	Experience Report	Scenario	Description	Experience Report
1: Abrupt stop at right turn	City: Vehicle approaches an intersection with give way sign. Without stopping at the sign it starts turning right. Then it stops, already on the road. An oncoming vehicle passes. Vehicle continues driv- ing after 5 seconds.	" she nearly hit a Waymo Chrysler Pacifica minivan because it stopped abruptly while making a right turn at the intersection." [83]	10: Unneces- sary lane change	(Two-lane) Highway: Vehicle is driving on right lane of two-lane highway. Two vehicles are in front. Vehicle changes to left lane.	"[] made an "unnecessary" lane change and, lastly, experienced the ap- parent near-collision." "When turning left into a 2 lane turning lane, the car whipped fast across to the in- ner lane which was unnecessary and scary." [225]
2: Reluc- tant turn right due to pedes- trian	City: Vehicle stops before a right turn, a pedestrian stands on other side of street, moves a little. There is no crosswalk. Vehicle slowly turns with some stop- ping.	" When the road cleared and it was safe to turn right, the car didn't budge. I thought this was a bug at first, but [] there was a pedestrian standing very close to the curb, giving the avk- ward body language that he was plan- ning on jaywalking." [129]	11: Un- expected stop	City: Vehicle brakes after a while. In some distance is a green traffic light. Af- ter a while, oncoming vehicle passes. Vehicle still waits. Scenario is similar to Scenarios 6 and 7.	"She nearly hit a Waymo Chrysler mini- van because it stopped abruptly" [83]
3: Long wait at intersec- tion to turn left	(Two-way) Highway: Intersection is ap- proached, traffic light is green. Vehicle stops behind traffic light, then turns left, then waits for 10 seconds to turn a bit, then it stops again. After 60 seconds an oncoming vehicle passes. The vehicle fi- nally turns left.	" Seconds tick by, and the vehicle () is letting too many opportunities pass by without turning " [115]	12: Strong brakes and quick decisions	City: Vehicle enters the city and strongly brakes to reach speed limit. A curve is taken relatively fast, and a right turn at intersection is taken without stopping. The behavior of the vehicle should re- produce robotic driving.	"Google can adjust the level of ag- gression in the software, and the self-driving prototypes currently tooling around Mountain View are throttled to act like nervous student drivers. In the early versions [], the vehicles were programmed to be highly aggressive." [129]
4: Abrupt stop at right turn	City: Identical scenario as Scenario 1 except for waiting time: 20 seconds in- stead of 5 seconds.	See Scenario I.	13: Fast accel- eration and slow downs	(Two-lane) Highway: Two vehicles in front of the own vehicle. After a while it accelerates until distance between ve- hicle and the front vehicle is needlessly short. When distance is dangerously short, vehicle brakes and vehicle falls off. This is repeated three times.	" It kept accelerating unnecessarily fast to reach speed limit from stops and slow downs" [225]
5: Stop- ping vehicle at intersec- tion	(Two-way) Highway: Similar scenario to Scenario 3, but own vehicle immedi- ately starts left turning process and re- mains in intersection. After more than a minute the vehicle finally turns.	Stopping vehicle on intersection. [12]	14: Ve- hicle is very slow	(Two-way) Highway: Vehicle drives and after a while drives relatively slow. While driving slowly, it passes parking vehicles. After a while it accelerates again.	"Car does not really drive fast." [133]
6: Vehi- cle stops at an invisi- ble red traffic light	City: Vehicle stops after some driving at intersection. Intersection has traffic lights, but they are invisible to driver. After a while oncoming vehicle passes. Vehicle still waits and then continues driving. Scenario is similar to Scenarios 7 and 11.	"Sometimes, the vans don't understand basic road features, such as metered red and green lights that regulate the pace of cars merging onto freeways." [83]	15: Ve- hicle got way too close to another vehicle	(Two-lane) Highway: Vehicle slowly gets closer and closer to vehicle driving in front. When distance is too small to be able to brake, vehicle slows down a bit to increase distance to other vehicle.	"Car tried to turn left into oncoming traffic, driver had to avoid collision," The [] robotaxi got "way too close" to another car [] [69]
7: Vehi- cle stops at green traffic light	City: Scenario is similar to Scenarios 6 and 11. Different to Scenario 6 green traffic light is visible.	"the car was too hesitant to make a left turn at an intersection where it had a green light, but not a separate light for left turns." [242]	16: Vehi- cle moves very slow	(Two-way) Highway: Vehicle drives and slows down. After a while it accelerates again.	[] moved "super slow" at one point [][69]
8: Un- clear vehicle pedes- trian interac- tion	City: Vehicle turns right, there is a cross- walk with a pedestrian in front of it. Pedestrian moves a bit, but does not cross the street. Vehicle stops in front of the crosswalk and waits. After a while it slowly moves forward. Then it stops again. It repeats this behavior. After a while it crosses the crosswalk and con- tinues driving.	See Scenario 2.	17: Slow truck is not over- taken	(Two-way) Highway: In front of the own vehicle is a vehicle and a truck. The other vehicle overtakes truck, own vehicle stays behind the truck. Differ- ent road types are passed: curvy and straight, view on the road ahead is clear.	"a Waymo vehicle that got stuck behind a city bus that pulled over to load and unload passengers while still partially blocking the lane." [242]
9: Vehi- cle stops, child crosses	City: While driving, vehicle stops abruptly. It waits and after a while a child crosses street. Vehicle continues driving.	"Car stopped all of the sudden. Why? []Child run out between some cars." [289]			

Table 6.1: Description of the resulting scenarios, derived from the experience reports. On the left is the related scenario with a short description, in the middle column is the scenario description derived from the discussion while clustering the scenarios and in the right column is the relevant extract from the original scenario description.

Influence of Unexpected Driving Behavior on the Driver of an Autonomous Vehicle

descriptions lack details or information about the environment. Other traffic participants could also have been involved, the weather could have been different to how we imagine it, and we do not know on what road type the scenario was set. Through thorough discussion and testing, we got a common ground on how the scenarios could be interpreted. We had three main settings in the simulation. Therefore, the participants had a bias towards some scenarios since the landscape looked familiar, which sometimes resulted in expectations in a certain driving behavior.

6.1.2 Method



Figure 6.2: Screenshots from selected scenarios. Top row (from left to right): a) Scenario 1: vehicle just turned right and stopped while an oncoming vehicle passes b) Scenario 2: vehicle is about to turn right, a pedestrian is on the other side of the road c) Scenario 9: a child crosses the street in front of the vehicle. Bottom row: d) Scenario 11: vehicle stops in the city unexpectedly before green traffic lights e) Scenario 8: vehicle waits in front of a crosswalk, a pedestrian is on the sidewalk f) Scenario 3: vehicle is standing on the intersection in the middle of a left turn, an oncoming vehicle passes.

After defining and implementing the scenarios (see Figure 6.2 for example screenshots), we conducted a study to determine the key topics and thoughts of the participants in those unexpected driving situations. We used a within-subject design with 17 scenarios. The independent variable was the condition which were the simulated situations. The conditions are counterbalanced using the Mersenne Twister algorithm [192]. The dependent variables were think-aloud statements [134], explanation requests and the time until this requests. The study lasted one hour per participant. Each scenario had one unexpected event at prescribed times that was unique for each condition.

Participants We recruited 26 participants (m=15, f=11), 15 were between 20-29 years old, 3 were 30-39 years old and 3 were older than 60 years. All participants had a driving license and had experience in driving regularly. The majority agreed or totally agreed that they enjoy not to pay attention when being the co-driver (N = 17, Mean = 1.88, IQR = 2), stated that they were rather patient while driving (N = 17, Mean = 2.3, IQR = 1) and preferred to drive rather defensively (N = 20, Mean = 2.11, IQR = 0) on a scale from 1 to 4. They stated they would like to request further information in case of disturbances during a train ride (N = 22 agree or strongly agree, Mean = 1.69, IQR =1). One can see some parallels in riding a train compared to driving in a self-driving vehicle,

therefore we expect that participants that require further information in a train, would also request information in unexpected driving situations in autonomous vehicles.

Apparatus The study was conducted in a driving simulator. The scenarios were shown on three screens in front of the vehicle. A keyboard tracked the input of the participant in case of an explanation request. The level of automation was a SAE Level 5 [130], the driving was conducted by the simulation and the participant was in the driving seat, but did not need to intervene.

Procedure The procedure and aim of the study was explained to the participants and they then signed a consent form. After filling in a demographic questionnaire, the participants took part in the study. All participants experienced each of the 17 scenarios. During the study the participants were asked to think-aloud and report anything they notice, what the consequence might be and why the vehicle drives the way it does. If they wished to have an explanation from the vehicle they could indicate so by pressing a button. After each scenario they completed a questionnaire. It surveys, if and how the participants wanted to influence the driving maneuver by asking them to write a short statement. Furthermore, it asked if they were glad to be able to request an explanation on binary scale. On a 4 point Likert scale they rated how safe they felt and how frustrated they were with the driving behavior of the vehicle. After the study the participants were asked if there were any questions.

The qualitative data was first transcribed and then analyzed, using qualitative analysis [3, 85]. The data was coded by an open coding scheme [45]. Following this approach, we first assigned codes to distinct participant's statements separately and then collected the codes of two participants. Those codes were then compared and the two participants were re-coded on a common first draft of the codebook.

The next two participants were coded separately again before counter-checking and adapting the results. Rules concerning certain repetitive statements (e.g. "That is annoying.") were assigned to follow in the further coding process. Some codes were defined as redundant by us in the process and new codes were defined. Through this, a common understanding of the codes and their representation in the data was found. The rest of the participants' statements were then divided among us and coded separately. In the end we compared the codes of a part of the participants. The inter-rater reliability was evaluated by calculating the Cohen's Kappa coefficient [110]. This resulted in a value of 0.86, indicating a very good agreement between the raters. Using this iterative approach, we developed and refined the codes until everyone agreed with the assigned codes. By identifying relationships between the open codes, the axial codes or also code categories, were defined.

6.1.3 Results

The results of the study are presented in two parts: the qualitative part in which the statements of the think-aloud method are analyzed and the quantitative part in which the collected explanation request events are presented.

Influence of Unexpected Driving Behavior on the Driver of an Autonomous Vehicle

Code	Definition	Count	Example				
	Emotion and Evaluation - Negative						
E1	Dissatisfaction with driving behavior	265	"This is totally unnecessary."				
E2	Desire for explanation	138	"I would like an explanation for that."				
E3	Irritation / Confusion	131	"[] for no real reason."				
E4	Annoyance	122	"That's kind of frustrating already."				
E5	Feeling of safety	116	"I think it is totally dangerous."				
E6	Impatience	61	"We need to move on now."				
E7	Uncertainty	34	"We're not going any faster. Or do we? "				
E8	Helplessness	19	"I feel abandoned."				
E9	Comfort	13	"[] that's not very comfortable for me."				
E10	Envy of others	13	"Well, now we are pretty slow and the others are faster. Isn't this nice [ironic]?"				
E11	Feeling of being left out	4 Emoti	" [The car knew that but] that wasn't obvious to me, though."				
E10	Catiofaction with driving habeviag	Emoti	on and Evaluation - Positive				
E1Z	Satisfaction with driving benavior	82	"I would call that reasonable driving now."				
E13	Preving of safety	37	[] and that makes me reel sale.				
	Praise	29	So far the situation is perfectly solved by the car.				
E10 E16	Delief	22	"Hete off Dhow I wouldn't have national "				
E10		15	Hats off! Pnew, I wouldn't have noticed.				
	Trust	Z	terpretation and Peacon				
	Looking for reason in the environment (other		"It might have alound down headyna of the kid. It could also be that we simply reached				
R1	troffic participants	252	the destination "				
D2		494	ute desultation.				
κz	Cidelessness	131	it's absolutely impossible to understand why we are standing here.				
R3	Vehicle status	100	"There is no car visible far and wide. It feels to me like there's been a mistake."				
R4	Interpretation of vehicle	44	"It may have noticed that other cars were trying to overtake us and therefore slowed				
			Canability of the Car				
C1	Comparison to own driving behavior	180	"I would do that completely differently."				
62	Mental model autonomous system / driving	163	"[] but I do not think [the car] saw that, at least I did not."				
C3	Humanization / Assigning personality to car	41	"Please keep driving, dear car, and do not stay there. Hello?"				
C4	Competence of the car	40	"[] but I can't imagine it's predicted that before."				
C5	Ignorance of car driving behavior	25	"And I think that you have such a tension whether the car still registers everything."				
C6	Driving style	19	"I would definitely see the situation [] as aggressive driving and it is not quite clear to				
			me why we are driving so aggressively."				
C7	Overestimating ability of car	7	"Okay, I guess the car knew the other car would turn right."				
			Interaction				
11	Desire to interact / Change driving behavior	154	"I would really like to step in and get on the gas right now."				
12	Awareness of being an impediment for others	29	"Well, we're pretty much obstructing other traffic, too."				
13	Desire to communicate with car	26	"Okay [] car Please explain to me why you're not overtaking []?				
13	Social interaction with pedestrians	12	"I would book on [the pedestrian] go or don't go!"				
14	Social interaction with pedestinans	12	"So it would be cool if the car bad a short message: we're still waiting 2 minutes or balf				
15	Temporal demand	12	a minute."				
16	Safety of others	4	"And maybe the car thinks he's running into me. And so, he rather stops. [] the car was unsure whether the pedestrian was crossing the road improperty."				
	Prediction						
P1	Confusion of future driving maneuvers	60	"We will overtake this time. I suppose. But don't drive faster. Or do we?"				
22	Expectation of future driving maneuver /	FO	"No appointing traffic. Then I would gav now we drive left before creative corrections are correct."				
r2	behavior	50	I solo oncoming trainc. Then I would say how, we drive left, before another car comes.				
P3	experience	47	at some point."				

Table 6.2: The codebook of the qualitative study. The codebook is set up of the six categories and the corresponding codes. Related to every code is an abbreviation is on the left, the definition of the code, the number of counts and an example of the code.

Analysis According to the Codebook Results

We assigned codes for 2.499 statements. 37 individual codes and 5 axial codes, or *code categories*, were found through open coding. Table 6.2 shows the codebook with example participant quotes. The code categories are *Emotion and Evaluation*, *Interpretation and Reason*, *Capability of the vehicle*, *Interaction* and *Future Driving Prediction*.

Emotion and Evaluation

The code category *Emotion and Evaluation* is divided according to positive and negative emotions. More than a third of codes belong to this category, showing that emotions had a strong influence on the thoughts of the drivers in the study. Negative emotions clearly prevail positive emotions (83% of all emotion codes).

Negative Emotions The negative emotions include *Impatience, Dissatisfaction with driving behavior, Helplessness, Irritation/Confusion, Envy of others, Feeling of being left out, Uncertainty, Comfort, Desire for explanation, Annoyance* and *Feeling of safety.* The code *Comfort* refers to the driving style of the autonomous vehicle in a scenario and not the driving simulator itself. Most often (in 35% of all negative emotion statements) participants mentioned that they were dissatisfied with the driving behavior during the scenarios. This validates the implemented scenarios and indicates that driving behavior in these scenarios were indeed unexpected. Additionally, participants desired an explanation, had a feeling of irritation or confusion about the situation or were annoyed by the driving behavior. Many participants mentioned that they thought that the driving maneuver was dangerous, represented by the high number of codes in "Feeling of safety" (15%).

Positive Emotions The positive emotion category consists of the codes *Praise, Relief, Satisfaction with driving behavior, Feeling of safety, Understanding* and *Trust.* If the uttered emotion was coded positive it was rather general satisfaction with the driving behavior. Sometimes the participants mentioned that even though they understood why the vehicle behaved in a certain way, they would have driven differently. Moreover, in scenarios in which they thought to have understood the behavior of the vehicle, participants also praised the driving decisions of the vehicle. That happened mainly in two scenarios, in Scenario 8, in which the vehicle waited for a pedestrian in front of a street crossing and in Scenario 9, in which the vehicle waited for a child that ran across the street. In Scenario 12, in which the vehicle is aggressively braking and accelerating, some were startled by a fast right turn, but the others (N=15) were satisfied with the driving behavior.

Feeling of safety was both related to negative and positive emotion and either meant that the participants thought the driving behavior was dangerous or that they felt safe.

Interpretation and Reason

The category *Interpretation and Reason* is the second largest category. Since participants were prompted to explain the vehicle's driving behavior, they tried to interpret the driving behavior of the vehicle. With a representation of 48%, the most influential code was *Looking for reason in the environment or other traffic participants* (e.g. "Hello? Ah it's probably about, I don't know, pedestrians? Cyclists?"). They suspected that the other traffic participants influenced the driving behavior and guessed that the vehicle got some information about other vehicles and behaved accordingly.

Moreover, they expected that the status of the vehicle (Code: *Vehicle Status*) could have some influence on the driving behavior, either that the sensors did not work properly or that there was an error in the software of the vehicle.

They also thought that the vehicle might have interpreted the situation wrongly (Code: *Interpretation of vehicle*) and therefore changed the driving behavior.

Depending on their mental model of the abilities of the vehicle, participants overestimated the abilities of the vehicle and wondered, for example, if the vehicle could have detected a vehicle that was not yet visible. Some provided an explanation by assigning anthropomorphic characteristics to the vehicle (e.g. "Maybe he (the own vehicle) is afraid of the other vehicles.", "Because he could feel the vehicle from a distance."). These statements are very different to a statement of a different participant ("Software is not advanced enough to perform an overtaking maneuver"), which describes the vehicle behavior on a more technical and logical level. These two examples show the spectrum of the mental models of the participants. Others considered technical aspects that could have influenced the situation such as erroneous software or hardware ("The sensor isn't working that's my assessment").

Moreover, participants were uncertain of the abilities of the vehicle ("The situation is not very transparent, maybe that is also difficult for the vehicle."). Interestingly, participants' own explanations did not change during the study. Participants who explained the behavior with sensor errors in the beginning tended to stick to that explanation.

The code "mental model" (see Code C2) and the reasoning of the participants are similar, and sometimes overlap. We coded the mental model when a participant expressed statements about the capabilities of the vehicle and reasoning when there is a conclusion referring to the current vehicle behavior.



Figure 6.3: Time in the scenario and the corresponding average explanation request times in black indication for each scenario.

Capability of the Vehicle

The most prominent code in the category "Capability of the vehicle" is *Comparison to own driving behavior* followed by the *Mental model of autonomous system*. Participants notably often compared

Identifying the Needs of Drivers in Unexpected Driving Situations

the driving behavior of the vehicle to their own driving decisions which often would have be different. One participant, for example, remarked in Scenario 13, in which the vehicle follows a vehicle and accelerates and brakes several times that "I wouldn't always change the spacing so abruptly, I'd do that differently if I was driving myself.".

Especially in Scenario 8 participants compared their driving behavior to the driving behavior of the autonomous vehicle. In this scenario the vehicle slowly approaches a crosswalk. Since there is a pedestrian in front of the crosswalk, participants mentioned that they would also slowly approach the crosswalk, but eventually drive on. This indicates that participants would have liked to adapt the driving style of the autonomous vehicle to their own driving style – however, only in scenarios in which participants rated their own competence higher than that of the vehicle. In Scenario 9, the vehicle stops until a child crosses the street. In this scenario some participants mentioned that they themselves would not have noticed the child. Even though some doubted that future vehicles would be able to handle this situation well, reactions were mostly positive.

It is insightful to collect statements about the mental model of the participants. Since the mental model is just a representation of the real world, all aspects, such as the capability of the vehicle or how it works, are covered in several categories. The code in this category collects all those aspects from different categories.

Interaction

Even though participants were informed that no interaction with the vehicle was possible, they often wanted to have options to interact. A significant amount of codes (65%) in this section indicates the *Desire to interact or change the driving behavior* of the vehicle. This can be distinguished between the wish to take over control of the vehicle and the mere wish to change the driving behavior of the vehicle.

While less frequently mentioned, other interesting aspects were the fear of being an impediment for others others when other vehicles need to wait (Code: *Awareness of being an impediment for others*, 12%) and communication with the vehicle (Code: *Desire to communicate with vehicle*, 11%). This also includes a certain degree of feeling embarrassed for the driving behavior of the vehicle and a feeling of discomfort.

In contrast, the wish to interact with pedestrians was not very prevalent (Code: *Social interaction with pedestrians*, 5%). Within our study, participants wanted to get information about the driving decisions of other autonomous vehicles. Especially the codes "Safety of others" and "Awareness of being an impediment for others", are a sign that the participants were aware that their vehicle could be a traffic hazard.

Future Driving Prediction

Participants either were confused about future driving behavior or they expected a different driving maneuver from their own driving experience. The codes in this category are *Confusion of future driving maneuvers*, *Prediction of driving behavior out of own experience* and *Expectation of future driving maneuver / behavior* (e.g. "He might want to overtake him now.").



Influence of Unexpected Driving Behavior on the Driver of an Autonomous Vehicle

Figure 6.4: Average explanation requests for each scenario in percent.

Explanation Request Times

Participants were not shown an explanation of the scenario by default, to not influence the driver with the content of a particular explanation. Instead, the participants were informed that they could indicate that they would need an explanation in the course of a scenario by pressing a button. We thus assessed *whether* a participant requested an explanation and *when* the request was made. The moment to request an explanation can be either *before*, *during* or *after* the scenario. We defined the start of a scenario when the vehicle depending on the scenario, either has stopped (Scenarios 1-9 and 11), starts to de- or accelerate (Scenarios 13 - 16), has the same velocity as the truck (Scenario 17) or starts the lane change maneuver (Scenario 10). Scenario 12 is an exception since it does not have a clear starting time. The end of the unexpected driving behavior is defined at the moment when the vehicle drives smoothly again or the scenario is finished.

In most cases participants asked for an explanation *during* the scenario (N = 288 times) (see Figure 6.3 for the time in a scenario and the corresponding average time for an explanation request). This is somewhat still surprising as we expected that for some scenarios the participants will ask for an explanation *afterwards*. But this was just sporadically the case (7 times). In four times, an explanation was requested afters scenario 10. This can be explained by the definition of the scenario, the vehicle was driving normally again therefore, for the evaluation the scenario was over. Nevertheless, the participants probably still waited for something to happen that would explain the lane change. When that did not happen, they requested an explanation. In some scenarios, some participants did not request an explanation after 24.3 seconds. Scenario 17 was the scenario in which participants requested an explanation the latest (t = 44s) and Scenario 1 was the scenario in which participants required an explanation the earliest (t = 6s). The high explanation request numbers indicate that the chosen scenarios are suitable to test unexpected driving behavior of autonomous vehicles. We assume

that if the system itself detects unusual driving behavior it should suggest to provide an explanation to the driver.

6.1.4 Discussion

The results show that drivers are interested in getting explanations. There are just three situations (scenario 1, 10 and 12) in which less than 50% of the participants requested an explanation. Scenarios 1 and 10 last less than 20 seconds which could explain why participants did not ask for an explanation. In scenario 12, the vehicle drove not smoothly which also lasted for a short time. Therefore, it can be concluded that participants are higher likely to ask for an explanation in case of a long lasting unexpected behavior of the vehicle.

Nevertheless, the expectations for the level of explanations vary depending on the mental model of the driver, the state of technology available and the type of situation. The identification of the needed explanation level for a specific user is generally difficult. The explanations ideally should provide experiences which enable users to adapt their current mental models towards a better reflection of the reality. For instance, one participant stated that the vehicle always has problems at the same intersection, even though the situations were unrelated to each other. This is a typical misconception created by the urge to find an explanation of a phenomenon which cannot be explained with the current mental model. For this user it would already be helpful to get further information, for example that the system takes its decision based on information about the traffic situation. Many participants assume that the information used for decisions comes only from the vehicle's sensor system. In particular when future mobility infrastructure will provide much better networking between vehicles and environment, explanations can help to broaden the mental models of drivers adequately. For instance, indicating the source of the information (vehicle-to-vehicle communication, smart infrastructure, sensor system of the autonomous vehicle) on which the current behavior is based on, might be beneficial for the driver. In summary, the diversity of observed mental models is a challenge for explanation design. Good explanation design can trigger a learning process in drivers, which improves the mental model of the driver and minimizes the difference between the mental model and the reality. Simply providing a one-size-fits-all explanation, which most likely will be too complex or too simple for some users, is rather unlikely to achieve this goal. Therefore, information about the context of the driving situation and interaction with the environment or the own vehicle is relevant to drivers in unexpected driving situations. The study uses scenarios that occur nowadays on the road with still supervised autonomous vehicles. Therefore, our findings pertain to the limited automation today, but are not validated for autonomous driving experiences in future autonomous vehicles.

6.1.5 Design Implications

From our results, we derive design recommendations for cockpit elements of SAE Level 5 vehicles that can be grouped into two broader categories, *design for adequate control and communication*, and *design for different mental models*. We acknowledge, however that the suggested implications might not be comprehensive. They should therefore be extended and validated in further studies.

Design for Adequate Control and Communication

Our results motivate a closer look at drivers' interaction options. Across most driving situations, participants expressed a wish for interaction during autonomous driving. These wishes can be summarised into two categories: Either *direct control* of the driving task or *communication* with the vehicle and other road users.

Regarding direct control, participants wanted to adjust speed (i.e. to accelerate or brake), to stop or continue driving, to back up or to navigate differently. Participants did not necessarily want to take over full control of the vehicle in all situations, but to slightly adapt the driving behavior. Taking over full control was even perceived as not desirable in autonomous driving. For example, one participant stated that "[...] here I would have to intervene and continue driving by myself [which] would be stupid". With regard to communication, participants wanted to get more information on the situation in general and to update the vehicle's system information about traffic rules, signs and context interpretation ("[I would like to] tell the vehicle that the pedestrian won't cross the street."). They also expressed their wish for communication with other road users, that is, other vehicles or pedestrians. Taking drivers out of the loop in this regard, evoked feelings of frustration.

To date however, the most prevalent form of interaction with highly automated vehicles are take over requests that primarily evolved from technical or legal considerations, not from an HCI perspective. Such take over requests may lead to a lower acceptance of and loss of trust in autonomous driving [282]. In contrast, the insights gained in this study highlight the need for interaction options beyond this "all or nothing" approach – designing interaction in a way that provides *adequate* means for control and communication in autonomous driving.

More concretely, we suggest the following design implications to design for adequate control and communication:

- Input option to change driving behavior [282]
- Change characters of vehicle information system [32]
- Option to collaborate or interact through an interface with pedestrians or drivers of other vehicles [211]
- Collaboration with the own vehicle/ Helping the vehicle decide in difficult situations [282]
- Option to request an explanation from the vehicle [300]

Design Explanations for Different Mental Models

Mental models represent individually constructed, subjective interpretations of the world based on prior experiences and assumptions [137] and are usually incomplete [215]. This presents a challenge for designing communication between user and system, for example, through explanation. The results of our study confirm this challenge: When prompted to explain autonomous driving behavior, participants do so in very diverse ways.

Adapting explanation interfaces in a way that takes into account the diversity of mental models could support drivers in "speaking their language". This is in line with the work by Miller [199] who,

Identifying the Needs of Drivers in Unexpected Driving Situations

drawing on psychology, argues that the recipient of an explanation is more likely to trust the system if the explanation is simpler. Another example for the different needs of users regarding explanations is the work by Cawsey [44], who presents the user modelling component of the EDGE explanatory dialogue system. It shows that adapting an explanation according to a user's level of understanding and thus updating the user model according to the input and knowledge of the user, the explanation is more effective.

From our results, we suggest that explanations for different mental models should address the following aspects in particular:

- An indication of confidence of the vehicle in the driving decision [117]
- Retrievable information from the environment (for example through smart infrastructure or connected cars) and from the vehicle system (for example detected vehicles by the vehicle) [295, 299]
- Navigation interface that indicates future driving path [264]
- Internal vehicle status about possible errors or service information

Their concrete implementation could be assessed in user-centric research processes as presented in [71].

6.2 Developing Adequate Explanations in Unexpected Driving Situations

This section is based on the following publication:

 Gesa Wiegand, Matthias Schmidmaier, Thomas Weber, Yuanting Liu, and Heinrich Hussmann. 2019. I Drive - You Trust: Explaining Driving Behavior Of Autonomous Cars. In: *Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems* (CHI EA '19).

Please refer to the beginning of this thesis for a detailed statement of collaboration.

In the previous section, we identified design guidelines for explanations provided by autonomous vehicles. We know now what to consider when designing for an interface that puts the drivers needs in unexpected scenarios into focus. In this section, we determine the explanations that should be provided to drivers of autonomous vehicles in unexpected driving situations. This section explores the practical implementation of an explanation interface to answer what needs to be visualized in the vehicle, for the user to understand autonomous driving behavior (SAE Level 4 or 5 [130]).

Situation awareness (SA) describes a driver's awareness of the surrounding environment while driving autonomously [74]. Situation awareness should not just increase trust, which might lead to over-trust in autonomous systems and accidents, but should calibrate trust to an adequate level. With calibrated trust, the user is aware of the limitations of the systems and therefore can adjust the trust either by trusting the system more or less. SA calibrates trust in the automation algorithm, by clarifying the abilities of the algorithm, and can enhance the take over behavior [171].

In case of a driving maneuver of the autonomous vehicle, the driver might require an explanation, to prevent distrust in the vehicle or take back the driving control from the vehicle. Explanations can increase trust [267], but transparency of systems does not automatically achieve that goal [53]. Therefore, explanations have to be designed in direct agreement to its expected purpose. For example, when aiming for system transparency, the user of the system must understand the abilities of a system to be able to intervene competently in the control of the system. To design explanation components for the driving decisions of autonomous vehicles, we first need an understanding of the expert and non-expert mental model of autonomous systems to identify the differences between the user mental models [71]. The targeted explanation is intended to improve the driver's situation awareness in driving decisions of the autonomous vehicle that have no obvious cause.

According to Eiband et al. [71] a target mental model can be identified by adding evaluated key components of the user mental model to key components of the expert mental model. We evaluated the improvement of situation awareness with a visual explanation of the scenario in a study. Additionally we evaluated the user's mental model of the autonomous vehicle to identify a target model. With this, the elements that should be included in an explanation for unexpected driving maneuvers, are listed.

Identifying the Needs of Drivers in Unexpected Driving Situations



Figure 6.5: Configurable explanation screen consisting of nine visual components.

6.2.1 Method

To have a basic understanding of explanation visualization preferences of users, we did a questionnaire (N=20). We compared statistical explanations, represented by a decision tree, text and speech based explanations, an environment model, graphical sensor data and real world image explanations. The preferred visualization was an environmental model, which we chose as visualization output. According to Eiband et al. [71] a deep understanding of what to explain is an advantageous prerequisite to set up a transparent system. To achieve that, an expert mental model of the system is first elicited by conducting a focus group. In the next step a user mental model is evaluated by conducting a study in a driving simulator. Those results enable us to set up the target mental model of the system, which answers the question what information should be visualized to explain autonomous driving behavior.

Expert Mental Model We first gathered insights from autonomous driving experts, how they explain an autonomous driving system. Therefore, we conducted a focus group (N = 6, f = 1, m = 5). The experts defined three categories which describe an autonomous driving system: perception, deliberation and action. *Perception* includes all components concerning sensors, object detection, localization, tracking, maps, data fusion and fusion. *Deliberation* includes route planning, environment model, environment prediction and trajectory planning. Conducting the calculated driving *actions*

requires control of the car's actuation and a Human-Computer-Interface (HCI). Concerning the comprehension of autonomous systems, the experts stated: "To understand the driving decision of the autonomous vehicle an environment model would be sufficient for most end users." and "Trajectory planning is too complex for normal users."

They additionally stated that the vehicle should apologize to the user in case it made a mistake and it should be triggered and adapted according to type and significance of a situation. Also, offering a real-time description of the system's behavior by pictograms or some kind of abstract visualization could be helpful for the developers of the system. However, the experts did not believe that normal users would want to get deep insight into the system's algorithm, as it would be hard to understand.

Target Mental Model To determine the target mental model, key components of the user mental model need to be enhanced by key components of the expert mental model. Within this work we identify the user mental model in the scope of a study (N = 16, f = 4, m = 12). By defining the differences between the expert mental model and non-expert mental model and by priority allocation, the key components for the target mental model are extracted.

6.2.2 Study

We used a within-subjects design so every participant with the only independent variable being the situation through which every participant drove through. Eight participants started with condition *Deer* and the other half started with condition *intersection*. The conditions were two different situations, which are described in Table 6.3.

	Situation A: Deer	Situation B: Intersection
Scenario Description	The autonomous vehicle drives through a forest, after some time there is a deer behind a tree on the right hand side.	The autonomous vehicle enters a town and stops at a green traffic light at an intersection.
Autonomous Vehicle Be- havior	Slowing down.	Stopping at green traffic light.
Behavior Explanation	The deer might run on the street and get hit by the vehicle.	An emergency vehicle is de- tected by cloud information and might cross the intersection even though the traffic light is red.

 Table 6.3: Situation description and the corresponding explanation for the situations.

The dependent variables were visualization elements of the explanation. The explanation consisted of abstract visualizations of different layers (1-9), representing the autonomous system's components (see Figure 6.5). The element "My movement prediction" (1) was an area in front of the vehicle which indicates, where the vehicle might move next. "Object's movement prediction" (2) was an area in front of the object visualizing the estimated next movement of the object. "My travel route" (3) was a line on the road visualizing the planned route. The detected objects were visualized by the element

Identifying the Needs of Drivers in Unexpected Driving Situations

"Recognized objects" (4). "Context information" (5) included the background information of the situation, abstracting information e.g. from the cloud. Abstract "Sensor symbols" (6) showed from which sensor (e.g. the sensors of the own vehicle or vehicle2vehicle information) the information was retrieved. The "Sensor range" (7) information of the vehicle was visualized by a transparent region around the vehicle. "Environment information" (8) included houses or trees in the scenario. "Infrastructure" (9) displayed the road and the traffic lights.

The other dependent variable was the level of situation awareness. The participants reached a certain level of situation awareness according to their perception (L1), comprehension (L2) and prediction (L3) of the situation. The statements of the participants were classified using the following levels:

Level 0

No situation awareness level: We are driving.

Level 1: Perception (Elements) Situation A: Deer in the forest. Vehicle is slowing down Situation B: Green traffic light. Vehicle stopped.

Level 2: Comprehension: (Dependencies and significance of objects recognized) *Situation A:* Vehicle is slowing down because there is a deer/object in the forest. *Situation B:* Vehicle stops even though traffic light is green.

Level 3: Prediction: (Future actions are predicted) *Situation A:* Vehicle could collide with deer. *Situation B:* Vehicle stops because a rescue vehicle could turn around the corner and collide with the own vehicle.

Participants The study was conducted in a driving simulator with 16 participants (12 male, 4 female) with ten participants being between 20-30 years old, three between 31-40, two between 41-50 and one between 51-60 years old. All of them were employees of a research institute and therefore had a high technical affinity.

Apparatus The study was conducted in a driving simulator (see setup in Figure 6.6) with three monitors. The steering wheel had no function during the study since the vehicle was driving autonomously and the participant is not able to intervene in the driving. The driving simulation was controlled by the supervisor while participants were manipulating the explanation screen via a separate interface on a tablet and answered a questionnaire on an additional laptop. The simulation was implemented in Unity 3D (Version: 2018.2.8f1)⁹. The explanation screen (see Figure 6.5) was displayed in the main monitor.

Procedure The participants were first welcomed and the demographic questionnaire was filled in. The participants answered before the study the question: *In your own words: How would you explain an autonomous vehicle?* (*What does it do? How does it work? How does it come to its decision?* Before driving in the simulator, the participants were informed that they are driving in an autonomous vehicle, therefore no intervention of them is needed during the study. As secondary task

⁹ https://unity.com/, last access May 2020



Figure 6.6: Participant selects explanation interface elements during the study.

the participants watched a video and counted people therein to simulate a realistic autonomous driving experience. As main task, participants were confronted with the two different driving situations (see Table 6.3) After each condition they were asked what happened in the environment and what the vehicle did. Those statements were used to assess the situation awareness of the participants without an explanation. Then, subsequent to the vehicle's reaction, a visual explanation of what happened in the environment was presented to the participant. The participants chose their preferred explanation by removing those elements that were not necessary for their situation understanding. Afterwards the participants were asked again what happened in the situation. The statements of the participants were divided into the three level of situation awareness. According to these statements, the situation awareness of the participants answered after the study the question: *After seeing the autonomous vehicle in action, how has your view on how an autonomous vehicle works changed compared to your previous answer?*

6.2.3 Results

We first analyze the situation awareness level [73] of the participants to derive the difference of the SA of the participants before an explanation of the situation was given to them and after an explanation was given. Then we formed a user mental model from the statements of the participants during the study. From this user mental model we can derive a target mental model by comparing it with the expert mental model.

Situation Awareness The difference between the level of situation awareness before (Situation A: Mean = 0.81, SD = 0.83, Situation B: Mean = 0.88, SD = 0.96) and after (Situation A: Mean = 1.81,

Identifying the Needs of Drivers in Unexpected Driving Situations



Figure 6.7: Frequency at which participants chose the explanation elements.

SD = 0.98, Situation B: Mean = 2.06, SD = 1.12) the explanation was evaluated with the Wilcoxon signed rank test. The results (p = 0.0001) showed significant increase in situation awareness (see Figure 6.9).

Preferred Explanation Elements The other independent variable that was evaluated are the preferred explanation elements. The chosen elements are shown in Figure 6.7. The least chosen element is sensor symbols which indicates that the interpretation of sensor symbols might not be intuitive.

User Mental Model From the participants statements after the study, we derived a mental model of the participants towards autonomous driving systems. The mental model of users composes of elements that are similar to the components of the mental model of experts (see Figure 6.8). The statement "An autonomous vehicle is a vehicle, equipped with different types of technologies that work together in a way that allows the vehicle to move and navigate in its environment freely and independently and according to given rules - without human interference." includes the description of sensors, algorithms, fusion, navigation, trajectory planning and actuation in the words of a non-expert. But the description of an autonomous system in the words of an non-expert also differs significantly from expert knowledge. The ability of an autonomous vehicle is overrated by non-experts ("It is able to foresee dangers") or the user has other expectations of the system ("Explains what it is doing"). It

	63%					56%
Sensors		No	hun	nan	С	ontrol
50%	38%		31%			
Deliberation / Algorithms /	Control (Steering, etc.) based on other components		Route Planning			
Decision Making / ML	19%	5	13%			13%
43%	Obje Dete		oject etect	ion	C F	Object Prediction
	Safer	Safer 1		3%	,)	13%
	Driving Image 13% Detect		ion			
			13 /0		с ,	Explanation
Trajectory Planning	Human	e Sensor Fusion				6% ADAS

Figure 6.8: Elements mentioned during User Mental Model evaluation. The numbers indicate the percentage of participants that mentioned each element.

remains in the study supervisors assessment to assign the mental model components to the statements of the participants.

Target Mental Model After the evaluation of the non-expert and expert user mental model a target mental model is identified (see Figure 6.10). The common notion of the functionality of an autonomous system is displayed in the overlapping area. The key components that are rated by the users as not as important and the wrongly identified elements of the users are excluded from the target mental model. Therefore, even though "Data and Sensor Fusion", "Map" and "Environmental Model" was mentioned by non-experts and experts it is excluded from the target mental model. The element "Sensors" is rated low as chosen element (see Figure 6.7). Nevertheless, as it is a high priority component during the User Mental Model evaluation (see Figure 6.8) we added it in the target mental model.

6.2.4 Discussion

In this section we measured the effect of an explanation on the situation awareness level of the participants. Additionally, we identify the context information needed in an explanation. In the third evaluation part, we identify a target mental model by identifying the differences and similarities of



Identifying the Needs of Drivers in Unexpected Driving Situations

Figure 6.9: Situation Awareness of the participants before and after getting provided with an explanation of the driving scenario.

the mental model of experts and users. We measure the quality of explainability using the driver's levels of situation awareness. Situation awareness is important if the driver needs to take over control during a driving scenario. With this study, we showed that situation awareness can be increased significantly by showing an explanation of the cause of unexpected driving behavior. It remains to prove the benefit of increased situation awareness through explanation screens in autonomous driving. In our scope of work, the driver is not required to driver manually, therefore the driving performance does not need to be improved by increasing situation awareness. It is imaginable that the acceptance and joy of driving can be increased by offering an explanation of the driving context.

The participants were able to adapt the explanation to their personal preferences. The resulting explanations were very similar, resulting in some explanation elements that are chosen less often than others. Adaptable interfaces and personalization are important topics that are discussed frequently. In our study though, we can not see of a clear benefit from changing the explanation elements when an improved explanation is chosen. Nevertheless, it is imaginable though, to adapt explanations according to the workload of the driver when driving manually or on the immersion of the driver in a secondary task. Depending on this immersion, the driver is more willing to get more information about the driving context or decides to stay immersed in the secondary task.

The third contribution of this work is the target mental model that we identified by comparing expert and user mental model. While it is a legitimate approach to identify a suitable explanation, the target mental model needs to be verified in a next step. It needs to be proven that some explanation elements should be excluded while others are crucial to the understanding of the driving context.

Design Implications From our study we can derive following design implications:

		Expert Mental Model
Explanation Human Intervention No Human Control	Target Mental ModelObject Detection4Sensors6Trajectory Planning1,5Route Planning3Environment Prediction4,5,8ML and Decision Algorithms5Vehicle Control1	Localization / Tracking Confidence ⁶ Human-Computer Interaction ⁴
Safer Driving Advanced Driving Assistand Systems	Sensor / Data Fusion ^{6,7} Map ⁹ Environmental Model ⁸	
User mental Model		

Figure 6.10: Target mental model consisting of components found in both mental models (shaded) with a selection of the key components (green). Superscript numbers (1-9) indicate which explanation components are used for visualization (Figure 6.5).

- Situation Awareness Aim for increased situation awareness when it is needed to increase safety and acceptance.
- Adaptation Know what to adapt: Aim for optimized system acceptance.
- **Target Mental Model** Aim for a broad target mental model on which most users can agree upon.

6.2.5 Limitations

Within the scope of our work, the driver is not required to take over the steering wheel and therefore has no time limit for explanation interpretation. In a real driving environment the time to interpret and understand the explanation is limited and also the driving context changes rapidly. Therefore, the explanation needs to be adapted situation dependent for the driver to quickly grasp the context of the driving behavior.

As we focused on *what* to visualize, a next step is to evaluate in a structured manner *how* to visualize the explanation and *how* to evaluate the quality of an explanation. In this context, adaptive explanations can be addressed to provide individual optimization for various user types. Considering the expert level of the user can result in different visualization methods. The technical interested user might need an explanation of the deep learning algorithm to fully understand the vehicle's behavior. The optimal time to show the explanation can lead to the question if the driver wants to influence the autonomous driving behavior. Possible advantages could be the constant control of the driving behavior and adaptions through the driver if the current autonomous driving behavior differs from desired or expected actions.

6.3 Lessons Learned

In this chapter, we focused on drivers in unexpected traffic situations, driving in an autonomous vehicle. We first explored the thoughts and statements of participants experiencing a simulated unexpected driving situation. From this qualitative study we gained design implications for autonomous vehicles. Then, in the next section, we developed an explanation for drivers in unexpected driving situations. From this study we identified a target mental model which includes the explanation components to understand the cause of the driving behavior. Furthermore, we increase situation awareness by displaying an explanation to drivers. This explanation is then adapted by the participants. In the following, wo focus on some lessons that we learned during the last two sections.

Realistic scenarios can be derived by analyzing experience reports In the first section we developed 17 scenarios resulting from a web search, which were implemented in a driving simulation. From this first step we examine how real world scenarios can be deployed in a driving simulation. It is an unusual approach, but holds several novel ideas: in other studies, the scenarios are not based on real world scenarios, but on scenarios that the designers of the study have specified. Those are valid and the study design can be easily fitted to those scenarios. For example, studies that use vehicles that are parked on the side [282] are good for conditions that require similar scenarios since it has a high repeatability. In the scenario of a person crossing the road, the person can cross from right, left or with different velocities to avoid a learning effect. Nevertheless, the scenarios stay comparable. With realistic scenarios that are derived from a web search, the scenarios are very diverse with different numbers of participants and in different traffic settings.

In our case we aimed for many scenarios and approached the question about how the participant feels and thinks in diverse scenarios. Therefore, we needed many scenarios. From our approach we can learn that collecting real world examples and use them in a study gives the necessary basis to gain diverse insights of the driver's behavior and thoughts. Our approach can be applied for other studies as well to gain broad insights in case that diverse scenarios are aimed for. Furthermore, it is suitable for studies based on real world experiences without having the means to collect them with current technology. Another lesson learned was that this approach of collecting real world scenarios leaves space for interpretation of the concrete scenario. Every scenario description is subjective and the perception of driving behavior can differ between people. While one describes a driving behavior as stressful or aggressive someone else might describe it as normal driving behavior. Therefore, we got diverse opinions, by discussing scenarios through clustering among several persons. Then one might find an agreement on how a situation might have happened. We argue that the approach of using real world scenarios increases the ecological validity of the study. "Ecological validity [...] examines whether the findings of a study can be generalized to naturalistic situations" [8]. We took good care of including naturalistic situations of autonomous driving behavior in the selected scenarios from the web search. While the autonomous driving behavior will further develop with advancements in the technology, the scenarios will be valid for current studies when aiming for naturalistic situations.

With the codebook we identify overarching themes and correlations between categories. After developing and implementing the scenarios in a driving simulator, we conducted a think-aloud study. In this way we collected qualitative statements of the participants concerning the current driving behavior of the vehicle. While this is an effortful process, the resulting codes build up a diverse codebook with codes concerning every driving impression of the scenario. Through the discussion to develop the categories, coding guidelines can be developed which help in coding consistently. Additionally, overarching themes can be identified and correlations between categories are identified. Sometimes, the content of one category is similar to another category, but the meaning is slightly different. In this case, insights can be gained, by exploring the definition of the category. For example, the code *Mental Mode Autonomous System/Driving* is included in other categories as well for example in the category *Humanization*. In one case, it stands for how participants think the vehicle works and in the other case how drivers behave towards the vehicle. This distinction is important, but is sometimes difficult to code in some statements. Qualitative analysis is a powerful tool to get opinions and information from participants. Nevertheless, the coding is dependent on the coder, therefore, the coders have great responsibility to stay as objective as possible, without a certain goal in mind.

The environment influences the desire for an explanation request. During the study, participants rather requested an explanation when no apparent cause for the unexpected driving behavior is visible. From the codebook (see Table 6.2), it can be derived that participants request interaction options and explanations. Those results give indications for future studies. For the design of a study with focus on explanation interfaces, it might be better to not show the reason for unexpected driving behavior. Then the cause of the behavior can be derived from the explanation interface. Since some sort of interaction is desired by the participants, the focus can be put on the interaction itself. Through future work, it can be identified, if an explanation is sufficient to lessen frustration and dissatisfaction with the driving behavior, or if a take over request is a better option to offer drivers.

The adaption of explanations should take drivers' mental models into account. In Section 6.2, we surveyed the mental models of how people explain how autonomous vehicles perceive their environment and manage to drive on their own, first in a focus group and then in a study. Even though we took great care to evaluate the mental models in depth, the mental models give just an impression of the current opinion of the participant. The mental model changes over time and experience and might differ after experiencing other driving scenarios. Mental models are very individual, depending on the person and their experience, but some attributes are mentioned more frequently than others. It depends on the target design whether the general mental model of users is important or whether individual outliers give more understanding of how adaptive and user centered explanations can be designed.

Explanation should aim for high situation awareness. From the study in Section 6.2, we derive preferred visualization components in explanations for unexpected driving behavior. While the explanation elements differ depending on the situation, the difference is negligible. There is a trend towards the usefulness of some elements such as the object's movement prediction compared to others, such as sensor symbols. Since visualizations should be as easy as possible to swiftly understand the meaning of the situation, the work on developing a target mental model showed that explanations need to be optimized for every use case. The target mental model gives an indication for the concrete case of autonomous vehicles that drive unexpectedly. To evaluate the explanation interface, we used situation awareness and assessed if the explanation improved situation awareness of the driver. In our case, the explanation significantly improved the situation awareness. The quality of an explanation does not solely rely on situation awareness therefore in a next step, other criteria need to be devel-

oped to rate a provided explanation. This can aim for good usability, understanding, stress reduction or unobtrusiveness, to name just a few.

Conclusion The just discussed attributes gave insights on the lessons that we learned with the present chapter. Developing methods to evaluate unexpected driving scenarios, collecting user statements during unexpected driving behavior and then constructing a target mental model for explanations was valuable to derive design guidelines for explanations of unexpected driving behavior. Users' previous knowledge and mental models is crucial to adapt explanations of smart systems. Through the reflection of the results from this chapter, we call for an explanation design that aims for a high level of acceptance instead of completeness and accurate terminology. We derived design implications to design for *adequate control and communication* and *different mental models* in autonomous vehicles (see Section 6.1.5). By developing methods to quantify eligibility of provided explanations, the explanations become comparable and optimizable. With the work of a quantitative analysis and by setting up a codebook, it is possible to develop a standardized approach to quantitatively asses explanations. We have shown the focus and the important categories of explanations in case no explanation is provided to the user. By providing users with explanations, it can be evaluated if these categories improve and if the importance of the categories remains. From the two sections we derived 12 design implications. To sum them up, we have four overarching themes:

Adaptability

With the individual adaptation of, for example, vehicle driving behavior or the interface of the vehicle infotainment system, the user's individual preferences can be taken into account. By optimizing for system acceptance, it can be prevented to adapt features that do not need to be adapted.

Providing information

The user should be provided upon request with various information. Complete and easy explanations, confidence of the vehicle, information about the environment, smart infrastructure information or the internal vehicle status are necessary information to be able to decide upon future driving actions and to form an own opinion about the driving behavior. While driving autonomously, increasing situation awareness through information should be in accordance with the acceptance of the system.

Prediction

Showing future driving decisions and behaviors gives the driver the autonomy to intervene and change the course of driving.

Collaboration

The possibility to collaborate with the environment and the own vehicle, leaves the driver with autonomy.

7 Conclusion

With the results of empirical research projects, we contribute a thorough analysis of the opportunities of smart infrastructure. This thesis investigates the potential of smart infrastructure for the user, with a focus on drivers of autonomous vehicles. We focus on three different aspects: A design space for 3D in-car AR, out-of-view visualizations to create awareness of traffic objects or situations that are outside the view of the driver, and the needs of drivers in unexpected driving situations. Through this research, we gain a thorough understanding of the potential of smart infrastructure for drivers of autonomous vehicles. This chapter first answers the corresponding research questions of this thesis, then summarizes the contributions and finally gives an outlook on future work.

RQ1 What are potential user groups and use cases of smart highway infrastructure? *The potential user groups and use cases are drivers of semi-autonomous vehicles, operators and emergency services. The use cases are identified and collected in a scenario catalog and evaluated in studies.*

The analysis to identify potential stakeholder and use cases for smart infrastructure is based on several interviews and focus groups (see Chapter 3). We identify highway operators, emergency services, drivers of autonomous vehicles and teleoperators as stakeholder of smart infrastructure. Identifying potential user groups is important to analyze individual needs and to find solutions for those. The needs of the highway operators and emergency services include for example accurate traffic flow management or the exact localization of emergencies. Over the course of the exploration, we narrow down the problem space for this thesis to drivers of vehicles. Through the analysis of what it means to be the driver of an autonomous vehicle, we realize that drivers can fulfill different roles, depending on the level of automation of the vehicle. Drivers can either be in an active role, when they are manually driving the vehicle, or with increasing level of automation, the role changes to the role of a supervisor of the vehicle. With fully autonomous vehicles, the role is equal to a passive passenger who is free to be immersed in secondary tasks. Depending on the role in the vehicle, the use cases range from aiding drivers in TORs or explaining unexpected driving behavior to a passenger.

The goal of implementing smart infrastructure is to improve the driving experience and traffic flow for all traffic participants. Through a case study, we collect scenarios in a scenario catalog. These scenarios can be differentiated between scenarios that drivers instantly need to react on and scenarios in which drivers have more time to inform themselves and prepare for upcoming traffic situations. Establishing a scenario catalog enables to design studies and consider the consequences of individual scenarios. If, for example, a traffic jam is hidden behind a curve, the degree of criticality can increase or decrease depending on the distance of the scenario. Smart infrastructure can furthermore realize explanations of unexpected driving behavior by communicating context information in a traffic situation. Summarizing, the case study helps to clarify the problem space and to identify first application purposes for smart infrastructure applications.

RQ2 What are design dimensions for 3D in-car AR applications when 5G and smart infrastructure enable communication, entertainment or more information about the environment? *The*

Conclusion

design dimensions for 3D in-car AR applications that include environment information are identified through a use case set and a design space.

In Chapter 4 we define a design space for 3D in-car AR. From two focus groups, a literature analysis, and the resulting use case set, we derive the design space and its design dimensions. This defines the design dimensions for applications that are set in 3D augmented reality for the interior of vehicles. Use cases span from entertainment applications to navigation or security. In the course of this thesis we benefit from this design space which provides us a tool to design applications for semiautonomous and autonomous vehicles. This design space extends existing design spaces [108, 144] for the use of applications of 3D augmented reality within the vehicle. By limiting the problem space to the interior and excluding windows, many existing use cases are not within our problem space. This reflects the expectation of a changed vehicle interior in future vehicles, in which passengers might be turned towards each other. With user stories and the designs of 3D in-car AR applications, this thesis reflects on design dimensions and the limitations of them. Over the course of this work, we come to the conclusion that novel application cases that exploit the full potential of autonomous vehicles and 5G are still rare. We argue that by providing a use case set of possible applications, the direction of future applications can be shaped. The use cases are defined without considering technical limitations. Therefore, even though the low latency of 5G, necessary for more interactive AR and VR applications in vehicles¹, is not yet available, we define use cases that can be accompanied with our design space in the design process. Therefore, the design dimensions might need to be extended when further applications are realized.

RQ3 How can we display information about events outside driver's field of view? A complete digital twin of the highway prepares drivers for take over requests and hints on objects outside the view of the driver improve situation awareness.

In three driving simulator studies in Chapter 5 we explore possibilities of displaying information of events outside the view of the driver. We display critical traffic events in a digital twin that shows a replication of the highway, and implement different out-of-view techniques to point to critical events. The implemented visualizations increase the situation awareness of drivers and enhance the performance in a TOR. Therefore, we conclude that information from smart infrastructure improves the safety during a take over request.

Most visualization techniques that we explore further in this thesis are new, such as a stereoscopic 3D representation of the highway, or are usually not used in automotive contexts, such as a wedge [106]. Therefore, using these visualization techniques in a new context, the validity is tested. Nevertheless, automotive visualizations have different features than visualizations in virtual reality or games, for example the height dimension is not as relevant when using the whole windscreen, since the street is often on one height level. Therefore, using visualizations from other domains can be useful, as we show it with the minimap visualization, but for some, the requirements of drivers in SAE Level 3 need to be taken into account. While increasing situation awareness with non-intrusive hints on other objects is relevant for drivers that are still required to take over control of the vehicle, visualizations of out-of-view events need more research to fit more adequately to the problem space. Even though we explore multiple out-of view visualizations, we could not find a one-fits-all solution. In situations in which drivers need to take over control from the vehicle, a hint on objects outside their field of

¹ https://www.digi.com/blog/post/5g-applications-and-use-cases, last accessed May 2020

view with the wedge visualization can increase driver's situation awareness. If the situation does not require the driver to take over control immediately, a more detailed visualization, such as a representation of the highway, can be shown to the driver to improve situation awareness and take over performance. All studies were Wizard-of-Oz [193] studies, in which we assume that smart infrastructure can communicate with the vehicle and that out-of-view information is available to the driver. While this limits the ecological validity, the controlled environment in the lab provides the advantage of communicating a concept with minimal effort, and without the technical implementation of several kilometers of smart infrastructure. It enables to explore visualizations which could result in a high workload in take over situations.

RQ4 What do drivers of autonomous vehicles in unexpected driving situations need and how do they want to interact with the vehicle in that moment? Drivers of autonomous vehicles want to have the option to have some sort of control, need the opportunity to interact and to intervene with the vehicle and get an explanation for the unexpected driving behavior.

In a qualitative study (see Chapter 6.1), we evaluate the thoughts of participants while experiencing unexpected driving situations. We derive a codebook, which points out required cockpit elements for autonomous vehicles and derive design recommendations for autonomous vehicles. With the study, we explore the fundamental need of a driver, who is experiencing a driving situation without the option to interact with or change the vehicle's driving behavior in a qualitative driving simulator study.

From this study, we derive that unexpected driving behavior often provoked negative feelings in drivers. Furthermore, drivers were expecting to see the reason for the vehicle's behavior in the environment. This was not the case, therefore, we can conclude that in case of situations in which drivers can not conclude themselves, what caused the driving behavior, an explanation should be provided. Since many drivers compared their driving behavior to the autonomous vehicles' driving behavior, personalizing it should be considered in future vehicles. Furthermore, drivers want to interact or communicate with the vehicle and are confused about the future driving maneuver. This clearly indicates a lack of situation awareness level 3, therefore, interfaces need to take the future driving path into account as well.

The qualitative study combined with realistic driving scenarios provides insights from participants struggling with an automated driving decision. To increase the ecological validity, we base the study on scenarios that we derive from a web search of real world accounts of unexpected driving. The method to derive those can be used by researchers who also struggle with the problem of having to design a study of real life events without having a good understanding of the settings of those events. With that, researchers can choose from 17 unexpected driving scenarios when testing their interaction with autonomous vehicles in unexpected driving situations.

Since not every road type and surely not every unexpected driving scenario is covered by the study, the codebook can be extended to include other categories that are relevant when technological limitations change. Nevertheless, the design dimensions and guidelines based on the codebook, should be taken into account when designing future autonomous vehicle concepts.

RQ5 How can vehicle behavior be explained if an autonomous vehicle reacts unexpectedly on smart infrastructure information? Unexpected driving can be explained by developing a target

Conclusion

mental model and adapting the explanation to the user's preferences. In the course of this thesis, we develop design guidelines for autonomous vehicles that drive unexpectedly.

In Chapter 6 we show how the vehicle behavior can be explained, when the vehicle is reacting on situations that get transmitted through I2V or V2V communication. Through the development of design guidelines, we show that HMI designers should take the drivers in autonomous vehicles into account to enable them to intervene, communicate and receive information. In Section 6.1 we derive design guidelines for designing an interface for drivers in unexpected driving situations. Not just explanations are important in this situation, but also interaction possibilities. The explanations should aim to increase the situation awareness of drivers and the design could also include anthropomorphic features. The driver should be informed about the informations the vehicle received and how reliable the received information is. In Section 6.2 we compare the user mental model to an expert mental model and derive a target mental model, which combines elements of both mental models, for example detection of objects in the surroundings. This can be used to explain the driving behavior of an autonomous vehicle in an unexpected driving situation. The explanation interface increases the situation awareness of the drivers.

7.1 Summary of the Contributions

We contribute with this thesis an exploration of user groups and use cases for smart infrastructure information. Furthermore, we provide a design space for 3D in-car AR to encourage and support designers with a tool to design novel interaction and interfaces for highly automated vehicles, in which the interaction space is centered in the interior of the vehicle.

Next, we contribute an exploration of visualizations that inform drivers of upcoming critical traffic situations and improve the driving performance and situation awareness. We present a novel visualization in S3D that represents the traffic and environment of a highway, to prepare drivers of upcoming critical events. By setting up a scenario catalog, we can distinguish between scenarios in which drivers should be warned immediately and scenarios in which the driver can be informed earlier of a required intervention. We compare HUD visualizations that inform about other traffic participants and warn drivers of upcoming critical situations to increase situation awareness.

We identify drivers' needs in unexpected driving situations. With a systematic web search of experience reports, we explore a method to increase the ecological validity of simulated situations in studies. With the results from a qualitative study, we set up a codebook that contains the categories that concern drivers in unexpected driving situations. Furthermore, we contribute the elements, providing context and situation information, which explanations could contain and sort them according to their importance for drivers. Furthermore, we contribute a target mental model, which contains categories that should be considered when designing explanations for drivers in unexpected driving situations. Finally, we present design guidelines for the interaction with drivers in unexpected driving scenarios.

7.2 Directions of Future Work

From this thesis, researchers and practitioner can derive in-depth insights about applications of information from smart infrastructure which can be further examined. We learned that smart infrastructure offers great potential for drivers on the road. We are faced with new technology such as autonomous vehicles, connected vehicles, brought-in devices or platooning, which can enhance the journey on the road, make traffic flow smoother and reduce the environmental impact. By providing all drivers with information independent of their vehicles' abilities, a shared knowledge base is created. This knowledge base can be retrieved by the system itself, but also by other drivers. Since this thesis has a focus on the highway, the use cases for urban scenarios remain largely unexplored. It is imaginable that through smart traffic lights and connected mobile devices, pedestrians, cyclists or other vulnerable road users greatly benefit from smart infrastructure as well. In the following, we focus on the future direction for design implications, visualizations and explanations in vehicle interfaces with smart infrastructure information.

Design Opportunities of Connecting Smart Infrastructure and Vehicles The work on the design space shows that it is valuable to provide designers with a tool that allows them to explore possible designs and gives them a structured approach to design new concepts and applications. While we validate the design space for applications in the context of this thesis, it has not yet been validated for other design requirements, for example for entertainment purposes. Furthermore, the design space is limited to the technological advances, therefore it needs to be adapted with progressive technical development. For example, the dimensions of a technological implementation to realize S3D need to shrink in their form factor to be implemented in vehicles. This results in new opportunities, not yet existing design dimensions or newly found solutions. With our design space, we hope to encourage designers to explore new use cases and engineers to develop new technologies that enable use cases from our use case set.

Aim for Improved Situation Awareness in Driver's Visualizations So far, not much research has been done to visualize information coming from the infrastructure or connected vehicles. This thesis reveals new aspects that have not yet been fully explored. It is safety critical to not distract the driver from the manual driving task. For autonomous vehicles, this does not apply. Since the automation level of vehicles today shift further to fully autonomous functionalities, the realization of visualizations which require a higher cognitive load becomes possible. The time of the TOR can be extended as the system analyzes data from traffic situations in several kilometers distance. Drivers can be provided with additional information concerning the reason why a take over is necessary.

This work combines visualizations in augmented reality while driving. A study in the wild, under real life conditions would open new challenges. For example, motion sickness while wearing an AR HMD could occur. Static objects in AR or darkening the windows to obscure the moving environment outside could therefore limit the application. The problem of motion sickness refrains many people from reading while driving, even though they would like to spend their time in an autonomous vehicle effectively. While Vovk et al. [280] could not find motion sickness during simulations, they did not test automotive use cases. Motion sickness is not covered in depth in this thesis, but should be considered in further research in dynamic AR visualizations.

Conclusion

Considering our work with a digital twin visualization, a digital twin that shows the whole scenario from a top down view is implementable with smart infrastructure or even just with connected vehicles. Nevertheless, this work implements only a limited number of possible visualizations, and future work should explore different aspects that have not been covered by this work, such as the level of abstraction of the digital twin or how to manage visualizations with a high traffic load.

We can derive from the conducted studies that the problem space of the visualizations should be matched to a clear automation level and situation. One visualization might allow for small distances in a take over request maneuver while another visualization is too distracting in that case and makes more sense while driving fully autonomously. A factor that plays a role in this context is also the mental state of the driver, which has not been considered in this work. This influences the readiness of the driver to take over control of the vehicle. If the driver is fully immersed in a secondary task, it might be more difficult to prepare for the driving task and could therefore reduce the performance during a TOR. To continue this research direction, it would be recommendable to assess the workload of various visualizations in the context of different automation levels. Considering different concept vehicles, it is possible that the location of visualizations changes in future automated vehicles, because the driver is turned away from the steering wheel. The visualizations should then adapt and consider the differences that would arise due to the changed orientation of the driver.

Explaining Autonomous Driving Behavior We propose a concept of what should be included in an explanation in case of unexpected driving behavior. In a next step this concept should be transferred to a quantitative study to assess the advantage of the proposed explanation elements over other explanations. Today, researchers tend to implement system limits in their studies without reflecting on the cause of the system limit. Even though this is valid, a more realistic study design could improve the study validity with a realistic take over maneuver. Many TOR situations that are implemented in studies today, such as a standing vehicle, could already be solved by autonomous vehicles itself. The advantage of implementing realistic TOR scenarios lies in the improvement of credibility for participants as to how they should react in such situations and a more accurate TOR design. The study is limited to the experiences of unexpected driving behavior, therefore it can be expected that the needs change as soon as the technology is further developed, leading to less or different unexpected driving situations. When comparing the results we derived in our study, we see similarities compared to Lee et al. [167]. Their method is different to ours and they did not implement a study, but collected their results with focus groups. Nevertheless, some categories are mentioned in our study as well (e.g. Information Needs, User Interface, Safety & Security). The different method results in categories that are rather generally valid, also for other automation level (e.g. personalization, accessibility and privacy). While our design guidelines apply to a specific problem area, related work recommends further design guidelines [7, 167]. When combined, a more exhaustive guide can be derived how users should be supported in their interaction in autonomous vehicles.

Autonomous driving is a technology that will change the mobility of people in a fundamental way. There is considerable effort from research and industry to enable autonomous driving and smart infrastructure. Nevertheless, industry is still facing technological limitations concerning the implementation of smart infrastructure. Through the results and reflections of this thesis, we derive that traffic participants benefit extensively from the connection with smart infrastructure and paves the way for better integration of autonomous vehicles and smart infrastructure.

LIST OF FIGURES

1.1 1.2	Sensor setup on the highway	5
	and Philipp Quentin)	7
2.1	Examples of augmented reality in the literature:A: A hint on a critical situation after a turn by Werneke et al. [293], B: A concept of a stereoscopic 3D solution by Continental-Automotive ² , C: A projection of a path by Tran et al. [271], D: Several highlighted objects, paths and persons by Park et al. [220]	22
3.1 3.2	Firefighter Station. . Firefighter Station. .	40 41
4.1 4.2	Design Space for In-Car AR	56
4.3	implication example (Section 'Example Scenario') is shown	58 63
5.1	Example use cases and prototypes of the digital twin of the highway. A: Birds eye view with warning symbol, B: Construction site, C: Standing vehicle on the side shoulder, D: Beginning of a traffic jam, E: Simulated traffic scene, F: Real image of traffic scene in E.	70
5.2	Sketches for visualizations of critical situations: Sketch A: Merging, Sketch B: Bad Weather Sketch C: Ghost Driver, Sketch D: Fast vehicle from behind	71
5.3	Visualizations of warning drivers of accidents with context information in AR	72
5.4	VR Cockpit with digital twin and representation of the own vehicle. The secondary task, which is used in the study is placed near the A-Pillar, above the steering wheel. The digital twin is floating above the gearshift in between driver and passenger. Within it, one can see the surrounding traffic situation. The green A indicates the autonomous mode.	74
5.5	Pre-Warning of accident on own lane. The mode indication turns yellow since a take over request is imminent.	74
5.6	Take over request with video task turned off, the proposed driving trajectory is display next to the manual mode indication.	74

² https://www.continental-automotive.com/en-gl/Passenger-Cars/Information-Management/ Display-Solutions, last access April 2020

5.7	Workload scores of the NASA TLX. The error bars show the standard error of each	
	category	78
5.8	Sketches for the pre-study.	81
5.9	Implemented visualizations, from left to right: Baseline, Arrows, Symbol, Wedge	82
5.10	Driving simulator in the study setup.	83
5.11	Situation Awareness Levels and overall SAGAT score.	84
5.12	NASA TLX Scores. The error bars show the standard error.	85
5.13	System Usability Score	86
6.1	Work flow to derive unexpected driving scenarios and conclude design implications	92
6.2	Screenshots from selected scenarios. Top row (from left to right): a) Scenario 1:	
	vehicle just turned right and stopped while an oncoming vehicle passes b) Scenario 2:	
	vehicle is about to turn right, a pedestrian is on the other side of the road c) Scenario 9:	
	a child crosses the street in front of the vehicle. Bottom row: d) Scenario 11: vehicle	
	stops in the city unexpectedly before green traffic lights e) Scenario 8: vehicle waits in	
	front of a crosswalk, a pedestrian is on the sidewalk f) Scenario 3: vehicle is standing	
	on the intersection in the middle of a left turn, an oncoming vehicle passes.	95
6.3	Time in the scenario and the corresponding average explanation request times in black	
	indication for each scenario.	99
6.4	Average explanation requests for each scenario in percent.	101
6.5	Configurable explanation screen consisting of nine visual components	106
6.6	Participant selects explanation interface elements during the study	109
6.7	Frequency at which participants chose the explanation elements.	110
6.8	Elements mentioned during User Mental Model evaluation. The numbers indicate the	
6.0	percentage of participants that mentioned each element.	111
6.9	Situation Awareness of the participants before and after getting provided with an	110
C 10	explanation of the driving scenario.	112
6.10	larget mental model consisting of components found in both mental models (shaded)	
	with a selection of the key components (green). Superscript numbers (1-9) indicate	112
	which explanation components are used for visualization (Figure 6.5).	113

LIST OF TABLES

1	Collaboration in this thesis.	ix
1.1	Summary of the research questions addressed in this thesis	3
4.1	The use case set for entertainment applications. Each category features entries ac- companied by a short title, description and examples from either related work or focus groups. The entries from related work are cited and the results from the fo- cus group are labeled with the primary user group ($D = Driver$, $P = Passenger$, $C =$	
4.2	Co-Driver)	50
	The entries from related work are cited and the results from the focus group are labeled with the primary user group (D = Driver $P = Passenger C = Co_2 Driver$)	52
4.3	The use case set for navigation, convenience and vehicle monitoring applications. Each category features entries accompanied by a short title, description and examples from either related work or focus groups. The entries from related work are cited and the results from the focus group are labeled with the primary user group (D = Driver, $P = Passenger$, $C = Co-Driver$).	54
5.1	Scenario catalog in which the scenarios are sorted in ones that are relevant for warn- ings in short and long distances.	80
6.1	Description of the resulting scenarios, derived from the experience reports. On the left is the related scenario with a short description, in the middle column is the scenario description derived from the discussion while clustering the scenarios and in the right column is the relevant extract from the original scenario description	94
6.2	The codebook of the qualitative study. The codebook is set up of the six categories and the corresponding codes. Related to every code is an abbreviation is on the left,	71
()	the definition of the code, the number of counts and an example of the code	97
0.3	Situation description and the corresponding explanation for the situations	107

REFERENCES

- Ashraf Abdul, Jo Vermeulen, Danding Wang, Brian Y. Lim, and Mohan Kankanhalli. "Trends and Trajectories for Explainable, Accountable and Intelligible Systems: An HCI Research Agenda". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*. CHI '18. ACM, 2018, 582:1–582:18. DOI: 10.1145/3173574.3174156 (cited on p. 25).
- [2] Amina Adadi and Mohammed Berrada. "Peeking Inside the Black-Box: A Survey on Explainable Artificial Intelligence (XAI)". In: *IEEE Access* 6 (2018), pp. 52138–52160. DOI: 10.1109/ACCESS.2018.2870052 (cited on pp. 25, 26).
- [3] Anne Adams, Peter Lunt, and Paul Cairns. "A qualititative approach to HCI research". In: *Research Methods for Human-Computer Interaction*. Cambridge University Press, 2008, pp. 138–157. DOI: 10.1017/CB09780511814570.008 (cited on p. 96).
- [4] Ahmed E Aktan, Arthur J Helmicki, and Victor J Hunt. "Issues in health monitoring for intelligent infrastructure". In: *Smart Materials and Structures* 7.5 (1998), pp. 674–692. DOI: 10.1088/0964-1726/7/5/011 (cited on p. 16).
- [5] Abdul J. Alami, Khaled El-Sayed, Afif Al-Horr, Hassan Artail, and Jinhua Guo. "Improving the Car GPS accuracy using V2V and V2I Communications". In: 2018 IEEE International Multidisciplinary Conference on Engineering Technology (IMCET). 2018, pp. 1–6. DOI: 10. 1109/IMCET.2018.8603032 (cited on p. 10).
- [6] Lauralee Alben. "Quality of Experience: Defining the Criteria for Effective Interaction Design". In: *Interactions* 3.3 (1996), pp. 11–15. DOI: 10.1145/235008.235010 (cited on p. 23).
- [7] Saleema Amershi, Dan Weld, Mihaela Vorvoreanu, Adam Fourney, Besmira Nushi, Penny Collisson, Jina Suh, Shamsi Iqbal, Paul N. Bennett, Kori Inkpen, Jaime Teevan, Ruth Kikin-Gil, and Eric Horvitz. "Guidelines for Human-AI Interaction". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*. CHI '19. Association for Computing Machinery, 2019, pp. 1–13. DOI: 10.1145/3290605.3300233 (cited on pp. 27, 122).
- [8] Chittaranjan Andrade. "Internal, External, and Ecological Validity in Research Design, Conduct, and Evaluation". In: *Indian journal of psychological medicine* 40.5 (2018), pp. 498– 499. DOI: 10.4103/IJPSYM.IJPSYM_334_18 (cited on pp. 5, 114).
- [9] Rolf Koch Andrew J. Woods Tom Docherty. "Image distortions in stereoscopic video systems". In: *Stereoscopic displays and applications IV*. Vol. 1915. International Society for Optics and Photonics, 1993, pp. 36–49. DOI: 10.1117/12.157041 (cited on p. 58).

- [10] Cecilia R. Aragon and Marti A. Hearst. "Improving Aviation Safety with Information Visualization: A Flight Simulation Study". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. CHI '05. Association for Computing Machinery, 2005, pp. 441–450. DOI: 10.1145/1054972.1055033 (cited on pp. 18, 67).
- [11] Alexandre Armand, David Filliat, and Javier Ibañez-Guzman. "Ontology-based context awareness for driving assistance systems". In: 2014 IEEE Intelligent Vehicles Symposium Proceedings. 2014, pp. 227–233. DOI: 10.1109/IVS.2014.6856509 (cited on p. 10).
- [12] azcentral.com. We Followed a Waymo Self-driving Car for Miles, Here's What we Saw. Video. Retrieved September 8, 2019 from https://www.youtube.com/watch?v= spw176TZ7-8.2018 (cited on p. 94).
- [13] Ronald T. Azuma. "A Survey of Augmented Reality". In: *Presence: Teleoperators and Virtual Environments* 6.4 (1997), pp. 355–385. DOI: 10.1162/pres.1997.6.4.355 (cited on p. 20).
- [14] Elizabeth Bacon. "Defining Interaction Design". In: *Interactions* 12.3 (2005), pp. 34–35. DOI: 10.1145/1060189.1060213 (cited on pp. 22, 23).
- [15] Patrick Bader, Valentin Schwind, Niels Henze, Stefan Schneegass, Nora Broy, and Albrecht Schmidt. "Design and Evaluation of a Layered Handheld 3D Display with Touch-sensitive Front and Back". In: *Proceedings of the 8th Nordic Conference on Human-Computer Interaction: Fun, Fast, Foundational.* NordiCHI '14. ACM, 2014, pp. 315–318. DOI: 10.1145/ 2639189.2639257 (cited on p. 59).
- [16] Victoria A. Banks, Alexander Eriksson, Jim O'Donoghue, and Neville A. Stanton. "Is partially automated driving a bad idea? Observations from an on-road study". In: Applied Ergonomics 68 (2018), pp. 138–145. DOI: https://doi.org/10.1016/j.apergo.2017. 11.010 (cited on p. 9).
- [17] Pavlo Bazilinskyy, Sebastiaan Petermeijer, Veronika Petrovych, Dimitra Dodou, and Joost de Winter. "Take-over requests in highly automated driving: A crowdsourcing multimedia survey on auditory, vibrotactile, and visual displays". In: *Transportation Research Part F Traffic Psychology and Behaviour* 56 (2018), pp. 82–98. DOI: 10.1016/j.trf.2018.04. 001 (cited on p. 13).
- [18] Johannes Beller, Matthias Heesen, and Mark Vollrath. "Improving the driver-automation interaction: An approach using automation uncertainty". In: *Human factors* 55.6 (2013), pp. 1130–1141. DOI: 10.1177/0018720813482327 (cited on p. 29).
- [19] Carl Bergenhem, Steven Shladover, Erik Coelingh, Christoffer Englund, and Sadayuki Tsugawa. "Overview of platooning systems". In: *Proceedings of the 19th ITS World Congress, Oct* 22-26, Vienna, Austria (2012). 2012 (cited on p. 7).
- [20] Viktor Berggren, Aneta Vulgarakis, Athanasios Karapantelakis, Elena Fersman, Keven Wang, Leonid Mokrushin, Nicolas Schrammar, and Rafia Inam. 5G teleoperated vehicles for future public transport. https://www.ericsson.com/en/blog/2017/6/5g-teleoperatedvehicles-for-future-public-transport. Accessed: 2020-05. 2017 (cited on p. 44).
- [21] Mark Billinghurst, Adrian Clark, and Gun Lee. "A Survey of Augmented Reality". In: Foundations and Trends Human-Computer Interaction 8.2-3 (2015), pp. 73–272. DOI: 10.1561/ 1100000049 (cited on p. 57).
- [22] Myra Blanco, Jon Atwood, Holland M Vasquez, Tammy E Trimble, Vikki L Fitchett, Joshua Radlbeck, Gregory M Fitch, Sheldon M Russell, Charles A Green, Brian Cullinane, et al. "Human factors evaluation of level 2 and level 3 automated driving concepts". In: *Meyer G., Beiker S. (eds) Road Vehicle Automation 3. Lecture Notes in Mobility.* 2015, pp. 456–461. DOI: 10.1007/978-3-319-40503-2_9 (cited on pp. 13, 30).
- [23] Mate Boban, Apostolos Kousaridas, Konstantinos Manolakis, Josef Eichinger, and Wen Xu. "Connected Roads of the Future: Use Cases, Requirements, and Design Considerations for Vehicle-to-Everything Communications". In: *IEEE Vehicular Technology Magazine* 13.3 (2018), pp. 110–123 (cited on p. 44).
- [24] F. Boccardi, R. W. Heath, A. Lozano, T. L. Marzetta, and P. Popovski. "Five disruptive technology directions for 5G". In: *IEEE Communications Magazine* 52.2 (2014), pp. 74–80. DOI: 10.1109/MC0M.2014.6736746 (cited on pp. 17, 47).
- [25] Mariusz Bojarski, Philip Yeres, Anna Choromanska, Krzysztof Choromanski, Bernhard Firner, Lawrence Jackel, and Urs Muller. Explaining How a Deep Neural Network Trained with End-to-End Learning Steers a Car. 2017. arXiv: 1704.07911 [cs.CV] (cited on p. 26).
- [26] Felix Bork, Christian Schnelzer, Ulrich Eck, and Nassir Navab. "Towards Efficient Visual Guidance in Limited Field-of-View Head-Mounted Displays". In: *IEEE Transactions on Vi*sualization and Computer Graphics 24.11 (2018), pp. 2983–2992. DOI: 10.1109/TVCG. 2018.2868584 (cited on p. 80).
- [27] Shadan Sadeghian Borojeni, Lewis Chuang, Wilko Heuten, and Susanne Boll. "Assisting Drivers with Ambient Take-Over Requests in Highly Automated Driving". In: *Proceedings* of the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. Automotive'UI 16. Association for Computing Machinery, 2016, pp. 237–244. DOI: 10.1145/3003715.3005409 (cited on pp. 14, 47).
- [28] Shadan Sadeghian Borojeni, Lars Weber, Wilko Heuten, and Susanne Boll. "From Reading to Driving: Priming Mobile Users for Take-over Situations in Highly Automated Driving". In: *Proceedings of the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services*. MobileHCI '18. Association for Computing Machinery, 2018. DOI: 10.1145/3229434.3229464 (cited on pp. 14, 30, 47).
- [29] Martijn Bout, Anna Pernestål Brenden, Maria Klingegård, Azra Habibovic, and Marc-Philipp Böckle. "A Head-Mounted Display to Support Teleoperations of Shared Automated Vehicles". In: Proceedings of the 9th International Conference on Automotive User Interfaces and Interactive Vehicular Applications Adjunct. AutomotiveUI '17. Association for Computing Machinery, 2017, pp. 62–66. DOI: 10.1145/3131726.3131758 (cited on p. 44).
- [30] Doug Bowman, Ernst Kruijff, Joseph J LaViola Jr, and Ivan P Poupyrev. 3D User interfaces: theory and practice. Addison-Wesley, 2004 (cited on p. 57).
- [31] Tobias Brandt. "Information Systems in Automobiles Past, Present, and Future Uses". In: (2013). DOI: 10.13140/RG.2.1.3337.9688 (cited on pp. 52, 55).
- [32] Michael Braun, Anja Mainz, Ronee Chadowitz, Bastian Pfleging, and Florian Alt. "At Your Service: Designing Voice Assistant Personalities to Improve Automotive User Interfaces". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*. CHI '19. Association for Computing Machinery, 2019, pp. 1–11. DOI: 10.1145/3290605.3300270 (cited on p. 103).

- [33] Virginia Braun and Victoria Clarke. "Using thematic analysis in psychology". In: *Qualitative Research in Psychology* 3.2 (2006), pp. 77–101. DOI: 10.1191/1478088706qp063oa. eprint: https://www.tandfonline.com/doi/pdf/10.1191/1478088706qp063oa (cited on pp. 92, 93).
- [34] John Brooke. "SUS-A quick and dirty usability scale". In: *Usability evaluation in industry* 189.194 (1996), pp. 4–7 (cited on p. 79).
- [35] Nora Broy. "Stereoscopic 3D user interfaces: exploring the potentials and risks of 3D displays in cars". PhD thesis. 2016. DOI: http://dx.doi.org/10.18419/opus-8851 (cited on pp. 52-54, 57).
- [36] Nora Broy, Florian Alt, Stefan Schneegass, and Bastian Pfleging. "3D Displays in Cars: Exploring the User Performance for a Stereoscopic Instrument Cluster". In: ACM Press, 2014, pp. 1–9. DOI: 10.1145/2667317.2667319 (cited on pp. 52–54).
- [37] Nora Broy, Mengbing Guo, Stefan Schneegass, Bastian Pfleging, and Florian Alt. "Introducing Novel Technologies in the Car: Conducting a Real-world Study to Test 3D Dashboards". In: *Proceedings of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '15. ACM, 2015, pp. 179–186. DOI: 10.1145/2799250.2799280 (cited on pp. 19, 21, 47, 52–54, 58).
- [38] Nora Broy, Verena Lindner, and Florian Alt. "The S3D-UI Designer: Creating User Interface Prototypes for 3D Displays". In: *Proceedings of the 15th International Conference on Mobile and Ubiquitous Multimedia*. MUM '16. ACM, 2016, pp. 49–55. DOI: 10.1145/3012709. 3012727 (cited on p. 47).
- [39] John L Campbell, Christian M Richard, James L Brown, and Marvin McCallum. "Crash warning system interfaces: human factors insights and lessons learned". In: *DOT HS* 810 (2007), p. 697 (cited on pp. 19, 20, 30).
- [40] Claudia Campolo, Antonella Molinaro, Antonio Iera, and Francesco Menichella. "5G Network Slicing for Vehicle-to-Everything Services". In: *IEEE Wireless Communications* 24.6 (2017), pp. 38–45. DOI: 10.1109/MWC.2017.1600408 (cited on p. 18).
- [41] Struart K. Card and Jock Mackinlay. "The structure of the information visualization design space". In: Proceedings of VIZ '97: Visualization Conference, Information Visualization Symposium and Parallel Rendering Symposium. 1997, pp. 92–99. DOI: 10.1109/INFVIS.1997. 636792 (cited on p. 24).
- [42] John M Carroll. Making use: scenario-based design of human-computer interactions. MIT press, 2000 (cited on p. 26).
- [43] Oliver M. J. Carsten and Marieke H. Martens. "How can humans understand their automated cars? HMI principles, problems and solutions". In: *Cognition, Technology & Work* 21 (2018), pp. 3–20. DOI: 10.1007/s10111-018-0484-0 (cited on p. 27).
- [44] Alison Cawsey. "User modelling in interactive explanations". In: User Modeling and User-Adapted Interaction 3.3 (1993), pp. 221–247. DOI: 10.1007/BF01257890 (cited on p. 104).
- [45] Kathy Charmaz. Constructing grounded theory: A practical guide through qualitative analysis. Sage Publications Limited, 2006 (cited on p. 96).
- [46] Michelene TH Chi. "Three types of conceptual change: Belief revision, mental model transformation, and categorical shift". In: *International handbook of research on conceptual change*. Routledge, 2009, pp. 89–110. DOI: 10.4324/9780203874813-9 (cited on p. 27).

- [47] Damee Choi, Toshihisa Sato, Takafumi Ando, Takashi Abe, Motoyuki Akamatsu, and Satoshi Kitazaki. "Effects of cognitive and visual loads on driving performance after take-over request (TOR) in automated driving". In: *Applied Ergonomics* 85 (2020), p. 103074. DOI: https://doi.org/10.1016/j.apergo.2020.103074 (cited on pp. 11, 13).
- [48] Jong Kyu Choi and Yong Gu Ji. "Investigating the Importance of Trust on Adopting an Autonomous Vehicle". In: *International Journal of Human–Computer Interaction* 31.10 (2015), pp. 692–702. DOI: 10.1080/10447318.2015.1070549. eprint: https://doi.org/10.1080/10447318.2015.1070549 (cited on p. 14).
- [49] Lewis L. Chuang, Stella F. Donker, Andrew L. Kun, and Christian P. Janssen. "Workshop on The Mobile Office". In: Adjunct Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '18. ACM, 2018, pp. 10–16. DOI: 10.1145/3239092.3239094 (cited on p. 55).
- [50] Andy Cockburn, Amy Karlson, and Benjamin B. Bederson. "A Review of Overview+detail, Zooming, and Focus+context Interfaces". In: ACM Computing Surveys 41.1 (2009). DOI: 10.1145/1456650.1456652 (cited on p. 19).
- [51] Caroline Conti, Joao Lino, Paulo Nunes, and Luis D. Soares. "Spatial and temporal prediction scheme for 3D holoscopic video coding based on H.264/AVC". In: 2012 19th International Packet Video Workshop (PV). 2012, pp. 143–148. DOI: 10.1109/PV.2012.6229727 (cited on p. 59).
- [52] Arthur P. Cracknell and Ladson W.B. Hayes. Introduction to remote sensing. Taylor & Francis, 2007 (cited on p. 10).
- [53] Henriette Cramer, Vanessa Evers, Satyan Ramlal, Maarten Van Someren, Lloyd Rutledge, Natalia Stash, Lora Aroyo, and Bob Wielinga. The effects of transparency on trust in and acceptance of a content-based art recommender. Vol. 18. 5. 2008, pp. 455–496. DOI: 10. 1007/s11257-008-9051-3 (cited on pp. 25, 105).
- [54] Arnaud De La Fortelle, Xiangjun Qian, Sébastien Diemer, Jean Grégoire, Fabien Moutarde, Silvère Bonnabel, Ali Marjovi, Alcherio Martinoli, Ignacio Llatser, Andreas Festag, and Katrin Sjöberg. "Network of automated vehicles: the AutoNet 2030 vision". In: *ITS World Congress*. electronic proceedings. 2014 (cited on pp. 1, 16).
- [55] Nico DeMattia. BMW unveiled its BMW i Inside Future sculpture at CES 2017. https: //www.bmwblog.com/2017/01/04/bmw-unveiled-its-bmw-i-inside-futuresculpture-at-ces-2017/. Accessed: 2020-05. 2006 (cited on p. 50).
- [56] Inc DesignworksUSA. BMW Holographic Interface. http://www. bmwgroupdesignworks.com/work/bmw-i-inside-future/. Accessed: 2020-05. 2006 (cited on pp. 20, 47, 52-54).
- [57] Morton Deutsch. "The Effect of Motivational Orientation upon Trust and Suspicion". In: *Human Relations* 13.2 (1960), pp. 123–139. DOI: 10.1177/001872676001300202. eprint: https://doi.org/10.1177/001872676001300202 (cited on p. 15).
- [58] Anind K Dey. "Understanding and using context". In: *Personal and ubiquitous computing* 5.1 (2001), pp. 4–7 (cited on p. 18).
- [59] Arindam Dey, Mark Billinghurst, Robert W. Lindeman, and J. Edward Swan. "A Systematic Review of 10 Years of Augmented Reality Usability Studies: 2005 to 2014". In: *Frontiers in Robotics and AI* 5 (2018), p. 37. DOI: 10.3389/frobt.2018.00037 (cited on pp. 20, 25).

- [60] Stefan Diewald, Andreas Möller, Luis Roalter, Matthias Kranz, et al. "Mobile device integration and interaction in the automotive domain". In: *AutoNUI: Automotive Natural User Interfaces Workshop at the 3rd International Conference on Automotive User Interfaces and Interactive Vehicular Applications (AutomotiveUI 2011)*. 2011 (cited on p. 24).
- [61] Murat Dikmen and Catherine M. Burns. "Autonomous Driving in the Real World: Experiences with Tesla Autopilot and Summon". In: *Proceedings of the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. Automotive'UI 16. Association for Computing Machinery, 2016, pp. 225–228. DOI: 10.1145/3003715. 3005465 (cited on p. 89).
- [62] Dikmen, Murat. Investigating Driver Experience and Augmented Reality Head-Up Displays in Autonomous Vehicles. 2017 (cited on p. 21).
- [63] Yadolah Dodge. "Kolmogorov–Smirnov Test". In: *The Concise Encyclopedia of Statistics*. Springer New York, 2008, pp. 283–287. DOI: 10.1007/978-0-387-32833-1_214 (cited on pp. 77, 84).
- [64] Neil A. Dodgson. "Autostereoscopic 3D displays". In: *Computer* 38.8 (2005), pp. 31–36.
 DOI: 10.1109/MC.2005.252 (cited on p. 58).
- [65] Finale Doshi-Velez and Been Kim. "Towards A Rigorous Science of Interpretable Machine Learning". In: Ml (2017), pp. 1–13. arXiv: 1702.08608 (cited on pp. 25, 26, 31).
- [66] Graham Dove, Nicolai Brodersen Hansen, and Kim Halskov. "An Argument For Design Space Reflection". In: *Proceedings of the 9th Nordic Conference on Human-Computer Interaction*. NordiCHI '16. Association for Computing Machinery, 2016. DOI: 10.1145 / 2971485.2971528 (cited on p. 23).
- [67] Mary T. Dzindolet, Scott A. Peterson, Regina A. Pomranky, Linda G. Pierce, and Hall P. Beck.
 "The role of trust in automation reliance". In: *International Journal of Human-Computer Studies* 58.6 (2003). Trust and Technology, pp. 697–718. DOI: https://doi.org/10.1016/S1071-5819(03)00038-7 (cited on p. 25).
- [68] Lilian Edwards and Michael Veale. "Slave to the Algorithm? Why a 'Right to an Explanation' Is Probably Not the Remedy You Are Looking For". In: *Duke Law & Technology Review 18* 16 (2017), p. 18. DOI: 10.2139/ssrn.2972855 (cited on p. 25).
- [69] Amir Efrati. With Waymo Robotaxis, Customer Satisfaction Is Far From Guaranteed. Web. Retrieved September 8, 2019 from https://www.theinformation.com/articles/ with-waymo-robotaxis-customer-satisfaction-is-far-from-guaranteed?. 2019 (cited on p. 94).
- [70] Malin Eiband, Daniel Buschek, Alexander Kremer, and Heinrich Hussmann. "The impact of placebic explanations on trust in intelligent systems". In: *Conference on Human Factors in Computing Systems - Proceedings* (2019). DOI: 10.1145/3290607.3312787 (cited on p. 28).
- [71] Malin Eiband, Hanna Schneider, Mark Bilandzic, Julian Fazekas-Con, Mareike Haug, and Heinrich Hussmann. "Bringing Transparency Design into Practice". In: 23rd International Conference on Intelligent User Interfaces. IUI '18. ACM, 2018, pp. 211–223. DOI: 10.1145/ 3172944.3172961 (cited on pp. 29, 104–106).

- [72] Mica R Endsley. "Situation awareness global assessment technique (SAGAT)". In: Proceedings of the IEEE 1988 National Aerospace and Electronics Conference. IEEE. 1988, 789–795 vol.3. DOI: 10.1109/NAECON.1988.195097 (cited on pp. 12, 67).
- [73] Mica R Endsley. "Toward a theory of situation awareness in dynamic systems". In: *Situational Awareness*. Routledge, 2017, pp. 9–42 (cited on p. 109).
- [74] Mica R. Endsley. "Situation Awareness in Future Autonomous Vehicles: Beware of the Unexpected". In: *Proceedings of the 20th Congress of the International Ergonomics Association* (*IEA 2018*). Ed. by Sebastiano Bagnara, Riccardo Tartaglia, Sara Albolino, Thomas Alexander, and Yushi Fujita. Springer International Publishing, 2019, pp. 303–309 (cited on p. 105).
- [75] Mica R. Endsley. "Toward a Theory of Situation Awareness in Dynamic Systems". In: *Human Factors* 37.1 (1995), pp. 32–64. DOI: 10.1518/001872095779049543. eprint: https://doi.org/10.1518/001872095779049543 (cited on p. 11).
- [76] Tristan C. Endsley, Kelly A. Sprehn, Ryan M. Brill, Kimberly J. Ryan, Emily C. Vincent, and James M. Martin. "Augmented Reality Design Heuristics: Designing for Dynamic Interactions". In: *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* 61.1 (2017), pp. 2100–2104. DOI: 10.1177/1541931213602007. eprint: https://doi.org/10.1177/1541931213602007 (cited on p. 24).
- [77] Barrett Ens, Juan David Hincapié-Ramos, and Pourang Irani. "Ethereal Planes: A Design Framework for 2D Information Spaces in 3D Mixed Reality Environments". In: SUI 2014 -Proceedings of the 2nd ACM Symposium on Spatial User Interaction (2014). DOI: 10.1145/ 2659766.2659769 (cited on p. 24).
- [78] Sandra Epple, Fabienne Roche, and Stefan Brandenburg. "The Sooner the Better: Drivers' Reactions to Two-Step Take-Over Requests in Highly Automated Driving". In: *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* 62.1 (2018), pp. 1883– 1887. DOI: 10.1177/1541931218621428. eprint: https://doi.org/10.1177/ 1541931218621428 (cited on pp. 13, 30).
- [79] Alexander Eriksson, Sebastiaan M. Petermeijer, Markus Zimmermann, Joost C. F. de Winter, Klaus J. Bengler, and Neville A. Stanton. "Rolling Out the Red (and Green) Carpet: Supporting Driver Decision Making in Automation-to-Manual Transitions". In: *IEEE Transactions* on Human-Machine Systems 49.1 (2019), pp. 20–31. DOI: 10.1109/THMS.2018.288386 (cited on p. 14).
- [80] Alexander Eriksson and Neville A. Stanton. "Takeover Time in Highly Automated Vehicles: Noncritical Transitions to and From Manual Control". In: *Human Factors The Journal of the Human Factors and Ergonomics Society* 59.4 (2017). PMID: 28124573, pp. 689–705. DOI: 10.1177/0018720816685832. eprint: https://doi.org/10.1177/0018720816685832 (cited on p. 13).
- [81] Sarah Faltaous, Martin Baumann, Stefan Schneegass, and Lewis L. Chuang. "Design Guidelines for Reliability Communication in Autonomous Vehicles". In: *Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '18. Association for Computing Machinery, 2018, pp. 258–267. DOI: 10.1145/3239060.3239072 (cited on p. 29).
- [82] Rui Fan, Jianhao Jiao, Haoyang Ye, Yang Yu, Ioannis Pitas, and Ming Liu. "Key Ingredients of Self-Driving Cars". In: *CoRR* abs/1906.02939 (2019). arXiv: 1906.02939 (cited on p. 10).

- [83] Ryan Felton. Google's Self-Driving Cars Have Trouble With Basic Driving Tasks: Report. Video. Retrieved September 8, 2019 from https://jalopnik.com/googles-selfdriving-cars-have-trouble-with-basic-drivi-1828653280. 2018 (cited on p. 94).
- [84] Frank Flemisch, Matthias Heesen, Tobias Hesse, Johann Kelsch, Anna Schieben, and Johannes Beller. "Towards a Dynamic Balance between Humans and Automation: Authority, Ability, Responsibility and Control in Shared and Cooperative Control Situations". In: *Cognition Technology & Work* 14.1 (2012), pp. 3–18. DOI: 10.1007/s10111-011-0191-6 (cited on p. 12).
- [85] Uwe Flick. An introduction to qualitative research. Sage Publications Limited, 2018 (cited on p. 96).
- [86] Yannick Forster, Frederik Naujoks, Alexandra Neukum, and Lynn Huestegge. "Driver compliance to take-over requests with different auditory outputs in conditional automation". In: *Accident Analysis & Prevention* 109 (2017), pp. 18–28. DOI: https://doi.org/10.1016/ j.aap.2017.09.019 (cited on pp. 11, 13).
- [87] Lex Fridman, Li Ding, Benedikt Jenik, and Bryan Reimer. "Arguing Machines: Human Supervision of Black Box AI Systems That Make Life-Critical Decisions". In: *CoRR* abs/1710.04459 (2019). arXiv: 1710.04459 (cited on p. 29).
- [88] Anna-Katharina Frison, Philipp Wintersberger, Tianjia Liu, and Andreas Riener. "Why Do You Like to Drive Automated?: A Context-dependent Analysis of Highly Automated Driving to Elaborate Requirements for Intelligent User Interfaces". In: *Proceedings of the 24th International Conference on Intelligent User Interfaces*. IUI '19. ACM, 2019, pp. 528–537. DOI: 10.1145/3301275.3302331 (cited on p. 28).
- [89] Peter Fröhlich, Matthias Baldauf, Marion Hagen, Stefan Suette, Dietmar Schabus, and Andrew L. Kun. "Investigating Safety Services on the Motorway: The Role of Realistic Visualization". In: *Proceedings of the 3rd International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '11. Association for Computing Machinery, 2011, pp. 143–150. DOI: 10.1145/2381416.2381440 (cited on p. 21).
- [90] Ken Funk, Beth Lyall, Jennifer Wilson, Rebekah Vint, Mary Niemczyk, Candy Suroteguh, and Griffith Owen. "Flight Deck Automation issues". In: *The International Journal of Aviation Psychology* 9.2 (1999), pp. 109–123. DOI: 10.1207/s15327108ijap0902_2. eprint: https://doi.org/10.1207/s15327108ijap0902_2 (cited on p. 13).
- [91] Jean-Michael Georg, Johannes Feiler, Frank Diermeyer, and Markus Lienkamp. "Teleoperated Driving, a Key Technology for Automated Driving? Comparison of Actual Test Drives with a Head Mounted Display and Conventional Monitors". In: 2018 21st International Conference on Intelligent Transportation Systems (ITSC). IEEE. 2018, pp. 3403–3408. DOI: 10. 1109/ITSC.2018.8569408 (cited on p. 44).
- [92] Mario Gerla, Eun-Kyu Lee, Giovanni Pau, and Uichin Lee. "Internet of vehicles: From intelligent grid to autonomous cars and vehicular clouds". In: 2014 IEEE World Forum on Internet of Things (WF-IoT). 2014, pp. 241–246. DOI: 10.1109/WF-IoT.2014.6803166 (cited on pp. 17, 67).

- [93] Christian Gold, Daniel Damböck, Lutz Lorenz, and Klaus Bengler. ""Take over!" How long does it take to get the driver back into the loop?" In: *Proceedings of the Human Factors* and Ergonomics Society Annual Meeting 57.1 (2013), pp. 1938–1942. DOI: 10.1177 / 1541931213571433. eprint: https://doi.org/10.1177/1541931213571433 (cited on pp. 13, 30, 67, 68).
- [94] Keyvan Golestan, Ridha Soua, Fakhri Karray, and Mohamed S. Kamel. "Situation awareness within the context of connected cars: A comprehensive review and recent trends". In: *Information Fusion* 29 (2016), pp. 68–83. DOI: https://doi.org/10.1016/j.inffus.2015.08.001 (cited on pp. 10, 11).
- [95] Chaohua Gong. "Human-Computer Interaction: Process and Principles of Human-Computer Interface Design". In: 2009 International Conference on Computer and Automation Engineering. 2009, pp. 230–233. DOI: 10.1109/ICCAE.2009.23 (cited on p. 18).
- [96] John D. Gould and Clayton Lewis. "Designing for Usability: Key Principles and What Designers Think". In: *Commun. ACM* 28.3 (1985), pp. 300–311. DOI: 10.1145/3166.3170 (cited on p. 23).
- [97] Mark D. Gross. "Design as exploring constraints". PhD thesis. Massachusetts Institute of Technology, 1985 (cited on p. 23).
- [98] Jens Grubert, Tobias Langlotz, Stefanie Zollmann, and Holger Regenbrecht. "Towards Pervasive Augmented Reality: Context-Awareness in Augmented Reality". In: *IEEE Transactions* on Visualization and Computer Graphics 23.6 (2017), pp. 1706–1724. DOI: 10.1109/TVCG. 2016.2543720 (cited on pp. 21, 55).
- [99] Uwe Gruenefeld, Dag Ennenga, Abdallah El Ali, Wilko Heuten, and Susanne Boll. "Eye-See360: Designing a Visualization Technique for out-of-View Objects in Head-Mounted Augmented Reality". In: *Proceedings of the 5th Symposium on Spatial User Interaction*. SUI '17. Association for Computing Machinery, 2017, pp. 109–118. DOI: 10.1145/3131277. 3132175 (cited on pp. 19, 81).
- [100] Uwe Gruenefeld, Ilja Koethe, Daniel Lange, Sebastian Weiß, and Wilko Heuten. "Comparing Techniques for Visualizing Moving Out-of-View Objects in Head-mounted Virtual Reality".
 In: 2019 IEEE Conference on Virtual Reality and 3D User Interfaces (VR). 2019, pp. 742–746. DOI: 10.1109/VR.2019.8797725 (cited on pp. 18, 80).
- [101] Uwe Gruenefeld, Andreas Löcken, Yvonne Brueck, Susanne Boll, and Wilko Heuten. "Where to Look: Exploring Peripheral Cues for Shifting Attention to Spatially Distributed Out-of-View Objects". In: Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '18. Association for Computing Machinery, 2018, pp. 221–228. DOI: 10.1145/3239060.3239080 (cited on pp. 2, 19).
- [102] Leo Gugerty et al. "Situation awareness in driving". In: *Handbook for driving simulation in engineering, medicine and psychology* 1 (2011) (cited on pp. 11, 12).
- [103] Leo J Gugerty. "Situation awareness during driving: Explicit and implicit knowledge in dynamic spatial memory." In: *Journal of Experimental Psychology: Applied* 3.1 (1997), pp. 42– 66. DOI: 10.1037/1076-898X.3.1.42 (cited on pp. 11, 12).

- [104] Riccardo Guidotti, Anna Monreale, Salvatore Ruggieri, Franco Turini, Fosca Giannotti, and Dino Pedreschi. "A Survey of Methods for Explaining Black Box Models". In: ACM Computing Surveys 51.5 (2018). DOI: 10.1145/3236009 (cited on pp. 25, 26).
- [105] David Gunning. "Explainable artificial intelligence (xai)". In: *Defense Advanced Research Projects Agency (DARPA), nd Web* 2 (2017) (cited on pp. 26, 27).
- [106] Sean Gustafson, Patrick Baudisch, Carl Gutwin, and Pourang Irani. "Wedge: Clutter-free Visualization of Off-screen Locations". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. CHI '08. ACM, 2008, pp. 787–796. DOI: 10.1145/1357054. 1357179 (cited on pp. 19, 81, 118).
- [107] Renate Haeuslschmid, Max von Bülow, Bastian Pfleging, and Andreas Butz. "Supporting Trust in Autonomous Driving". In: *Proceedings of the 22Nd International Conference on Intelligent User Interfaces*. IUI '17. ACM, 2017, pp. 319–329. DOI: 10.1145/3025171. 3025198 (cited on pp. 29, 52).
- [108] Renate Haeuslschmid, Bastian Pfleging, and Florian Alt. "A Design Space to Support the Development of Windshield Applications for the Car". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. CHI '16. ACM, 2016, pp. 5076–5091. DOI: 10.1145/2858036.2858336 (cited on pp. 2, 24, 25, 43, 48, 49, 51, 53, 55–57, 59, 118).
- [109] Renate Haeuslschmid, Laura Schnurr, Julie Wagner, and Andreas Butz. "Contact-Analog Warnings on Windshield Displays Promote Monitoring the Road Scene". In: *Proceedings* of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '15. Association for Computing Machinery, 2015, pp. 64–71. DOI: 10.1145/2799250.2799274 (cited on p. 18).
- Stephen M Haley and J Scott Osberg. "Kappa Coefficient Calculation Using Multiple Ratings Per Subject: A Special Communication". In: *Physical Therapy* 69.11 (1989), pp. 970–974.
 DOI: 10.1093/ptj/69.11.970. eprint: https://academic.oup.com/ptj/articlepdf/69/11/970/10758904/ptj0970.pdf (cited on p. 96).
- [111] Joanne L Harbluk, Y Ian Noy, and Moshe Eizenman. The impact of cognitive distraction on driver visual behaviour and vehicle control. Tech. rep. 2002 (cited on p. 53).
- [112] Ryan J Harrington, Carmine Senatore, John M Scanlon, and Ryan M Yee. "The Role of Infrastructure in an Automated Vehicle Future". In: *The Bridge* 48.2 (2018) (cited on p. 67).
- [113] Sandra G. Hart and Lowell E. Staveland. "Development of NASA-TLX (Task Load Index): Results of Empirical and Theoretical Research". In: *Human Mental Workload*. Ed. by Peter A. Hancock and Najmedin Meshkati. Vol. 52. Advances in Psychology. North-Holland, 1988, pp. 139–183. DOI: https://doi.org/10.1016/S0166-4115(08)62386-9 (cited on pp. 75, 79).
- [114] Jacob Haspiel, Na Du, Jill Meyerson, Lionel P. Robert Jr., Dawn Tilbury, X. Jessie Yang, and Anuj K. Pradhan. "Explanations and Expectations: Trust Building in Automated Vehicles". In: *Companion of the 2018 ACM/IEEE International Conference on Human-Robot Interaction*. HRI '18. Association for Computing Machinery, 2018, pp. 119–120. DOI: 10.1145/ 3173386.3177057 (cited on p. 29).

- [115] Andrew J. Hawkins. Riding in Waymo One, the Google spinoff's first self-driving taxi service. online. Retrieved September 8, 2019 from https://www.theverge.com/2018/12/ 5/18126103/waymo-one-self-driving-taxi-service-ride-safety-alphabetcost-app. 2018 (cited on p. 94).
- [116] Remo M.A. van der Heiden, Shamsi T. Iqbal, and Christian P. Janssen. "Priming Drivers Before Handover in Semi-Autonomous Cars". In: *Proceedings of the 2017 CHI Conference* on Human Factors in Computing Systems. CHI '17. ACM, 2017, pp. 392–404. DOI: 10. 1145/3025453.3025507 (cited on p. 18).
- [117] Tove Helldin, Göran Falkman, Maria Riveiro, and Staffan Davidsson. "Presenting System Uncertainty in Automotive UIs for Supporting Trust Calibration in Autonomous Driving". In: Proceedings of the 5th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '13. Association for Computing Machinery, 2013, pp. 210–217. DOI: 10.1145/2516540.2516554 (cited on p. 104).
- [118] Charlie Hewitt, Ioannis Politis, Theocharis Amanatidis, and Advait Sarkar. "Assessing Public Perception of Self-driving Cars: The Autonomous Vehicle Acceptance Model". In: *Proceedings of the 24th International Conference on Intelligent User Interfaces*. IUI '19. ACM, 2019, pp. 518–527. DOI: 10.1145/3301275.3302268 (cited on p. 91).
- [119] Gereon Hinz, Martin Büchel, Frederik Diehl, Malte Schellmann, and Alois Knoll. "Designing a far-reaching view for highway traffic scenarios with 5G-based intelligent infrastructure". In: 8. Tagung Fahrerassistenz. 2017 (cited on p. 29).
- [120] Gereon Hinz, Josef Eichinger, Martin Büchel, and Alois Knoll. "Proactive Video-based Use of Telecommunications Technologies in Innovative Motorway Scenarios". In: 5th GI/ITG KuVS Fachgespräch Inter-Vehicle Communication (2017), p. 41 (cited on pp. 1, 5, 16, 39).
- [121] Laurens Hobert, Andreas Festag, Ignacio Llatser, Luciano Altomare, Filippo Visintainer, and Andras Kovacs. "Enhancements of V2X communication in support of cooperative autonomous driving". In: *IEEE Communications Magazine* 53.12 (2015), pp. 64–70. DOI: 10. 1109/MC0M.2015.7355568 (cited on pp. 1, 16, 17).
- [122] Jean-Michel Hoc. "From human–machine interaction to human–machine cooperation". In: *Ergonomics* 43.7 (2000), pp. 833–843. DOI: 10.1080/001401300409044 (cited on p. 91).
- [123] Kevin Hoff and Masooda Bashir. "Trust in Automation: Integrating Empirical Evidence on Factors That Influence Trust". In: *Human Factors The Journal of the Human Factors and Ergonomics Society* 57 (2015), pp. 407–434. DOI: 10.1177/0018720814547570 (cited on p. 15).
- [124] Robert R. Hoffman, Shane T. Mueller, Gary Klein, and Jordan Litman. "Metrics for Explainable AI: Challenges and Prospects". In: *CoRR* abs/1812.04608 (2018). arXiv: 1812.04608 (cited on p. 26).
- [125] Nicolas S. Holliman, Neil A. Dodgson, Gregg E. Favalora, and Lachlan Pockett. "Three-Dimensional Displays: A Review and Applications Analysis". In: *IEEE Transactions on Broadcasting* 57.2 (2011), pp. 362–371. DOI: 10.1109/TBC.2011.2130930 (cited on p. 58).

- [126] Tim Horberry, Janet Anderson, Michael A. Regan, Thomas J. Triggs, and John Brown. "Driver distraction: The effects of concurrent in-vehicle tasks, road environment complexity and age on driving performance". In: Accident Analysis & Prevention 38.1 (2006), pp. 185– 191. DOI: https://doi.org/10.1016/j.aap.2005.09.007 (cited on p. 53).
- [127] Mark Horswill and Frank Mckenna. "Drivers' hazard perception ability: Situation awareness on the road". In: A cognitive approach to situation awareness: Theory and application (2004), pp. 155–175 (cited on p. 11).
- [128] Russell Hudyma, Michael Thomas, Paul Rose, and Rick Dorval. "Volumetric Three-Dimensional Display System". 2007 (cited on p. 59).
- [129] Matthew Inman. 6 things I learned from riding in a Google Self-driving Car. online. Retrieved September 8, 2019 from https://theoatmeal.com/blog/google_self_driving_car. 2018 (cited on p. 94).
- [130] SAE International. Taxonomy and definitions for terms related to driving automation systems for on-road motor vehicles. 2016 (cited on pp. 3, 9, 12, 38, 48, 53, 55, 68, 74, 96, 105).
- [131] Toni Ivergard and Brian Hunt. Handbook of control room design and ergonomics: a perspective for the future. CRC Press, 2008 (cited on p. 45).
- [132] Christian P. Janssen, Shamsi T. Iqbal, Andrew L. Kun, and Stella F. Donker. "Interrupted by my car? Implications of interruption and interleaving research for automated vehicles". In: *International Journal of Human-Computer Studies* 130 (2019), pp. 221–233. DOI: https: //doi.org/10.1016/j.ijhcs.2019.07.004 (cited on p. 92).
- [133] Jan-Keno Janssen. We drove the first public autonomous bus in Germany. Video. Retrieved September 8, 2019 from https://www.youtube.com/watch?v=K58wQvP15Fs. 2018 (cited on p. 94).
- [134] Monique Jaspers, Thiemo Steen, Cor Bos, and Maud Geenen. "The think aloud method: A guide to user interface design". In: *International journal of medical informatics* 73.11-12 (2004), pp. 781–95. DOI: 10.1016/j.ijmedinf.2004.08.003 (cited on pp. 91, 95).
- [135] Sankar Jayaram, Judy Vance, Rajit Gadh, Uma Jayaram, and Hari Srinivasan. "Assessment of VR technology and its applications to engineering problems". In: *Journal of Computing and Information Science in Engineering* 1.1 (2001), pp. 72–83. DOI: 10.1115/1.1353846 (cited on p. 58).
- [136] Jiun-Yin Jian, Ann M. Bisantz, and Colin G. Drury. "Foundations for an Empirically Determined Scale of Trust in Automated Systems". In: *International Journal of Cognitive Ergonomics* 4.1 (2000), pp. 53–71. DOI: 10.1207/S15327566IJCE0401_04. eprint: https: //doi.org/10.1207/S15327566IJCE0401_04 (cited on p. 72).
- [137] Philip N. Johnson-Laird. Mental Models. 1989 (cited on p. 103).
- [138] Gareth R. Jones and Jennifer M. George. "The Experience and Evolution of Trust: Implications for Cooperation and Teamwork". In: Academy of Management Review 23.3 (1998), pp. 531–546. DOI: 10.5465/amr.1998.926625. eprint: https://doi.org/10.5465/amr.1998.926625 (cited on p. 15).
- [139] Graham R. Jones, Delman Lee, Nicolas S. Holliman, and David Ezra. "Controlling perceived depth in stereoscopic images". In: *Stereoscopic Displays and Virtual Reality Systems VIII*. Vol. 4297. International Society for Optics and Photonics, 2001, pp. 42–54 (cited on p. 58).

- [140] Helmut Jorke, Arnold Simon, and Markus Fritz. "Advanced stereo projection using interference filters". In: Journal of the Society for Information Display 17.5 (2009), pp. 407–410. DOI: 10.1889/JSID17.5.407. eprint: https://onlinelibrary.wiley.com/doi/pdf/ 10.1889/JSID17.5.407 (cited on p. 58).
- [141] Denis Kalkofen, Erick Mendez, and Dieter Schmalstieg. "Interactive Focus and Context Visualization for Augmented Reality". In: *Proceedings of the 2007 6th IEEE and ACM International Symposium on Mixed and Augmented Reality*. ISMAR '07. IEEE Computer Society, 2007, pp. 1–10. DOI: 10.1109/ISMAR.2007.4538846 (cited on pp. 18, 67).
- [142] Eunsuk Kang, Ethan Jackson, and Wolfram Schulte. "An Approach for Effective Design Space Exploration". In: *Foundations of Computer Software. Modeling, Development, and Verification of Adaptive Systems.* Ed. by Radu Calinescu and Ethan Jackson. FOCS'10. Springer Berlin Heidelberg, 2011, pp. 33–54 (cited on p. 48).
- [143] Georgios Karagiannis, Onur Altintas, Eylem Ekici, Geert Heijenk, Boangoat Jarupan, Kenneth Lin, and Timothy Weil. "Vehicular Networking: A Survey and Tutorial on Requirements, Architectures, Challenges, Standards and Solutions". In: *IEEE Communications Surveys & Tutorials* 13.4 (2011), pp. 584–616. DOI: 10.1109/SURV.2011.061411.00019 (cited on p. 33).
- [144] Dagmar Kern and Albrecht Schmidt. "Design Space for Driver-based Automotive User Interfaces". In: Proceedings of the 1st International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '09. ACM, 2009, pp. 3–10. DOI: 10. 1145/1620509.1620511 (cited on pp. 2, 24, 43, 48, 49, 59, 118).
- [145] Siddartha Khastgir, Stewart Birrell, Gunwant Dhadyalla, and Paul Jennings. "Calibrating trust through knowledge: Introducing the concept of informed safety for automation in vehicles". In: *Transportation Research Part C: Emerging Technologies* 96 (2018), pp. 290–303. DOI: 10.1016/j.trc.2018.07.001 (cited on p. 15).
- [146] Hyun Goo R. Kim, Dora E. Angelaki, and Gregory C. DeAngelis. "The neural basis of depth perception from motion parallax". In: *Philosophical Transactions of the Royal Society B: Biological Sciences* 371.1697 (2016). DOI: 10.1098/rstb.2015.0256 (cited on p. 59).
- [147] Hyung J. Kim and Ji H. Yang. "Takeover Requests in Simulated Partially Autonomous Vehicles Considering Human Factors". In: *IEEE Transactions on Human-Machine Systems* 47.5 (2017), pp. 735–740. DOI: 10.1109/THMS.2017.2674998 (cited on p. 13).
- [148] Jeamin Koo, Jungsuk Kwac, Wendy Ju, Martin Steinert, Larry Leifer, and Clifford Nass. "Why did my car just do that? Explaining semi-autonomous driving actions to improve driver understanding, trust, and performance". In: *International Journal on Interactive Design and Manufacturing* 9.4 (2015), pp. 269–275. DOI: 10.1007/s12008-014-0227-2 (cited on pp. 28, 91).
- [149] Jeamin Koo, Dongjun Shin, Martin Steinert, and Larry Leifer. "Understanding driver responses to voice alerts of autonomous car operations". In: *International Journal of Vehicle Design* 70.4 (2016), pp. 377–392. DOI: 10.1504/IJVD.2016.076740. eprint: https://www.inderscienceonline.com/doi/pdf/10.1504/IJVD.2016.076740 (cited on p. 91).

- [150] Moritz Körber, Lorenz Prasch, and Klaus Bengler. "Why Do I Have to Drive Now? Post Hoc Explanations of Takeover Requests". In: *Human Factors The Journal of the Human Factors* and Ergonomics Society 60 (2018), pp. 305–323. DOI: 10.1177/0018720817747730 (cited on p. 29).
- [151] Robert Krajewski, Julian Bock, Laurent Kloeker, and Lutz Eckstein. "The highD Dataset: A Drone Dataset of Naturalistic Vehicle Trajectories on German Highways for Validation of Highly Automated Driving Systems". In: 2018 21st International Conference on Intelligent Transportation Systems (ITSC). 2018, pp. 2118–2125. DOI: 10.1109/ITSC.2018.8569552 (cited on p. 16).
- [152] Annkathrin Krämmer, Christoph Schöller, Dhiraj Gulati, and Alois Knoll. Providentia A Large Scale Sensing System for the Assistance of Autonomous Vehicles. 2019. arXiv: 1906. 06789 [cs.R0] (cited on pp. 1, 16).
- [153] Arun Kulshreshth and Joseph J. LaViola. "Dynamic Stereoscopic 3D Parameter Adjustment for Enhanced Depth Discrimination". In: *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems*. CHI '16. Association for Computing Machinery, 2016, pp. 177–187. DOI: 10.1145/2858036.2858078 (cited on p. 59).
- [154] Andrew L. Kun, Susanne Boll, and Albrecht Schmidt. "Shifting Gears: User Interfaces in the Age of Autonomous Driving". In: *IEEE Pervasive Computing* 15.1 (2016), pp. 32–38. DOI: 10.1109/MPRV.2016.14 (cited on p. 48).
- [155] Andrew L. Kun, Hidde van der Meulen, and Christian Janssen. "Calling While Driving: An Initial Experiment with HoloLens". In: (2017), pp. 200–206. DOI: 10.17077 / drivingassessment.1636 (cited on pp. 21, 50, 52).
- [156] Alexander Kunze, Stephen J. Summerskill, Russell Marshall, and Ashleigh J. Filtness. "Conveying Uncertainties Using Peripheral Awareness Displays in the Context of Automated Driving". In: *Proceedings of the 11th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '19. Association for Computing Machinery, 2019, pp. 329–341. DOI: 10.1145/3342197.3344537 (cited on pp. 19, 92).
- [157] Jinke D. Van Der Laan, Adriaan Heino, and Dick De Waard. "A simple procedure for the assessment of acceptance of advanced transport telematics". In: *Transportation Research Part C: Emerging Technologies* 5.1 (1997), pp. 1–10. DOI: https://doi.org/10.1016/S0968-090X(96)00025-3 (cited on p. 75).
- [158] Marc Lambooij, Marten Fortuin, Ingrid Heynderickx, and Wijnand IJsselsteijn. "Visual discomfort and visual fatigue of stereoscopic displays: A review". In: *Journal of Imaging Science and Technology* 53.3 (2009), pp. 30201–1 (cited on p. 58).
- [159] Sabine Langlois and Boussaad Soualmi. "Augmented reality versus classical HUD to take over from automated driving: An aid to smooth reactions and to anticipate maneuvers". In: 2016 IEEE 19th International Conference on Intelligent Transportation Systems (ITSC). 2016, pp. 1571–1578. DOI: 10.1109/ITSC.2016.7795767 (cited on pp. 21, 22).
- [160] David R. Large, Kyle Harrington, Gary Burnett, Jacob Luton, Peter Thomas, and Pete Bennett. "To Please in a Pod: Employing an Anthropomorphic Agent-Interlocutor to Enhance Trust and User Experience in an Autonomous, Self-Driving Vehicle". In: *Proceedings of the 11th International Conference on Automotive User Interfaces and Interactive Vehicular Ap-*

plications. AutomotiveUI '19. Association for Computing Machinery, 2019, pp. 49–59. DOI: 10.1145/3342197.3344545 (cited on p. 92).

- [161] Felix Lauber, Sophia Cook, and Andreas Butz. "Content Destabilization for Head-Mounted Displays". In: *Proceedings of the 33rd Annual ACM Conference on Human Factors in Computing Systems*. CHI '15. ACM, 2015, pp. 2139–2142. DOI: 10.1145/2702123.2702161 (cited on p. 56).
- [162] Eric Laurier. "Doing Office Work on the Motorway". In: *Theory, Culture & Society* 21.4-5 (2004), pp. 261–277. DOI: 10.1177/0263276404046070. eprint: https://doi.org/10.1177/0263276404046070 (cited on p. 55).
- [163] Jonathan Lazar, Jinjuan Heidi Feng, and Harry Hochheiser. Research methods in humancomputer interaction. Morgan Kaufmann, 2017 (cited on pp. 3, 4, 34).
- [164] Oscar Lazaro, Eric Robert, Lin Lan, Javier Gozalvez, Siebe Turksma, Fethi Filali, Fabio Cartolano, MA Urrutia, and Daniel Krajzewicz. "iTETRIS: An Integrated wireless and traffic platform for real-time road traffic management solutions". In: WWRF 21, Meeting of Wireless World Research Forum, October 13-15, 2008, Stockholm, Sweden. 2008 (cited on p. 17).
- [165] Jeong-Woo Lee, Chang-Rak Yoon, Jungyu Kang, Byoung-Jun Park, and Kyong-Ho Kim.
 "Development of lane-level guidance service in vehicle augmented reality system". In: 2015 17th International Conference on Advanced Communication Technology (ICACT). 2015, pp. 263–266. DOI: 10.1109/ICACT.2015.7224799 (cited on pp. 20, 21).
- [166] John D. Lee and Katrina A. See. "Trust in Automation: Designing for Appropriate Reliance". In: *Human Factors* 46.1 (2004). PMID: 15151155, pp. 50–80. DOI: 10.1518/hfes.46.1. 50_30392. eprint: https://doi.org/10.1518/hfes.46.1.50_30392 (cited on pp. 15, 29).
- [167] Seul Chan Lee, Chihab Nadri, Harsh Sanghavi, and Myounghoon Jeon. "Exploring User Needs and Design Requirements in Fully Automated Vehicles". In: *Extended Abstracts of the* 2020 CHI Conference on Human Factors in Computing Systems. CHI EA '20. Association for Computing Machinery, 2020, pp. 1–9. DOI: 10.1145/3334480.3382881 (cited on pp. 27, 122).
- [168] Michael G. Lenné, Thomas J. Triggs, Christine M. Mulvihill, Michael A. Regan, and Bruce F. Corben. "Detection of Emergency Vehicles: Driver Responses to Advance Warning in a Driving Simulator". In: *Human Factors* 50.1 (2008). PMID: 18354977, pp. 135–144. DOI: 10.1518/001872008X250557. eprint: https://doi.org/10.1518/001872008X250557 (cited on p. 19).
- [169] Bruno Lepri, Nuria Oliver, Emmanuel Letouzé, Alex Pentland, and Patrick Vinck. "Fair, transparent, and accountable algorithmic decision-making processes". In: *Philosophy & Technology* 31.4 (2018), pp. 611–627. DOI: 10.1007/s13347-017-0279-x (cited on p. 25).
- [170] Brian Y. Lim and Anind K. Dey. "Evaluating Intelligibility Usage and Usefulness in a Context-Aware Application". In: *Human-Computer Interaction. Towards Intelligent and Implicit Interaction*. Ed. by Masaaki Kurosu. Springer Berlin Heidelberg, 2013, pp. 92–101 (cited on p. 26).

- [171] Brian Y. Lim, Anind K. Dey, and Daniel Avrahami. "Why and Why Not Explanations Improve the Intelligibility of Context-Aware Intelligent Systems". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. CHI '09. Association for Computing Machinery, 2009, pp. 2119–2128. DOI: 10.1145/1518701.1519023 (cited on p. 105).
- [172] Youn-Kyung Lim, Erik Stolterman, and Josh Tenenberg. "The Anatomy of Prototypes: Prototypes as Filters, Prototypes as Manifestations of Design Ideas". In: ACM Transactions Computer-Human Interaction 15.2 (2008). DOI: 10.1145/1375761.1375762 (cited on p. 23).
- [173] Patrick Lindemann, Tae-Young Lee, and Gerhard Rigoll. "Supporting Driver Situation Awareness for Autonomous Urban Driving with an Augmented-Reality Windshield Display". In: 2018 IEEE International Symposium on Mixed and Augmented Reality Adjunct (ISMAR-Adjunct). 2018, pp. 358–363. DOI: 10.1109/ISMAR-Adjunct.2018.00104 (cited on pp. 21, 22).
- [174] Patrick Lindemann and Gerhard Rigoll. "Exploring Floating Stereoscopic Driver-car Interfaces with Wide Field-of-view in a Mixed Reality Simulation". In: *Proceedings of the* 22Nd ACM Conference on Virtual Reality Software and Technology. VRST '16. ACM, 2016, pp. 331–332. DOI: 10.1145/2993369.2996299 (cited on pp. 21, 52–54).
- [175] Lars Lischke, Sven Mayer, Andreas Preikschat, Markus Schweizer, Ba Vu, Paweł W. Wozniak, and Niels Henze. "Understanding Large Display Environments: Contextual Inquiry in a Control Room". In: *Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems*. CHI EA '18. Association for Computing Machinery, 2018, pp. 1–6. DOI: 10.1145/3170427.3188621 (cited on p. 45).
- [176] Todd Litman. Autonomous vehicle implementation predictions. Victoria Transport Policy Institute Victoria, Canada, 2017 (cited on p. 47).
- [177] Chang Liu, Hong Zheng, and Chang Zong. "Study on Control Strategy of Steer-by-Wire Vehicle with Joystick". In: Advanced Materials Research 694-697 (2013), pp. 2061–2064. DOI: 10.4028/www.scientific.net/AMR.694-697.2061 (cited on p. 39).
- [178] Stein C Llanos and Kristine Jørgensen. "Do players prefer integrated user interfaces? A qualitative study of game UI design issues". In: *Proceedings of DiGRA 2011 Conference: Think Design Play.* 2011 (cited on p. 81).
- [179] Lutz Lorenz, Philipp Kerschbaum, and Josef Schumann. "Designing take over scenarios for automated driving: How does augmented reality support the driver to get back into the loop?" In: *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* 58.1 (2014), pp. 1681–1685. DOI: 10.1177/1541931214581351. eprint: https://doi.org/10.1177/1541931214581351 (cited on p. 21).
- [180] Michael N Louka, Morten A Gustavsen, and Svein Tore Edvardsen. "Using virtual reality to support multi-participant human-centered design processes for control room design". In: *Proceedings of 5th international topical meeting on nuclear plant instrumentation, controls, and human machine interface technology (NPIC & HMIT 2006) at the American Nuclear Society 2006 meeting, November. American Nuclear Society, Albuquerque.* 2006 (cited on p. 45).

- [181] Carlos Andrés Lozano-Garzon. "V2X Communications to support ITS services". In: 2019 2nd Latin American Conference on Intelligent Transportation Systems (ITS LATAM). IEEE. 2019, pp. 1–1. DOI: 10.1109/ITSLATAM.2019.8721328 (cited on p. 17).
- [182] N. Lu, N. Cheng, N. Zhang, X. Shen, and J. W. Mark. "Connected Vehicles: Solutions and Challenges". In: *IEEE Internet of Things Journal* 1.4 (2014), pp. 289–299. DOI: 10.1109/ JIOT.2014.2327587 (cited on p. 17).
- [183] Zhenji Lu and Joost C.F. de Winter. "A Review and Framework of Control Authority Transitions in Automated Driving". In: *Procedia Manufacturing* 3 (2015). 6th International Conference on Applied Human Factors and Ergonomics (AHFE 2015) and the Affiliated Conferences, AHFE 2015, pp. 2510–2517. DOI: https://doi.org/10.1016/j.promfg.2015. 07.513 (cited on p. 14).
- [184] Ernst Lueder. 3D Displays. Vol. 34. John Wiley & Sons, 2012. DOI: 10.1002 / 9781119962762 (cited on pp. 57, 58).
- [185] Thorsten Luettel, Michael Himmelsbach, and Hans-Joachim Wuensche. "Autonomous ground vehicles - Concepts and a path to the future". In: *Proceedings of the IEEE* 100.Special Centennial Issue (2012), pp. 1831–1839. DOI: 10.1109/JPROC.2012.2189803 (cited on p. 29).
- [186] Joseph B. Lyons, Garrett G. Sadler, Kolina Koltai, Henri Battiste, Nhut T. Ho, Lauren C. Hoffmann, David Smith, Walter Johnson, and Robert Shively. "Shaping Trust Through Transparent Design: Theoretical and Experimental Guidelines". In: Advances in Human Factors in Robots and Unmanned Systems. Ed. by Pamela Savage-Knepshield and Jessie Chen. Springer International Publishing, 2017, pp. 127–136 (cited on p. 25).
- [187] Allan MacLean, Richard M. Young, Victoria M.E. Bellotti, and Thomas P. Moran. "Questions, Options, and Criteria: Elements of Design Space Analysis". In: *Human–Computer Interaction* 6.3-4 (1991), pp. 201–250. DOI: 10.1080/07370024.1991.9667168. eprint: https://www.tandfonline.com/doi/pdf/10.1080/07370024.1991.9667168 (cited on p. 23).
- [188] Hiroshi Makishita and Katsuya Matsunaga. "Differences of drivers' reaction times according to age and mental workload". In: Accident Analysis & Prevention 40.2 (2008), pp. 567–575. DOI: https://doi.org/10.1016/j.aap.2007.08.012 (cited on p. 67).
- [189] Steve Mann, Tom Furness, Yu Yuan, Jay Iorio, and Zixin Wang. "All Reality: Virtual, Augmented, Mixed (X), Mediated (X, Y), and Multimediated Reality". In: *CoRR* abs/1804.08386 (2018). arXiv: 1804.08386 (cited on p. 56).
- [190] Stephen Marsland. Machine learning: an algorithmic perspective. Taylor&Francis, 2015 (cited on p. 25).
- [191] Marieke Martens and Arie van den Beukel. "The road to automated driving: Dual mode and human factors considerations". In: 2013, pp. 2262–2267. DOI: 10.1109/ITSC.2013. 6728564 (cited on p. 9).
- [192] Makoto Matsumoto and Takuji Nishimura. "Mersenne Twister: A 623-dimensionally Equidistributed Uniform Pseudo-random Number Generator". In: ACM Transactions Modeling Computer Simulation 8.1 (1998), pp. 3–30. DOI: 10.1145/272991.272995 (cited on p. 95).

- [193] David Maulsby, Saul Greenberg, and Richard Mander. "Prototyping an Intelligent Agent through Wizard of Oz". In: *Proceedings of the INTERACT '93 and CHI '93 Conference on Human Factors in Computing Systems*. CHI '93. Association for Computing Machinery, 1993, pp. 277–284. DOI: 10.1145/169059.169215 (cited on p. 119).
- [194] Markus Maurer, J Christian Gerdes, Barbara Lenz, Hermann Winner, et al. Autonomous driving. Vol. 10. 2016, pp. 978–3. DOI: 10.1007/978-3-662-48847-8 (cited on pp. 10, 28, 39, 43).
- [195] Steffen Maurer, Rainer Erbach, Issam Kraiem, Susanne Kuhnert, Petra Grimm, and Enrico Rukzio. "Designing a Guardian Angel: Giving an Automated Vehicle the Possibility to Override Its Driver". In: *Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '18. Association for Computing Machinery, 2018, pp. 341–350. DOI: 10.1145/3239060.3239078 (cited on p. 92).
- [196] Lydia M. J. Meesters, Wijnand A. IJsselsteijn, and Pieter J. H. Seuntiens. "A survey of perceptual evaluations and requirements of three-dimensional TV". In: *IEEE Transactions on Circuits and Systems for Video Technology* 14.3 (2004), pp. 381–391. DOI: 10.1109/TCSVT. 2004.823398 (cited on p. 58).
- [197] Vivien Melcher, Stefan Rauh, Frederik Diederichs, Harald Widlroither, and Wilhelm Bauer. "Take-Over Requests for Automated Driving". In: *Procedia Manufacturing* 3 (2015). 6th International Conference on Applied Human Factors and Ergonomics (AHFE 2015) and the Affiliated Conferences, AHFE 2015, pp. 2867–2873. DOI: https://doi.org/10.1016/j. promfg.2015.07.788 (cited on pp. 13, 30, 67, 68).
- [198] Paul Milgram and Fumio Kishino. "A Taxonomy of Mixed Reality Visual Displays". In: vol. E77-D, no. 12 (1994), pp. 1321–1329 (cited on p. 56).
- [199] Tim Miller, Piers Howe, and Liz Sonenberg. "Explainable AI: Beware of Inmates Running the Asylum Or: How I Learnt to Stop Worrying and Love the Social and Behavioural Sciences". In: (2017). arXiv: 1712.00547 (cited on pp. 28, 103).
- [200] Friedman Milton. "A correction: The use of ranks to avoid the assumption of normality implicit in the analysis of variance". In: *Journal of the American Statistical Association. American Statistical Association* 34.205 (1939), p. 109 (cited on pp. 76, 77, 85).
- [201] Florian Mirus, Peter Blouw, Terrence C. Stewart, and Jörg Conradt. "An Investigation of Vehicle Behavior Prediction Using a Vector Power Representation to Encode Spatial Positions of Multiple Objects and Neural Networks". In: *Frontiers in Neurorobotics* 13 (2019), p. 84. DOI: 10.3389/fnbot.2019.00084 (cited on p. 18).
- [202] Mathias Mohring, Christian Lessig, and Oliver Bimber. "Video see-through AR on consumer cell-phones". In: *Third IEEE and ACM International Symposium on Mixed and Augmented Reality*. 2004, pp. 252–253. DOI: 10.1109/ISMAR.2004.63 (cited on p. 58).
- [203] Brian Mok, Mishel Johns, Key J. Lee, David Miller, David Sirkin, Page Ive, and Wendy Ju. "Emergency, Automation Off: Unstructured Transition Timing for Distracted Drivers of Automated Vehicles". In: 2015 IEEE 18th International Conference on Intelligent Transportation Systems. 2015, pp. 2458–2464. DOI: 10.1109/ITSC.2015.396 (cited on pp. 13, 30, 67, 68).

- [204] Ann Morrison, Antti Oulasvirta, Peter Peltonen, Saija Lemmela, Giulio Jacucci, Gerhard Reitmayr, Jaana Näsänen, and Antti Juustila. "Like Bees around the Hive: A Comparative Study of a Mobile Augmented Reality Map". In: *Proceedings of the SIGCHI Conference* on Human Factors in Computing Systems. CHI '09. Association for Computing Machinery, 2009, pp. 1889–1898. DOI: 10.1145/1518701.1518991 (cited on p. 20).
- [205] Ghada Moussa, Essam Radwan, and Khaled Hussain. "Augmented Reality Vehicle system: Left-turn maneuver study". In: *Transportation Research Part C: Emerging Technologies* 21.1 (2012), pp. 1–16. DOI: https://doi.org/10.1016/j.trc.2011.08.005 (cited on pp. 20, 21).
- [206] Jörg Müller, Florian Alt, Daniel Michelis, and Albrecht Schmidt. "Requirements and Design Space for Interactive Public Displays". In: *Proceedings of the 18th ACM International Conference on Multimedia*. MM '10. ACM, 2010, pp. 1285–1294. DOI: 10.1145/1873951. 1874203 (cited on pp. 23, 49, 57, 59).
- [207] Elizabeth Mynatt, Jennifer Clark, Greg Hager, Dan Lopresti, Greg Morrisett, Klara Nahrstedt, George Pappas, Shwetak Patel, Jennifer Rexford, Helen Wright, et al. "A national research agenda for intelligent infrastructure". In: *arXiv preprint arXiv:1705.01920* (2017) (cited on p. 67).
- [208] Shojiro Nagata. "How to Reinforce Perception of Depth in Single Two-Dimensional Pictures". In: *Pictorial Communication in Virtual and Real Environments*. Taylor & Francis, Inc., 1991, pp. 527–545 (cited on p. 59).
- [209] Frederik Naujoks, Christoph Mai, and Alexandra Neukum. "The effect of urgency of takeover requests during highly automated driving under distraction conditions". In: Advances in Human Aspects of Transportation 7.Part I (2014), p. 431 (cited on p. 13).
- [210] Frederik Naujoks and Ingo Totzke. "Behavioral adaptation caused by predictive warning systems The case of congestion tail warnings". In: *Transportation Research Part F: Traffic Psychology and Behaviour* 26 (2014), pp. 49–61. DOI: https://doi.org/10.1016/j.trf.2014.06.010 (cited on p. 14).
- [211] Trung Thanh Nguyen, Kai Holländer, Marius Hoggenmueller, Callum Parker, and Martin Tomitsch. "Designing for Projection-Based Communication between Autonomous Vehicles and Pedestrians". In: *Proceedings of the 11th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '19. Association for Computing Machinery, 2019, pp. 284–294. DOI: 10.1145/3342197.3344543 (cited on p. 103).
- [212] Anne Nkoro and Yuri A Vershinin. "Current and future trends in applications of Intelligent Transport Systems on cars and infrastructure". In: 17th International IEEE Conference on Intelligent Transportation Systems (ITSC). IEEE. 2014, pp. 514–519. DOI: 10.1109/ITSC. 2014.6957741 (cited on pp. 29, 67).
- [213] Shirin Noei, Hugo Santana, Arman Sargolzaei, and Mohammadreza Noei. "Reducing Traffic Congestion Using Geo-Fence Technology: Application for Emergency Car". In: *Proceedings* of the 1st International Workshop on Emerging Multimedia Applications and Services for Smart Cities. EMASC '14. Association for Computing Machinery, 2014, pp. 15–20. DOI: 10.1145/2661704.2661709 (cited on p. 40).
- [214] Don Norman. The design of everyday things: Psychologie und Design der alltäglichen Dinge. Verlag Franz Vahlen, 2016. DOI: 10.15358/9783800648108 (cited on p. 22).

- [215] Donald A. Norman. "Some Observations on Mental Models". In: *Mental Models* 7.112 (1983), pp. 7–14 (cited on p. 103).
- [216] Robert Ighodaro Ogie, Pascal Perez, and Virginia Dignum. "Smart infrastructure: an emerging frontier for multidisciplinary research". In: *Proceedings of the Institution of Civil Engineers - Smart Infrastructure and Construction* 170.1 (2017), pp. 8–16. DOI: 10.1680/ jsmic.16.00002. eprint: https://doi.org/10.1680/jsmic.16.00002 (cited on pp. 1, 16).
- [217] Panos Papadimitratos, Arnaud D. La Fortelle, Knut Evenssen, Roberto Brignolo, and Stefano Cosenza. "Vehicular communication systems: Enabling technologies, applications, and future outlook on intelligent transportation". In: *IEEE Communications Magazine* 47.11 (2009), pp. 84–95. DOI: 10.1109/MCOM.2009.5307471 (cited on p. 17).
- [218] Raja Parasuraman and Victor Riley. "Humans and Automation: Use, Misuse, Disuse, Abuse". In: *Human Factors* 39.2 (1997), pp. 230–253. DOI: 10.1518/001872097778543886. eprint: https://doi.org/10.1518/001872097778543886 (cited on p. 15).
- [219] Raja Parasuraman, Thomas B. Sheridan, and Christopher D. Wickens. "A model for types and levels of human interaction with automation". In: *IEEE Transactions on Systems, Man,* and Cybernetics - Part A: Systems and Humans 30.3 (2000), pp. 286–297. DOI: 10.1109/ 3468.844354 (cited on p. 14).
- [220] Byoung-Jun Park, Jeong-Woo Lee, Changrak Yoon, and Kyong-Ho Kim. "A Study on Augmented Reality of a Vehicle Information Using Head-Up Display". In: 2016 6th International Conference on IT Convergence and Security (ICITCS). 2016, pp. 1–2. DOI: 10.1109/ ICITCS.2016.7740308 (cited on p. 22).
- [221] Siegmund Pastoor and Matthias Wöpking. "3-D displays: A review of current technologies". In: *Displays* 17.2 (1997), pp. 100–110. DOI: 10.1016/S0141-9382(96)01040-2 (cited on p. 58).
- [222] Luke Petersen, Lionel Robert, X. Jessie Yang, and Dawn Tilbury. "Situational Awareness, Driver's Trust in Automated Driving Systems and Secondary Task Performance". In: SAE International Journal of Connected and Autonomous Vehicles, Forthcoming (2019). DOI: 10. 2139/ssrn.3345543 (cited on pp. 12, 29).
- [223] Bastian Pfleging, Maurice Rang, and Nora Broy. "Investigating User Needs for Non-Driving-Related Activities during Automated Driving". In: *Proceedings of the 15th International Conference on Mobile and Ubiquitous Multimedia*. MUM '16. Association for Computing Machinery, 2016, pp. 91–99. DOI: 10.1145/3012709.3012735 (cited on pp. 24, 43, 48).
- [224] Tran Pham, Jo Vermeulen, Anthony Tang, and Lindsay MacDonald Vermeulen. "Scale Impacts Elicited Gestures for Manipulating Holograms: Implications for AR Gesture Design". In: *Proceedings of the 2018 Designing Interactive Systems Conference*. DIS '18. ACM, 2018, pp. 227–240. DOI: 10.1145/3196709.3196719 (cited on pp. 57, 59).
- [225] Pick up was too long. online. Retrieved September 8, 2019 from https://pbs.twimg.com/ media/D2RVjLUUwAIvOFi.png. 2019 (cited on p. 94).
- [226] Marina Plavšić, Gudrunk Klinker, and Heiner Bubb. "Situation awareness assessment in critical driving situations at intersections by task and human error analysis". In: *Human Factors and Ergonomics in Manufacturing & Service Industries* 20.3 (2010), pp. 177–191. DOI: 10.1002/hfm.20173 (cited on p. 11).

- [227] Kelly Pleskot. Genivi AR Manual. https://www.motortrend.com/news/genesisoffers-augmented-reality-owners-manual/. Accessed: 2020-05. 2018 (cited on pp. 52-54).
- [228] Ioannis Politis, Patrick Langdon, Damilola Adebayo, Mike Bradley, P. John Clarkson, Lee Skrypchuk, Alexander Mouzakitis, Alexander Eriksson, James W. H. Brown, Kirsten Revell, and Neville Stanton. "An Evaluation of Inclusive Dialogue-Based Interfaces for the Takeover of Control in Autonomous Cars". In: 23rd International Conference on Intelligent User Interfaces. IUI '18. Association for Computing Machinery, 2018, pp. 601–606. DOI: 10.1145/ 3172944.3172990 (cited on p. 14).
- [229] Kathrin Pollmann, Oilver Stefani, Amelie Bengsch, Matthias Peissner, and Mathias Vukelić. "How to Work in the Car of the Future?: A Neuroergonomical Study Assessing Concentration, Performance and Workload Based on Subjective, Behavioral and Neurophysiological Insights". In: *Proceedings of the 2019 CHI Conference on Human Factors in Computing Systems*. CHI '19. ACM, 2019, 54:1–54:14. DOI: 10.1145/3290605.3300284 (cited on p. 55).
- [230] Aris Polychronopoulos, Manolis Tsogas, Angelos J Amditis, and Luisa Andreone. "Sensor fusion for predicting vehicles' path for collision avoidance systems". In: *IEEE Transactions* on Intelligent Transportation Systems 8.3 (2007), pp. 549–562. DOI: 10.1109/TITS.2007. 903439 (cited on p. 18).
- [231] Darya Popiv, Christoph Rommerskirchen, Klaus Bengler, Markus Duschl, and Mariana Rakic. "Effects of assistance of anticipatory driving on driver's behaviour during deceleration phases". In: *European conference on human centered design for intelligent transport systems*. HUMANIST Publications Lyon. 2010, pp. 133–145 (cited on p. 19).
- [232] Peter W Pruyn and Donald P Greenberg. "Exploring 3d computer graphics in cockpit avionics". In: *IEEE Computer Graphics and Applications* 13.3 (1993), pp. 28–35. DOI: 10.1109/ 38.210489 (cited on p. 67).
- [233] Clarence E. Rash, Michael B. Russo, Tomasz R. Letowski, and Elmar T. Schmeisser. Helmetmounted displays: Sensation, perception and cognition issues. Tech. rep. Army Aeromedical Research Lab Fort Rucker AL, 2009 (cited on p. 58).
- [234] Majken Rasmussen, Esben Pedersen, Marianne Petersen, and Kasper Hornbæk. "Shapechanging interfaces: A review of the design space and open research questions". In: Conference on Human Factors in Computing Systems - Proceedings (2012). DOI: 10.1145/ 2207676.2207781 (cited on p. 24).
- [235] Regulation (EU) 2016/679 of the European Parliament and of the Council of 27 April 2016 on the protection of natural persons with regard to the processing of personal data and on the free movement of such data, and repealing Directive 95/46/EC (General Data Protection Regulation). European Parliament. 4, 2016. URL: http://data.europa.eu/eli/reg/ 2016/679/2016-05-04 (cited on pp. 25, 44).
- [236] Denise Rey and Markus Neuhäuser. "Wilcoxon-Signed-Rank Test". In: *International Ency-clopedia of Statistical Science*. Ed. by Miodrag Lovric. Springer Berlin Heidelberg, 2011, pp. 1658–1659. DOI: 10.1007/978-3-642-04898-2_616 (cited on pp. 76, 84).

- [237] Johannes Rhede, Christoph Wäller, and Peter Oel. "Der FAS Warnbaukasten. Strategie fuer die systematische Entwicklung und Ausgabe von HMI-Warnungen". In: *VDI-Berichte* 2134 (2011) (cited on p. 19).
- [238] Marco Tulio Ribeiro, Sameer Singh, and Carlos Guestrin. ""Why should i trust you?" Explaining the predictions of any classifier". In: *Proceedings of the ACM SIGKDD International Conference on Knowledge Discovery and Data Mining* 13-17-August-2016 (2016), pp. 1135–1144. DOI: 10.1145/2939672.2939778. arXiv: 1602.04938 (cited on p. 26).
- [239] Mireia Ribera and Àgata Lapedriza. "Can we do better explanations? A proposal of usercentered explainable AI". In: *IUI Workshops*. 2019 (cited on p. 28).
- [240] Andreas Riegler, Philipp Wintersberger, Andreas Riener, and Clemens Holzmann. "Augmented Reality Windshield Displays and Their Potential to Enhance User Experience in Automated Driving". In: *i-com* 18.2 (2019), pp. 127–149. DOI: 10.1515/icom-2018-0033 (cited on p. 18).
- [241] Tashko Rizov, Milan Kjosevski, and Risto Tashevski. "Driver Assistance Systems in Vehicles Using Augmented Reality–Benefits and Challenges". In: *Trans Motauto World* 2.5 (2017), pp. 201–206 (cited on pp. 20, 21).
- [242] Dan Robitzski. Exclusive: A Waymo One Rider's Experiences Highlight Autonomous Rideshare's Shortcomings. online. Retrieved September 8, 2019 from https://futurism. com/waymo-one-early-rider-autonomous-vehicle. 2018 (cited on p. 94).
- [243] Fabienne Roche, Anna Somieski, and Stefan Brandenburg. "Behavioral Changes to Repeated Takeovers in Highly Automated Driving: Effects of the Takeover-Request Design and the Nondriving-Related Task Modality". In: *Human Factors* 61.5 (2019). PMID: 30517032, pp. 839–849. DOI: 10.1177/0018720818814963. eprint: https://doi.org/10.1177/ 0018720818814963 (cited on pp. 11, 13).
- [244] Mary Beth Rosson and John M. Carroll. "Scenario-Based Design". In: *The Human-Computer Interaction Handbook: Fundamentals, Evolving Technologies and Emerging Applications*. L. Erlbaum Associates Inc., 2002, pp. 1032–1050 (cited on p. 26).
- [245] Sylvia Rothe, Daniel Buschek, and Heinrich Hußmann. "Guidance in Cinematic Virtual Reality-Taxonomy, Research Status and Challenges". In: *Multimodal Technologies and Interaction* 3.1 (2019), p. 19. DOI: 10.3390/mti3010019 (cited on p. 18).
- [246] Shuvendu Roy and Md. Sakif Rahman. "Emergency Vehicle Detection on Heavy Traffic Road from CCTV Footage Using Deep Convolutional Neural Network". In: 2019 International Conference on Electrical, Computer and Communication Engineering (ECCE). 2019, pp. 1– 6. DOI: 10.1109/ECACE.2019.8679295 (cited on p. 40).
- [247] Sonja Rümelin, Thomas Gabler, and Jesper Bellenbaum. "Clicks Are in the Air: How to Support the Interaction with Floating Objects Through Ultrasonic Feedback". In: *Proceedings of the 9th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '17. ACM, 2017, pp. 103–108. DOI: 10.1145/3122986.3123010 (cited on pp. 20, 21, 52–54).
- [248] Michelle L. Rusch, Mark C. Schall, Patrick Gavin, John D. Lee, Jeffrey D. Dawson, Shaun Vecera, and Matthew Rizzo. "Directing driver attention with augmented reality cues". In: *Transportation Research Part F: Traffic Psychology and Behaviour* 16 (2013), pp. 127–137. DOI: https://doi.org/10.1016/j.trf.2012.08.007 (cited on p. 21).

- [249] Clemens Schartmüller, Andreas Riener, Philipp Wintersberger, and Anna-Katharina Frison.
 "Workaholistic: On Balancing Typing- and Handover-performance in Automated Driving". In: Proceedings of the 20th International Conference on Human-Computer Interaction with Mobile Devices and Services. MobileHCI '18. ACM, 2018, 16:1–16:12. DOI: 10.1145 / 3229434.3229459 (cited on p. 55).
- [250] Bernhard Schick, Corinna Seidler, Seda Aydogdu, and Yu-Jeng Kuo. "Driving Experience Versus Mental Stress at Assisted Lateral Guidance". In: ATZ worldwide 121.2 (2019), pp. 68– 73. DOI: 10.1007/s38311-018-0210-9 (cited on p. 91).
- [251] Bill Schilit, Norman Adams, and Roy Want. "Context-aware computing applications". In: 1994 First Workshop on Mobile Computing Systems and Applications. IEEE. 1994, pp. 85– 90 (cited on p. 18).
- [252] Torben Schinke, Niels Henze, and Susanne Boll. "Visualization of Off-Screen Objects in Mobile Augmented Reality". In: Proceedings of the 12th International Conference on Human Computer Interaction with Mobile Devices and Services. MobileHCI '10. Association for Computing Machinery, 2010, pp. 313–316. DOI: 10.1145/1851600.1851655 (cited on pp. 19, 80, 81).
- [253] Albrecht Schmidt and Thomas Herrmann. "Intervention User Interfaces: A New Interaction Paradigm for Automated Systems". In: *Interactions* 24.5 (2017), pp. 40–45. DOI: 10.1145/ 3121357 (cited on p. 92).
- [254] Alexander Schnorbus and Thomas Gerlach. Georeferenzierte Punktwolke einer Arbeitsstelle längerer Dauer auf der Bundesautobahn 46 in Deutschland. Ed. by Markus Oeser. 2019. DOI: 10.18154/RWTH-2019-01954 (cited on p. 16).
- [255] Nadja Schömig, Katharina Wiedemann, Frederik Naujoks, Alexandra Neukum, Bettina Leuchtenberg, and Thomas Vöhringer-Kuhnt. "An Augmented Reality Display for Conditionally Automated Driving". In: Adjunct Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '18. Association for Computing Machinery, 2018, pp. 137–141. DOI: 10.1145/3239092. 3265956 (cited on p. 21).
- [256] Mary Shaw. "The Role of Design Spaces". In: *IEEE Software* 29.1 (2012), pp. 46–50. DOI: 10.1109/MS.2011.121 (cited on p. 23).
- [257] David Sirkin, Nikolas Martelaro, Mishel Johns, and Wendy Ju. "Toward Measurement of Situation Awareness in Autonomous Vehicles". In: *Proceedings of the 2017 CHI Conference* on Human Factors in Computing Systems. CHI '17. Association for Computing Machinery, 2017, pp. 405–415. DOI: 10.1145/3025453.3025822 (cited on p. 12).
- [258] Jelle Smet, Aykut Avci, Pankaj Joshi, Dieter Cuypers, and Herbert Smet. "P-159L: Late-News Poster: A Liquid Crystal Based Contact Lens Display Using PEDOT: PSS and Obliquely Evaporated SiO2". In: SID Symposium Digest of Technical Papers 43.1 (), pp. 1375–1378. DOI: 10.1002/j.2168-0159.2012.tb06061.x.eprint: https://onlinelibrary. wiley.com/doi/pdf/10.1002/j.2168-0159.2012.tb06061.x (cited on p. 47).
- [259] I Standard. "Ergonomic requirements for office work with visual display terminals (vdts)– part 11: Guidance on usability. ISO Standard 9241-11: 1998". In: *International Organization for Standardization* (1998) (cited on p. 55).

- [260] Simone Stumpf. "Horses for courses: Making the case for persuasive engagement in smart systems". In: *CEUR Workshop Proceedings* 2327 (2019) (cited on p. 26).
- [261] John M Sullivan, Michael J Flannagan, Anuj K Pradhan, and Shan Bao. Literature review of behavioral adaptations to advanced driver assistance systems. Tech. rep. 2016 (cited on p. 27).
- [262] J. Edward Swan, Adam Jones, Eric Kolstad, Mark A. Livingston, and Harvey S. Smallman. "Egocentric depth judgments in optical, see-through augmented reality". In: *IEEE Transactions on Visualization and Computer Graphics* 13.3 (2007), pp. 429–442. DOI: 10.1109/ TVCG.2007.1035 (cited on p. 58).
- [263] J. Edward Swan, Mark A. Livingston, Harvey S. Smallman, Dennis Brown, Yohan Baillot, Joseph L. Gabbard, and Deborah Hix. "A Perceptual Matching Technique for Depth Judgments in Optical, See-Through Augmented Reality". In: *IEEE Virtual Reality Conference* (VR 2006). 2006, pp. 19–26. DOI: 10.1109/VR.2006.13 (cited on p. 58).
- [264] Kohei Tanaka, Yasue Kishino, Tsutomu Terada, and Shojiro Nishio. "A Destination Prediction Method Using Driving Contexts and Trajectory for Car Navigation Systems". In: *Proceedings of the 2009 ACM Symposium on Applied Computing*. SAC '09. Association for Computing Machinery, 2009, pp. 190–195. DOI: 10.1145/1529282.1529323 (cited on p. 104).
- [265] Tito Tang, Frederic Chucholowski, and Markus Lienkamp. "Teleoperated driving basics and system design". In: ATZ worldwide 116.2 (2014), pp. 16–19. DOI: 10.1007/s38311-014-0018-1 (cited on p. 39).
- [266] Victor Ng-Thow-Hing, Karlin Bark, Lee Beckwith, Cuong Tran, Rishabh Bhandari, and Srinath Sridhar. "User-centered perspectives for automotive augmented reality". In: 2013 IEEE International Symposium on Mixed and Augmented Reality - Arts, Media, and Humanities (ISMAR-AMH). 2013, pp. 13–22. DOI: 10.1109/ISMAR - AMH. 2013.6671262 (cited on pp. 20, 25).
- [267] Nava Tintarev and Judith Masthoff. "Designing and Evaluating Explanations for Recommender Systems". In: *Recommender Systems Handbook*. Ed. by Francesco Ricci, Lior Rokach, Bracha Shapira, and Paul B. Kantor. Springer US, 2011, pp. 479–510. DOI: 10. 1007/978-0-387-85820-3_15 (cited on p. 105).
- [268] Marcus Tonnis, Verena Broy, and Gudrun Klinker. "A Survey of Challenges Related to the Design of 3D User Interfaces for Car Drivers". In: *Proceedings of the IEEE Conference on Virtual Reality*. VR '06. IEEE Computer Society, 2006, p. 134. DOI: 10.1109/VR.2006.19 (cited on p. 83).
- [269] Marcus Tonnis, Marina Plavšić, and Gudrun Klinker. "Survey and Classification of Head-Up Display Presentation Principles". In: *Proceedings of the International Ergonomics Association (IEA)*. CD-ROM Proceedings. 2009 (cited on p. 24).
- [270] Marcus Tonnis, Christian Sandor, Christian Lange, and Heiner Bubb. "Experimental Evaluation of an Augmented Reality Visualization for Directing a Car Driver's Attention". In: *Proceedings of the 4th IEEE/ACM International Symposium on Mixed and Augmented Reality*. ISMAR '05. IEEE Computer Society, 2005, pp. 56–59. DOI: 10.1109/ISMAR.2005.31 (cited on pp. 19, 21).

- [271] Cuong Tran, Karlin Bark, and Victor Ng-Thow-Hing. "A Left-Turn Driving Aid Using Projected Oncoming Vehicle Paths with Augmented Reality". In: *Proceedings of the 5th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '13. Association for Computing Machinery, 2013, pp. 300–307. DOI: 10.1145/2516540.2516581 (cited on pp. 21, 22).
- [272] John R Treat, NS Tumbas, ST McDonald, D Shinar, Rex D Hume, RE Mayer, RL Stansifer, and NJ Castellan. Tri-level study of the causes of traffic accidents: final report. Executive summary. Tech. rep. Indiana University, Bloomington, Institute for Research in Public Safety, 1979 (cited on p. 11).
- [273] Joe Tullio, Anind K. Dey, Jason Chalecki, and James Fogarty. "How It Works: A Field Study of Non-Technical Users Interacting with an Intelligent System". In: *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. CHI '07. Association for Computing Machinery, 2007, pp. 31–40. DOI: 10.1145/1240624.1240630 (cited on p. 27).
- [274] Hakan Urey, Kishore V. Chellappan, Erdem Erden, and Phil Surman. "State of the Art in Stereoscopic and Autostereoscopic Displays". In: *Proceedings of the IEEE* 99.4 (2011), pp. 540–555. DOI: 10.1109/JPROC.2010.2098351 (cited on pp. 57, 58).
- [275] Rick Van Krevelen and Ronald Poelman. "A Survey of Augmented Reality Technologies, Applications and Limitations". In: *International Journal of Virtual Reality (ISSN 1081-1451)* 9.2 (2010), p. 1. DOI: 10.20870/IJVR.2010.9.2.2767 (cited on p. 57).
- [276] Arthur B VanGundy. "Brain writing for new product ideas: an alternative to brainstorming". In: *Journal of Consumer Marketing* 1.2 (1984), pp. 67–74. DOI: 10.1108/eb008097 (cited on p. 51).
- [277] Jobil Varghese and Randy G. Boone. "Overview of Autonomous Vehicle Sensors and Systems". In: *Proceedings of the 2015International Conference on Operations Excellence and Service Engineering*. 2015 (cited on p. 10).
- [278] Willem Vlakveld, Nicole van Nes, Jonathan de Bruin, Luuk Vissers, and Merle van der Kroft. "Situation awareness increases when drivers have more time to take over the wheel in a Level 3 automated car: A simulator study". In: *Transportation Research Part F: Traffic Psychology* and Behaviour 58 (2018), pp. 917–929. DOI: https://doi.org/10.1016/j.trf.2018. 07.025 (cited on p. 67).
- [279] Ernst Von Glasersfeld. "An introduction to radical constructivism". In: *The invented reality* 1740 (1984) (cited on p. 28).
- [280] Alla Vovk, Fridolin Wild, Will Guest, and Timo Kuula. "Simulator Sickness in Augmented Reality Training Using the Microsoft HoloLens". In: *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*. CHI '18. Association for Computing Machinery, 2018, pp. 1–9. DOI: 10.1145/3173574.3173783 (cited on p. 121).
- [281] Daniel Wagner and Dieter Schmalstieg. First steps towards handheld augmented reality. 2003, pp. 127–135. DOI: 10.1109/ISWC.2003.1241402 (cited on p. 58).
- [282] Marcel Walch, Tobias Sieber, Philipp Hock, Martin Baumann, and Michael Weber. "Towards Cooperative Driving: Involving the Driver in an Autonomous Vehicle's Decision Making". In: *Proceedings of the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. Automotive'UI 16. ACM, 2016, pp. 261–268. DOI: 10. 1145/3003715.3005458 (cited on pp. 91, 103, 114).

- [283] Marcel Walch, Marcel Woide, Kristin Mühl, Martin Baumann, and Michael Weber. "Cooperative Overtaking: Overcoming Automated Vehicles' Obstructed Sensor Range via Driver Help". In: *Proceedings of the 11th International Conference on Automotive User Interfaces and Interactive Vehicular Applications*. AutomotiveUI '19. Association for Computing Machinery, 2019, pp. 144–155. DOI: 10.1145/3342197.3344531 (cited on p. 92).
- [284] Guy H. Walker, Neville A. Stanton, and Mark S. Young. "Feedback and driver situation awareness (SA): A comparison of SA measures and contexts". In: *Transportation Research Part F: Traffic Psychology and Behaviour* 11.4 (2008), pp. 282–299. DOI: https://doi. org/10.1016/j.trf.2008.01.003 (cited on pp. 11, 12).
- [285] Jingyan Wan and Changxu Wu. "The Effects of Lead Time of Take-Over Request and Nondriving Tasks on Taking-Over Control of Automated Vehicles". In: *IEEE Transactions on Human-Machine Systems* 48.6 (2018), pp. 582–591. DOI: 10.1109/THMS.2018.2844251 (cited on p. 13).
- [286] Jingyan Wan, Changxu Wu, and Yiqi Zhang. "Effects of lead time of verbal collision warning messages on driving behavior in connected vehicle settings". In: *Journal of Safety Research* 58 (2016), pp. 89–98. DOI: https://doi.org/10.1016/j.jsr.2016.07.003 (cited on p. 13).
- [287] Paul Watzlawick. How real is real?: Confusion, desinformation, communication. New York, 1977 (cited on p. 28).
- [288] Paul Watzlawick, John H Weakland, and Richard Fisch. Change: Principles of problem formation and problem resolution. WW Norton & Company, 2011 (cited on p. 28).
- [289] Waymo. Early Rider Moments: "Doing What It's Supposed To". Web. Retrieved September 8, 2019 from https://www.youtube.com/watch?v=Ei5tCVIFVIA. 2018 (cited on p. 94).
- [290] Adam Waytz, Joy Heafner, and Nicholas Epley. "The mind in the machine: Anthropomorphism increases trust in an autonomous vehicle". In: *Journal of Experimental Social Psychology* 52 (2014), pp. 113–117. DOI: https://doi.org/10.1016/j.jesp.2014.01.005 (cited on pp. 15, 30).
- [291] Florian Weidner and Wolfgang Broll. "Smart S3D TOR: Intelligent Warnings on Large Stereoscopic 3D Dashboards during Take-Overs". In: *Proceedings of the 8th ACM International Symposium on Pervasive Displays*. PerDis '19. Association for Computing Machinery, 2019. DOI: 10.1145/3321335.3324937 (cited on p. 21).
- [292] Christian Weiß. "V2X communication in Europe From research projects towards standardization and field testing of vehicle communication technology". In: *Computer Networks* 55.14 (2011). Deploying vehicle-2-x communication, pp. 3103–3119. DOI: https://doi.org/ 10.1016/j.comnet.2011.03.016 (cited on p. 16).
- [293] Julia Werneke and Mark Vollrath. "How to present collision warnings at intersections?—A comparison of different approaches". In: Accident Analysis & Prevention 52 (2013), pp. 91–99. DOI: https://doi.org/10.1016/j.aap.2012.12.001 (cited on pp. 19, 22).
- [294] Bo Westerlund. "Design space conceptual tool–grasping the design process". In: *Nordes, the Nordic Design Research Conference* 1 (2005) (cited on p. 23).
- [295] Gesa Wiegand. "Benefits and Challenges of Smart Highways for the User". In: *IUI Workshops*. 2019 (cited on p. 104).

- [296] Gesa Wiegand, Malin Eiband, Maximilian Haubelt, and Heinrich Hussmann. ""I'd like an Explanation for That!" Exploring Reactions to Unexpected Autonomous Driving". In: 22nd International Conference on Human-Computer Interaction with Mobile Devices and Services (MobileHCI '20), October 5–8, 2020, Oldenburg, Germany. MobileHCI '20. Association for Computing Machinery, 2020. DOI: 10.1145/3379503.3403554 (cited on p. ix).
- [297] Gesa Wiegand and Yuanting Liu. "Highway Sensor System as Enabler for Autonomous Driving". In: vol. CHI 2018 Workshop - Interacting with Autonomous Vehicles: Learning from other Domains(2018). 2018 (cited on p. ix).
- [298] Gesa Wiegand, Christian Mai, Kai Holländer, and Heinrich Hussmann. "InCarAR: A Design Space Towards 3D Augmented Reality Applications in Vehicles". In: *Proceedings of the* 11th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '19. Association for Computing Machinery, 2019, pp. 1–13. DOI: 10.1145/3342197.3344539 (cited on p. ix).
- [299] Gesa Wiegand, Christian Mai, Yuanting Liu, and Heinrich Hussmann. "Early Take-Over Preparation in Stereoscopic 3D". In: Adjunct Proceedings of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. AutomotiveUI '18. Association for Computing Machinery, 2018, pp. 142–146. DOI: 10.1145/3239092. 3265957 (cited on pp. ix, 18, 21, 40, 47, 52, 104).
- [300] Gesa Wiegand, Matthias Schmidmaier, Thomas Weber, Yuanting Liu, and Heinrich Hussmann. "I Drive - You Trust: Explaining Driving Behavior Of Autonomous Cars". In: *Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems*. CHI EA '19. Association for Computing Machinery, 2019. DOI: 10.1145/3290607.3312817 (cited on pp. ix, 103).
- [301] Graham Wilson, Stephen Brewster, Martin Halvey, and Stephen Hughes. "Thermal Icons: Evaluating Structured Thermal Feedback for Mobile Interaction". In: *Proceedings of the 14th International Conference on Human-computer Interaction with Mobile Devices and Services*. MobileHCI '12. ACM, 2012, pp. 309–312. DOI: 10.1145/2371574.2371621 (cited on p. 57).
- [302] Michael G. Wing, Aaron Eklund, and Loren D. Kellogg. "Consumer-Grade Global Positioning System (GPS) Accuracy and Reliability". In: *Journal of Forestry* 103.4 (2005), pp. 169– 173. DOI: 10.1093/jof/103.4.169. eprint: https://academic.oup.com/jof/ article-pdf/103/4/169/25133856/jof0169.pdf (cited on p. 10).
- [303] Susann Winkler, Julia Werneke, and Mark Vollrath. "Timing of early warning stages in a multi stage collision warning system: Drivers' evaluation depending on situational influences". In: *Transportation Research Part F: Traffic Psychology and Behaviour* 36 (2016), pp. 57–68. DOI: https://doi.org/10.1016/j.trf.2015.11.001 (cited on p. 19).
- [304] WMOTORS. Lykan Hypersport. http://www.wmotors.ae/the-lykan-hypersport. Accessed: 2020-06. 2018 (cited on pp. 47, 52-54).
- [305] Jacob O. Wobbrock and Julie A. Kientz. "Research Contributions in Human-Computer Interaction". In: *Interactions* 23.3 (2016), pp. 38–44. DOI: 10.1145/2907069 (cited on p. 4).
- [306] Jonathan Woetzel, Jaana Remes, Brodie Boland, Katrina Lv, Suveer Sinha, Gernot Strube, John Means, Jonathan Law, Andres Cadena, and Valerie Von der Tann. Smart cities: Digital solutions for a more livable future. Online, Accessed: 05-2020. 2018 (cited on p. 1).

- [307] Christine T. Wolf. "Explainability scenarios: Towards scenario-based XAI design". In: International Conference on Intelligent User Interfaces, Proceedings IUI Part F147615 (2019), pp. 252–257. DOI: 10.1145/3301275.3302317 (cited on p. 27).
- [308] Robert F. Woodbury and Andrew L. Burrow. "Whither Design Space?" In: Artificial Intelligence for Engineering Design, Analysis and Manufacturing 20.2 (2006), pp. 63–82. DOI: 10.1017/S0890060406060057 (cited on p. 23).
- [309] Paweł W. Wozniak, Lars Lischke, Sven Mayer, Andreas Preikschat, Markus Schweizer, Ba Vu, Carlo von Molo, and Niels Henze. "Understanding Work in Public Transport Management Control Rooms". In: *Companion of the 2017 ACM Conference on Computer Supported Cooperative Work and Social Computing*. CSCW '17 Companion. Association for Computing Machinery, 2017, pp. 339–342. DOI: 10.1145/3022198.3026341 (cited on p. 40).
- [310] Yiwen Xu and Youchun Jin. Remote road traffic data collection and intelligent vehicle highway system. US Patent 6,401,027. 2002 (cited on p. 17).
- [311] Qian Yang, Aaron Steinfeld, Carolyn Rosé, and John Zimmerman. "Re-Examining Whether, Why, and How Human-AI Interaction Is Uniquely Difficult to Design". In: *Proceedings of the 2020 CHI Conference on Human Factors in Computing Systems*. CHI '20. Association for Computing Machinery, 2020, pp. 1–13. DOI: 10.1145/3313831.3376301 (cited on p. 26).
- [312] Yucheng Yang, Burak Karakaya, Giancarlo C. Dominioni, Kyosuke Kawabe, and Klaus Bengler. "An HMI Concept to Improve Driver's Visual Behavior and Situation Awareness in Automated Vehicle". In: 2018 21st International Conference on Intelligent Transportation Systems (ITSC). 2018, pp. 650–655. DOI: 10.1109/ITSC.2018.8569986 (cited on p. 67).
- [313] Sol Hee Yoon, Young Woo Kim, and Yong Gu Ji. "The effects of takeover request modalities on highly automated car control transitions". In: Accident Analysis & Prevention 123 (2019), pp. 150–158. DOI: https://doi.org/10.1016/j.aap.2018.11.018 (cited on p. 13).
- [314] Veronica Zammitto. "Visualization techniques in video games." In: *EVA*. 2008, pp. 267–276. DOI: 10.14236/ewic/EVA2008.30 (cited on pp. 18, 67).
- [315] Quanwen Zhu, Long Chen, Qingquan Li, Ming Li, Andreas Nüchter, and Jian Wang. "3D LIDAR point cloud based intersection recognition for autonomous driving". In: 2012 IEEE Intelligent Vehicles Symposium. 2012, pp. 456–461. DOI: 10.1109/IVS.2012.6232219 (cited on p. 10).

Eidesstattliche Versicherung

(Siehe Promotionsordnung vom 12.07.11, § 8, Abs. 2 Pkt. 5)

Hiermit erkläre ich an Eidesstatt, dass die Dissertation von mir selbstständig und ohne unerlaubte Beihilfe angefertigt wurde.

München, den 2.6.2020

Gesa Wiegand